

To:	Ralph Cutié, Director
	Miami-Dade Aviation Department

From: Felix Jimenez, Inspector General

Date: February 8, 2024

Subject: OIG Report of Investigation Re: Underreporting of Gross Revenues by MDAD Tenant National Airport Services, Inc., Ref. IG19-0025-I

Enclosed please find the Office of the Inspector General's (OIG's) Report of Investigation regarding the Underreporting of Gross Revenues by Miami-Dade Aviation Department (MDAD) tenant National Airport Services, Inc. (NAS).

The report as a draft was provided to MDAD and NAS for review and the opportunity to provide a written response. MDAD's response, the only response received, is attached to the report as Appendix A.

MDAD's response concurred with the OIG's recommendations and indicated it will notify NAS to discontinue its practice of reporting revenue on a cash basis within 30 days and require NAS to resubmit within 60 days revised monthly gross revenue reports for the period of August 2020 to December 2023. The response also indicates that MDAD will bill any fees due including penalties and interest.

The OIG requests to be provided with a status report, in 120 days, regarding NAS's compliance with MDAD's requests that NAS stop reporting revenue on a cash basis and resubmit its gross revenue reports; as well as MDAD's collection of any fees, penalties, or interest payments from NAS. The status report should be received by the OIG on or before, June 10, 2024.

Attachment

cc: Evelyn Campos, Division Director, Professional Compliance, MDAD Ofelia Tamayo, Director, Audit and Management Services Department National Airport Services, Inc., c/o Lowell Kuvin, Esq. (under separate cover)



MIAMI-DADE COUNTY OFFICE OF THE INSPECTOR GENERAL REPORT OF INVESTIGATION Underreporting of Gross Revenues by MDAD Tenant National Airport Services, Inc. IG19-0025-I

I. INTRODUCTION & SYNOPSIS

The Miami-Dade County (County) Office of the Inspector General (OIG) initiated this investigation based on an anonymous complaint alleging misconduct by National Airport Services, Inc. (NAS), a tenant of the Miami-Dade Aviation Department (MDAD). The complaint alleged NAS was not paying the correct amount of opportunity fees to MDAD for services provided. The complaint also alleged that NAS was operating beyond the scope of its lease agreement by towing aircraft and dispensing fuel at the airside areas of Miami International Airport (MIA) from a modified baggage tug vehicle.

Incidents, similar to the allegation, of NAS operating beyond its authority, had been reviewed by MDAD in 2017 and 2018 closer in time to the complaint. The OIG reviewed MDAD's records of those incidents, interviewed MDAD staff, and NAS's owner. MDAD found violations and addressed them with NAS. Specifically, regarding the fueling violations, MDAD's Airside Operations imposed two civil fines. The OIG has confirmed that both of these citations are still open and unpaid.¹ From August 29, 2017, through September 11, 2022, Airside Operations issued no additional citations for fueling violations to NAS.² Based on MDAD's monitoring and enforcement actions, the OIG's investigation focused on the allegation of underreporting gross revenues.

The OIG investigation substantiated that NAS underreported gross revenues and underpaid opportunity fees to MDAD. The OIG presented the underreporting to the Miami-Dade State's Attorney Office (SAO) for possible prosecution. Following a lengthy investigation, the SAO declined to prosecute, finding that while NAS deliberately underreported gross revenues, there was no evidence that NAS did so with any fraudulent intent. The OIG found that NAS's underreporting of gross revenues was due to its consistent reporting of cash-basis revenues. It appears that both NAS and MDAD used the internal audit mechanism to determine and reconcile any shortfalls in reported gross revenues at year-end. Although NAS has paid outstanding opportunity fee balances, plus interest and penalties, this practice is problematic and subjects MDAD to unnecessary risk of non-payment of past due opportunity fees.

¹ The total civil penalty for each violation was \$260.00.

² In researching NAS citations, the OIG discovered a citation issued by Miami-Dade Fire Rescue on 10/21/2022, for failure to pay an inspection fee billed under life safety permit 21902-00112. While NAS eventually paid the inspection fee, in June of 2023, the citation balance had increased to \$4,210 due to noncompliance by NAS in correcting the violation within the time allowed and non-payment of the initial \$210 citation.

II. OIG JURISDICTIONAL AUTHORITY

In accordance with Section 2-1076 of the Code of Miami-Dade County, the Inspector General has the authority to make investigations of County affairs; audit, inspect, and review past, present, and proposed County and Public Health Trust programs, accounts, records, contracts, and transactions; conduct reviews, audits, inspections, and investigations of County and Public Health Trust departments, offices, agencies, and boards; and require reports from County and Public Health Trust officials and employees, including the Mayor, regarding any matter within the jurisdiction of the Inspector General.

III. ENTITIES OR INDIVIDUALS RELEVANT TO THIS INVESTIGATION

National Airport Services, Inc. (NAS)

NAS has been an active Florida for-profit corporation since October 14, 2014. The registered agent and president is Jorge Y. Alvarez, Jr., and the corporate address is 5439 NW 36th Street, Miami, Fl. 33166. NAS is in the business of cargo handling at MIA. NAS operates from Building 716, sections C, D and E, on the West side cargo area of MIA. As of November 2022, NAS employs approximately one hundred employees.

Real Estate Management and Development Division (MDAD Properties)

MDAD Properties oversees the administration of lease agreements under which airlines, cargo handlers, maintenance repair and overhaul (MRO) operators, Fixed Base Operators (FBO), and other tenants service the traveling public and each other. It consists of the following four sections: Terminal, MIA Non-Terminal (Cargo), General Aviation Airports (GAA), and Permits. Compliance is monitored through inspections of leased space by MDAD Properties personnel.

MDAD Finance Division, Revenue Section

MDAD Finance is responsible for preparing timely, accurate, and reliable monthly financial statements and aviation statistics; processing vendor invoices, generating customer invoices; collecting all department revenues; and maintaining accurate accounts receivable balances. Regarding issues relevant to this report, the Revenue Section of the MDAD Finance Division is charged with accepting and reviewing a tenant or permittee's Monthly Report of Gross Revenues (MRGR) form and processing the opportunity fees to be paid.

In consideration of the ability to transact business at the airport, tenants and permittees must pay the County an opportunity fee. The opportunity fee set under the agreements with NAS is seven percent of the monthly gross revenues exceeding the monthly rentals. The agreements with NAS define "gross revenues" as:

...all moneys paid or payable to, or considerations of determinable value received by the Lessee for sales made, transactions had, or services rendered in providing cargo handling and cargo related services to others ..., regardless of when or where the order is therefor is received, whether paid or unpaid, whether on a cash or credit basis or in consideration of any other thing of value.

The MRGR includes a list of customer names and the gross revenues derived from the services provided to a tenant/permittee's customer. The MRGR must be submitted to MDAD Finance by the 10th day of the month following the month in which services were rendered. It is the Revenue Sections' job to ensure that tenants/permittees are paying opportunity fees based on the agreement's definition of gross revenues.

IV. BACKGROUND

History of NAS Lease Agreements

Since approximately 2016, NAS has held consecutive lease agreements to operate in various locations at MIA.

Lease Agreement No. C-009584 - On April 26, 2016, MDAD and NAS entered Lease Agreement No. C-009584, for a term not to exceed five years, Exhibit 1. The leased premises were located in Building 716F on the Westside Cargo area of MIA.

The Lease Agreement described the opportunity fees due to MDAD for gross revenues, the required monthly revenue reports, and audit requirements. Per the agreement the authorized use of the leased space by NAS, included receiving, storing, consolidating, packaging, and transporting air cargo for airlines and NAS customers, including loading and unloading of such aircraft, except that such activity was not authorized in the terminal area of MIA. The lease specified that the cargo handling business was to be performed by NAS employees, and NAS had no right to employ agents to perform such business unless such agents had written authorization from MDAD through a permit or lease agreement.

On July 12, 2016, Lease C-009584 was amended to allow additional authorized uses of the leased space and additional prohibited activities. The added authorized use of leased space included repair, maintenance, and rental of ground service equipment. The added prohibited activities included storage of any fuel or chemical tanks.

Lease Agreement No. C-010718 - On January 31, 2019, Lease Agreement No. C-010718 was entered into between MDAD and NAS for a term of five years. Under this Agreement, the Lessee relocated to MDAD's Cargo Section at Building 716-E West.

Lease Agreement No. C-011987 - Lease Agreement C-011987 was entered into between MDAD, NAS, and National Airline Charters, Inc. (NAC) for a term of 1 year until September 22, 2022, to use warehouse and office space at Building 716 C and D.

Lease Agreement No. C-011952 - On October 6, 2022, by Resolution No. R-914-22, the Board of County Commissioners approved the Lease Agreement C-011952 entered into between MDAD, NAS, and National Airline Charters, Inc. (NAC) for a term of ten years, replacing and superseding Lease Agreement No. C-00-11987. Under this Agreement, "the Lessees will occupy the entire warehouse space at Building 716 C and D, plus first floor offices, and all mezzanine offices …" The agreement requires that the lessees pay MDAD a fee of \$400,000, payable in five (5) equal payments of \$80,000 per year. This Lease Agreement has no connection or impact on NAS' Lease Agreement No.C-010718.

The Lease Agreement described the opportunity fees due to MDAD for gross revenues, the required monthly revenue reports, and audit requirements. Per the agreement the authorized use of the leased space by NAS includes receiving, storing, consolidating, packaging, and transporting air cargo for NAS customers, including loading and unloading of such aircraft. The lease also authorizes use of the leased space for repair, maintenance, and rental of ground service equipment, and prohibited activities includes storage of any fuel or chemical tanks.

V. <u>CASE INITIATION & INVESTIGATIVE METHODOLOGY</u>

The OIG initiated the investigation based on an anonymous complaint alleging NAS was operating outside the scope of its MDAD Lease Agreement and underpaying opportunity fees to MDAD.

Due to the complainant's anonymity, the OIG was unable to follow up with the complainant for further details of the allegations. The OIG conducted a review of records related to NAS. The OIG reviewed and analyzed NAS lease agreements, permits, and conducted a review of NAS's opportunity fee payment history by analyzing the monthly gross revenue reports submitted to MDAD's Finance division. The OIG tested the veracity of NAS's monthly reports by comparing them to information gleaned from NAS invoices issued to customers and NAS audited financial statements. The OIG also interviewed NAS's president Jorge Alvarez and MDAD officials. This investigation was conducted in accordance with the *Principles and Standards for Offices of Inspector General* as promulgated by the Association of Inspectors General.

VI. INVESTIGATIVE FINDINGS

Based on MDAD's past enforcement actions regarding NAS operating outside the scope of its agreement the OIG's investigation focused on the allegation of underreporting gross revenues.³

OIG Finding – NAS Underreported Gross Revenues and Underpaid Opportunity Fees to MDAD

As part of the OIG's investigation, we reviewed NAS's MRGR submitted to MDAD. We also reviewed NAS invoices obtained from NAS customers, and NAS draft Financial Statements approved by NAS's independent outside auditor. The OIG review found that NAS materially underreported its gross revenues during each of the three years reviewed, 2018 through 2020. Below are the facts underlying this finding:

A. Comparison of NAS Gross Revenues as Reported in its MRGR vs Audited Financial Statements

The OIG compared NAS's audited financial statements to NAS's MRGRs submitted to MDAD for the same periods. **Table 1** below shows the gross revenues reported by NAS to MDAD in their monthly revenue reports for the years 2018 – 2020. **Table 2** shows the gross revenues NAS's independent auditors reported to MDAD for the same periods.

Reported by NAS	Cargo Handling Gross Sales	As %	All Other Gross Sales	As %	Total Gross Sales
Year ended July 2018	1,794,471	75%	602,201	25%	2,396,672
Year ended July 2019	3,763,087	94%	248,089	6%	4,011,176
Year ended July 2020	4,184,331	95%	203,685	5%	4,388,016
Totals	\$9,741,889	90%	1,053,975	10%	\$10,795,864

Table 1: NAS's MRGR Gross Revenues

³ Although not the focus of the investigation, MDAD's enforcement actions resulted in citations, which the OIG has found are still outstanding.

Reported by NAS	Cargo Handling Gross Sales	As %	All Other Gross Sales	As %	Total Gross Sales
Year ended July 2018	1,839,665	67%	911,852	33%	2,751,517
Year ended July 2019	4,782,084	92%	426,846	8%	5,208,930
Year ended July 2020	5,344,721	96%	203,682	4%	5,548,403
Totals	\$11,966,470	89%	\$1,542,380	11%	\$13,508,850

Table 2: NAS's Audited Financial Statement Gross Revenues

Clearly, there is a significant difference—year after year—between NAS gross revenues reported to MDAD and NAS's gross revenues detailed in NAS's audited financial statements. Table 3 below shows this analysis.

Reported by NAS	Gross Revenues Reported in Audited Financial Statements	Gross Revenues Reported to MDAD	Underreported Gross Revenues
Year ended July 2018	2,751,517	2,396,672	354,845
Year ended July 2019	5,208,930	4,011,176	1,197,754
Year ended July 2020	5,548,403	4,388,016	1,160,387
Totals	\$13,508,850	\$10,795,864	\$2,712,986

The cumulative difference for the three-year period reviewed by the OIG was \$2,712,986. These underreported gross revenues resulted in the underpayment of opportunity fees during the respective years of \$28,805, \$29,818, and \$65,564, when applying the applicable opportunity fee calculation. Table 4 below shows this analysis.

MIAMI-DADE COUNTY OFFICE OF THE INSPECTOR GENERAL REPORT OF INVESTIGATION Underreporting of Gross Revenues by MDAD Tenant National Airport Services, Inc.

Year ended	Gross Revenues Reported in Audited Financial Statements	7 % Opportunity Fees Owed to MDAD	Gross Revenues Reported to MDAD	7% Opportunity Fees Paid to MDAD	Total Opportunity Fees Underpayment
July 2018	2,751,517	65,745	2,396,672	36,940	28,805
July 2019	5,208,930	40,493	4,011,176	10,675	29,818
July 2020	5,548,403	65,564	4,388,016	0	65,564
Totals	\$13,508,850	\$171,802	\$10,795,864	\$47,615	\$124,184

Table 4: OIG Analysis of NAS Underpaid Opportunity Fees 2018 through 2020

B. NAS Underreporting due to Cash Basis Accounting.

The OIG determined that the main factor contributing to the underreporting was that NAS's monthly revenue reports to MDAD only reported cash received and not full gross revenues, which include cash received and accounts receivable. According to the NAS/MDAD Lease Agreement:

Section 3.02 <u>Opportunity Fee:</u>

... Lessees shall pay to the County the amount by which 7% of the monthly Gross Revenues, as defined in Article 3.10 (Gross Revenues), generated from its activities under this Agreement, *exceeds the monthly rentals*, as adjusted from time to time...

Section 3.10 Gross Revenues:

The term "Gross Revenues," as used in this Agreement means all moneys paid or payable to, or considerations of determinable value received by the Lessee for sales made, transactions had, or services rendered in providing cargo handling and cargo related services to others as defined in Article 2.04...regardless of when or where the order is therefor is received, whether paid or unpaid, whether on a cash or credit basis or in consideration of any other thing of value...

Based on the agreement, the opportunity fee payable to MDAD by NAS should be calculated as follows:

(Gross Revenue x 7%) – Monthly Rent = Opportunity Fee Payable to MDAD

During the course of the investigation, the OIG interviewed MDAD's Finance Controller. According to the Controller, the monthly gross revenue calculation is always the calculation listed in the contract, unless an appendix applies. The Controller stated that an Appendix did not apply in the case of NAS. She stated that gross revenues on the NAS contract should be calculated as shown above.

The Controller explained that once MDAD receives the audited financial statements, the Concessions Group (within MDAD Finance) analyzes the annual audits to ensure that the correct calculation of gross revenues is used. The Concessions Group uses the numbers provided from the audit and applies the correct opportunity fee calculation as stated by the contract. For example, the Concessions Group would ensure that gross revenues are not incorrectly limited to a cash basis. MDAD's Controller stated that at times, the audited financial statements provided to MDAD contain incorrect calculations of the gross revenues. For example, lessees (and their auditors) may try to submit credits on the monthly revenue reports (where monthly rent exceeded 7% of gross revenues), however, MDAD would not accept those credits. Following MDAD's calculations, any discrepancies along with the appropriate interest and late fees are billed to the contract holder.

The OIG also inquired as to whether MDAD Finance is notified of amendments to lease agreements. The Controller explained that similar to the Concessionaire Team, the employees in the Revenue Section within the Finance Division oversee lease amendments. While Properties creates the agreement, it is Finance that activates the billing. Accordingly, Finance has all contract amendments and makes necessary adjustments.

NAS's Explanation of its Gross Revenue Calculation

The OIG interviewed NAS owner, Mr. Alvarez, regarding NAS's underreporting of opportunity fees. Mr. Alvarez confirmed that he is the person who completes the MRGRs and submits them to MDAD. Mr. Alvarez stated that other than the language in the lease agreement, MDAD did not provide NAS with any additional written guidance or formula on how to properly compute the opportunity fees due to MDAD.

Mr. Alvarez explained he operates his business on a cash basis. He stated that it's hard to have all the invoices at the end of the month due to the back and forth with the airlines. Additionally, he stated he needs to be in receipt of funds from airlines and other customers to be able to make opportunity fee payments to MDAD. He provided an example that airlines will not pay an invoice if there are errors of any sort. Mr. Alvarez stated that at the end of the year when the audit is complete, everything is paid for in full.

Mr. Alvarez said that MDAD had never taken issue because they charge him interest and late fees and NAS always pays them. He stated that was the point of doing the independent audit. Mr. Alvarez then reiterated that NAS' opportunity fees needed to be paid in full prior to the start of the new lease agreement with MDAD.

In August 2023, the OIG conducted a follow-up email interview with the MDAD Controller regarding NAS's explanation of its use of cash basis accounting. The Controller responded that MDAD was unaware that NAS was reporting on a cash basis. MDAD does not audit the books and records of permittees or tenants. Unless the audit report brings the issue to MDAD's attention, it remains unaware. The controller did state that once she receives the final OIG report, she will raise the issue with NAS via formal letter reminding the company of the contract terms.

The OIG notes that the NAS 2018 Audit Report submitted to MDAD included the observation that NAS was reporting on a cash basis. Figure 1 below contains the observation from the Audit Report.

Figure 1: 2018 NAS Audit, page 6

Observation 2018-1

National Airport Services, Inc. is required to submit a monthly report of gross revenues to the Miami-Dade Aviation Department. This monthly report contains the computation of total fees due for the month. The fee payment is then due on the tenth day of the month following the month in which the gross revenues were received or accrued. During our audit procedures, we noted the monthly gross revenue reports submitted during the year ended July 31, 2018, were calculated on the cash basis. The report should have been prepared on the accrual basis in accordance with the agreement. This caused underreporting of revenue on a monthly basis and therefore an additional amount due to Miami-Dade Aviation Department of \$28,805, including estimated late fees of \$1,916 through July 31, 2018. Additional interest may have continued to accrue on this balance through the date this report was available to be issued.

The 2019 and 2020 audits included similar observations. NAS has failed to submit yearly independent audits to MDAD for the years 2021 and 2022. In March 2023, MDAD issued a demand letter to NAS, yet the audits have still not been submitted. According to MDAD Finance the 2023 audit is now also due.

Based on the foregoing, the OIG determined that this allegation against NAS is substantiated. The OIG presented the facts to the SAO for possible prosecution.

C. State Attorney's Office Analysis of NAS's Intent to Defraud

The underpayment of opportunity fees by NAS was originally referred to the SAO for criminal investigation. After reviewing the evidence derived by the OIG and SAO, the SAO investigation was closed without the filing of criminal charges. As documented in the SAO close-out memo, there is insufficient evidence to prove beyond a reasonable doubt that a crime was committed by NAS, despite NAS habitually underreporting its gross revenues to MDAD.

The SAO close out memo cites the misapplication of "Gross Revenue" Definition:

It appears that instead of reporting pursuant to the broad definition of gross revenues, i.e., all monies paid or payable, NAS was only reporting monies paid or received from its customers. These collected funds could be days, weeks or months after the actual sale or rendering of service. This is a direct violation of the lease agreement.

The SAO memo concludes by noting:

The investigation revealed that NAS's underreporting of gross revenues and failure to pay required fees at the time they were due was ongoing for at least 3 years. In each case, required annual audits have revealed the underreporting. In response to the audits, NAS paid the fees applicable to the underreported gross revenues, and additionally paid accrued interest assessed by MDAD on the outstanding balances. It should be noted that while the lease agreement contemplates occasional underreporting and provides for accrued interest on these outstanding balances, the underreporting here is different. In this case, NAS participated in a systematic ongoing course of conduct over the years. Further, the amounts unreported are material, in some cases close to 30% of annual gross revenue. Those material underreported opportunity fee amounts resulted in additional payments to MDAD of \$28,805, \$29,818, and \$65,564, respectively.

Given that the annual audits are supplied to MDAD, MDAD was aware that NAS was underreporting income. The investigation did not uncover evidence of NAS attempting to hide the underreporting. Based upon the above facts, the State cannot prove beyond a reasonable doubt that NAS had an intent to steal at the time of the monthly takings. This investigation is closed.

As noted in the SAO's Close Out Memo, it cannot be proved beyond a reasonable doubt that NAS's habitual underpayment of opportunity fees was committed with an intent to steal or defraud the County.

VII. ISSUANCE OF DRAFT REPORT FOR REVIEW AND COMMENT

This report, as a draft, was provided to MDAD and NAS, through its attorney, Lowell Kuvin, Esq., for their review and submission of discretionary written responses. We did not receive a response from NAS.

The OIG received a response from MDAD, a copy of which is attached to this report as Appendix A. MDAD concurred with the OIG's recommendations. The OIG is pleased with MDAD's concurrence and proposed actions going forward. Our specific comments related to MDAD's response are noted below each recommendation.

VIII. CONCLUSION AND RECOMMENDATIONS

The anonymous complainant's allegation that NAS underreported its gross revenues is substantiated. Relating to the performance of unauthorized services, MDAD has and continues to monitor, and if necessary, issue citations, specifically fueling citations. Although the investigation focused on the underreporting of gross revenues, in reviewing the citations issued by MDAD, the OIG found that NAS has long-overdue unpaid citations.

Regarding NAS's underreporting of gross revenues, NAS has done so knowingly and consistently. The MDAD lease agreement is very clear as to the broad definition of gross revenues. Revenues are to be reported when services are *rendered*, and revenue is *accrued*. However, NAS's owner stated that due to financial reasons, NAS reports revenues on a cash basis only, i.e., when the revenue is *paid* by the NAS customer.

Instead of taking steps to address this issue through the years, it appears that both NAS and MDAD used the internal audit mechanism to determine and reconcile any shortfalls in reported gross revenues at year-end. NAS then paid any outstanding opportunity fee balances, plus interest, and penalties. This practice subjects MDAD to unnecessary risk of non-payment of past due opportunity fees. The risk is exacerbated by NAS's failure to timely provide the contractually mandated yearly audits, which at present are three years in arrears (2021, 2022, and 2023).

Acquiescing to a company habitually underreporting its revenues, only to reconcile at year's end with added interest and penalties, is tantamount to extending unsecured credit. As a further concern, NAS's footprint at MIA has expanded with the latest ten-year lease agreement signed in late 2022. MDAD risks a proportional increase in delayed payments, interest, and penalties, therefore, the OIG makes the following recommendations:

Recommendation 1: MDAD should review its procedures to ensure that MDAD's Properties and Finance sections specifically communicate to ensure that lessees do not have outstanding audit reports or a history of gross revenue underreporting that should be considered prior to extending or renewing lease agreements.

OIG Comments to MDAD's Response: MDAD in its response concurred with the recommendation and committed to enhanced communications between the Finance Division and the Real Estate Management Division; to notifying Property Managers immediately of tenants with outstanding independent audit reports; and to encouraging eviction proceedings if not remedied within 60 days.

MDAD also indicated it will notify NAS to discontinue its practice of reporting revenue on a cash basis and require NAS to resubmit revised monthly gross revenue reports for the period of August 2020 to December 2023. MDAD will bill any fees due with interest and request that the County's Audit and Management Services Department do a follow-up audit of NAS.

The OIG concurs with this action and requests that MDAD advise the OIG of the revised filings and any monies collected from NAS as a result.

Recommendation 2: MDAD should review its procedures regarding tenants who habitually underreport gross revenues to determine if any new procedures, enhanced penalties, and training might be implemented to minimize the occurrence and the risk to MDAD or MDAD should exercise its rights under the contracts to terminate the agreements.

OIG Comments to MDAD's Response: MDAD concurred with the recommendation, and advised of its procedures, which require monitoring and reminders to tenants with outstanding balances. To the extent that MDAD's response indicates that its current procedures are adequate, the OIG recommends that additional training be conducted of Accounts Receivable Section employees and Property Managers to ensure that staff are reminded and up to date on the most current procedures.

Recommendation 3: Generally, MDAD should periodically review outstanding fines relating to citations issued to all lessees and take action to recover the fines and penalties due, particularly before a new lease agreement is executed. Specifically, as to NAS, MDAD should review the unpaid citations issued to NAS and recover the fines and penalties due.

OIG Comments to MDAD's Response: MDAD also concurred with this recommendation and stated it would notify the Clerk of Courts to revisit the outstanding citations. The OIG understands that the collection of citations is handled by the Clerk of Courts and that MDAD's ability to assist in those efforts is limited. The OIG recommends that periodic reminders be sent to the Clerk of Courts to ensure collection. Also, as noted in the OIG's recommendation, prior to the execution of a new lease, MDAD should ensure fines and penalties have been paid.

Miami-Dade County Office of the Inspector General



EXHIBIT 1

Lease Agreement #C-009584 Between Miami-Dade County and National Airport Services, Inc. (100 pages)

19-0025-l

Lease No. <u>C-010718</u> Cust. No.: <u>NATP39</u>

LEASE AGREEMENT BETWEEN **MIAMI-DADE COUNTY**, FLORIDA, AS LESSOR, AND **NATIONAL AIRPORT SERVICES, INC**., AS LESSEE, MIAMI INTERNATIONAL AIRPORT

31 THIS LEASE AGREEMENT ("Agreement") is made and entered into as of the day of <u>January</u>, <u>19</u>, (the "Agreement Date"), by and between **MIAMI-DADE COUNTY**, a political subdivision of the State of Florida ("County"), through its Aviation Department ("MDAD"), and **NATIONAL AIRPORT SERVICES**, **INC.**, a Florida Corporation ("Lessee").

WITNESSETH:

WHEREAS, Lessee is being relocated to Building 716-E (West) by MDAD, and, Lessee has offered to renovate its relocated space so as to make it suitable for Lessee's use. Such renovation work to be accomplished under the Lessor's Tenant Airport Construction-Reimbursable process (TAC-R); and

WHEREAS, the parties also agree that rent payment shall commence on the earlier of THREE MONTHS after the date the agreement is executed by MDAD or the Date of Beneficial Occupancy (DBO), with the DBO being the date on which (i) the Lessee occupies the facility in whole or part for any use other than for the purpose of completing the improvements, equipping the facility, or securing the necessary inspections, permits, and certificates, or (ii) Lessor determines that the renovated facility is capable of being used for its intended purpose;

NOW THEREFORE, and in consideration of the premises and of the mutual covenants and agreements herein contained, the parties hereto agree as follows:

ARTICLE 1 Term and Premises

1.01 <u>Term</u>:

The County hereby leases to the Lessee, and the Lessee hereby leases from the County, the premises described in Article 1.02 (the "Premises") hereof, for the purposes and uses set forth in Article 2 ("Use of Premises") hereof, commencing on the Agreement Date and extending thereafter for a **five year period** (the "Term") from the earlier of the Date of Beneficial Occupancy ("DBO"), as defined above and determined solely by the County, or THREE MONTHS after the Agreement Date (the earlier of such dates being the "Commencement Date"). Either party may terminate this Agreement upon 60 days written notice.

1.02 <u>Premises</u>:

The Premises leased herein are in **Building 716-E (West)**, located at 6025 NW 18th Street, Miami FL 33122 in the West Cargo Area of Miami International Airport ("Airport"), and are more particularly described as follows and as shown on Exhibits A, B and P dated October 04, 2018, attached hereto and made a part hereof ("Premises"):

Buidling 716-E West			
	Suites - Column	ID#	SQ.FT.
Ground Level A/C Office		#87161620	463
Ground Level A/C Office		#87161625	364
Non A/C Warehouse	Column 4L to 9L	#117161630	20,137
Mezzanine Level A/C Office			1,025
Mezzanine Level A/C Office			319
Mezzanine Level A/C Office			957
Vehicle Parking			4 spaces
Land			20,964

1.03 Suitability of Premises:

The Lessee acknowledges that the Premises are suitable for the Lessee's proposed use and that the County has no obligation to perform or cause to be performed any maintenance, repairs, clean-ups, painting, or the like. The Lessee's obligation under this Agreement, such as in Article 6.01(B) (Permits and Licenses), to obtain all operating permits required of the Lessee, shall not require the County to take any action or perform any tasks within the Premises to enable the Lessee to obtain such permits, including, but not limited to, certificates of occupancy, which shall remain the Lessee's exclusive obligation to perform in order to obtain such permits.

1.04 Adjustment and Relocation of Premises:

(A) The Premises leased hereunder are subject to relocation, or modification, by addition or deletion, with the mutual consent of the Lessee and the County Aviation Department (the "Department"), and Articles 1.02 (Premises) and 3.01 (Annual Rental) hereof and the exhibits to this Agreement shall then be administratively revised to reflect such relocation, modification, or deletion upon 30 days written notice to the Lessee by the Department. Relocated space may not be similar in size, configuration, or location to the Premises leased herein.

(B) In the event the Lessee requests approval to handle cargo for others within and from the Premises, pursuant to Article 2.04 (Cargo Handling of Others) hereof, if the Department, in its sole judgment, determines that the Lessee is leasing more space than it requires for its own operations, the Department may modify the Premises by deleting such excess space from the Premises and, in such event, Articles 1.02 (Premises) and 3.01 (Annual Rental) hereof and accompanying exhibits to this Agreement shall thereafter be administratively adjusted to reflect such deletion, upon not less than 30 days written notice to the Lessee.

1.05 County Right to Terminate for Airport Development:

The County shall have the right, at any time during the term of this Agreement, to terminate the Agreement upon not less than six months advance written notice to the Lessee, if the Premises are needed by the County for Airport development projects as approved by the Board of County Commissioners. Such termination shall not entitle the Lessee to any relocation costs, and in no event shall the County be responsible to the Lessee for any costs, damages, expenses, moving costs, loss of revenues, business interruption damages, or the like, caused by or arising out of such termination, nor shall such termination be deemed an eminent domain taking for any purpose.

1.06 Aircraft Parking

The Lessee shall be entitled to use aircraft parking position(s), as shown on Exhibit P, dated October 4, 2018, attached hereto and made a part hereof, as designated from time to time by the Department, for the parking and ground handling of the Lessee's regularly scheduled all-cargo flights, or for all-cargo flights routed to the Airport due to weather or emergency conditions, for time periods and at use fees as may be established from time to time by the Department. It shall be the responsibility of the Lessee to request use and assignment of each parking position in advance to secure approval of such aircraft parking from the Department, in accordance with procedures established from time to time by the Department, and to pay the established rates and fees required for such use and assignment. The Department shall make reasonable efforts to assign the Lessee aircraft parking position(s) most convenient to the Premises leased by the Lessee hereunder, subject to availability and within the assignment procedures established by the Department; provided, however, the County shall have no liability to the Lessee if such parking positions are not available for assignment to the Lessee. Use of the aircraft parking position(s) by the Lessee shall be solely for aircraft operated by or on behalf of the Lessee and shall not be transferred or assigned by the Lessee to others.

ARTICLE 2 Use of Premises

The County hereby grants to the Lessee the following general privileges, uses and rights, all of which shall be subject to the terms, conditions and covenants hereinafter set forth and all of which shall be non-exclusive on the Airport.

2.01 Authorized Use of Premises:

The Lessee shall use the Premises for the following purposes only:

- (A) As to the Ground Floor operation and office space:
 - (1) Receiving, storing, consolidating and packaging of air cargo for airlines. Handling and pre-cooling of flowers and other perishable fresh produce cargo for third parties (Lessee's customers) including pickup and delivery, loading and unloading of aircraft (prohibited in the Terminal Building area), refrigeration, storage, consolidation and dispatch of flowers and other perishable cargo transported or to be transported by air or land.
 - (2) Transportation, from and to other designated premises, of air cargo of the Lessee's customers to and from the leased premises.
 - (3) Transportation of air cargo to and from aircraft, including the loading and unloading of such aircraft, except that such loading and unloading of aircraft is not authorized in the Terminal Building Area of the Airport. The maintenance and storage of cargo handling vehicles and equipment of the Lessee is prohibited except in areas specifically leased by the Lessee hereunder, unless otherwise approved in writing by the Department.
 - (4) Preparation of manifests and other documentation and the performance of administrative functions directly related to the services provided pursuant Sections (1) through (3) above.
 - (5) Operation of bonded and non-bonded warehouses.

- (6) For normal office operations associated with the Lessee's cargo handling business.
- (7) Such other activities as are directly related or incidental to the authorized cargo handling uses of the Premises.
- (8) All activities listed above related to the Lessee's cargo handling business shall be performed by employees of the Lessee. There shall be no right of the Lessee to employ agents to perform activities hereunder unless such agents have authorization from the Department through permit or lease to operate on the Airport.
- (B) Paved ramp space, immediately adjacent to the leased Premises and not designated for aircraft parking, may be used for the following purposes only:
 - (1) Staging, loading and unloading of air cargo shipped or to be shipped by the Lessee.
 - (2) Parking of operable aircraft service equipment of the Lessee related to its cargo handling activities.

The Lessee shall not permit these activities to interfere with designated service roads or fire lanes or so as to interfere with the activities of other.

2.02 Common Use Aircraft Parking Positions (if applicable):

Common use aircraft parking positions, adjacent or proximate to the leased Premises, may be used for the following purposes only:

- (A) For loading and unloading of all-cargo aircraft owned, operated, wet-leased, dryleased or chartered by the Lessee.
- (B) For the performance of transit or turnaround aircraft maintenance on all-cargo aircraft owned, operated, wet-leased, dry-leased or chartered by the Lessee.
- (C) For the fueling and servicing of all-cargo aircraft being handled by the Lessee.

Common use aircraft parking positions shall not be used for the parking or storage of ground service equipment, containers or like equipment.

2.04 Vehicular Parking:

- (A) Vehicular parking on the landside, immediately adjacent to the leased Premises may be used only as truck loading and unloading stalls and for business guest parking. The Lessee shall not park its own vehicles nor permit its officers or employees to park in such stalls.
- (B) Common use public parking areas adjacent to or in the vicinity of the Building 716 may be used only by officers, employees and business guests of the Lessee and other tenants of the Building and nearby buildings. Vehicles of the Lessee which are not required for day-to-day operations of the Lessee or which are not operable shall not be parked at, in or near the Building in which the Premises are located nor elsewhere on the Airport.

- (C) Parking of any type (i.e. employee, public or business guest, trucks, etc.) on any landscaped or grassed area is prohibited.
 - 2.05 Cargo Handling of Others:

Intentionally Omitted

2.06 <u>Non-Flyable Aircraft</u>: Aircraft in non-flyable condition shall not be parked or stored on the leased Premises for a period in excess of 60 days without the prior written approval of the Department, which shall not be unreasonable withheld or delayed.

ARTICLE 3 Rentals and Payments

3.01 Annual Rental:

As of the Rent Commencement Date, as annual rental for the lease of the Premises, the Lessee shall pay to the County, commencing on the Commencement Date as defined in Article 1.01, the sum of \$331,595.65 payable in twelve equal monthly installments of \$27,632.97 in U.S. funds, on the first day of each and every month in advance and without billing, at the offices of the Department as set forth in Article 3.06 (Address for Payments). Said rental is computed as follows:

Building 716-E West				
Description	SQ.FT.	RATE	ANNUAL	MONTHLY
A/C Ground Office	463	\$15.00	\$6,945.00	\$578.75
A/C Ground Office	364	\$15.00	\$5,460.00	\$455.00
Non A/C Cargo Warehouse	20,137	\$12.25	\$246,678.25	\$20,556.52
Mezzanine A/C Office	1,025	\$12.60	\$12,915.00	\$1,076.25
Mezzanine A/C Office	319	\$12.60	\$4,019.40	\$334.95
Mezzanine A/C Office	957	\$12.60	\$12,058.20	\$1,004.85
Vehicle Parking	4 spaces	\$55.00	\$2,640.00	\$220.00
Land	20,964	\$1.95	\$40,879.80	\$3,406.65
*Total:			\$331,595.65	\$27,632.97

*Plus applicable utilities and State Sales Tax, as required by law.

Rental rates include common area janitorial, excludes land. Appraised rate includes a charge of \$.10 per square foot as a trash removal fee.

3.02 <u>Opportunity Fee</u>:

The Lessee shall pay to the County the amount by which **7%** of the monthly Gross Revenues, as defined in Article 3.10 (Gross Revenues), generated from its

activities under this Agreement, exceeds the monthly rentals, as adjusted from time to time, under Article 3.01 and 3.05. The Lessee shall pay such amount to the County by the tenth day of the month following the month in which the Gross Revenues were received or accrued. The percentage fees payable on any unreported Gross Revenues, determined by the annual audit required pursuant to Article 3.13 (Annual Audit Required) are considered, for the purposes of Article 3.07 (Late Payment Charge), as having been due on the tenth day of the month following the month during which such unreported Gross Revenues were received or accrued. Such opportunity fee is not a payment for the lease or license to use the Premises, but rather payment for the Lessee's privilege of doing business on the Airport.

3.03 <u>Security Deposit</u>:

Unless otherwise provided in the Airport Use Agreement executed by the Lessee, prior to occupancy of the Premises, the Lessee shall pay to the County an amount equal to two times the required total monthly rental as determined pursuant to Article 3.01 above, plus applicable State sales tax thereon, as security for the payment of the Lessee's obligations hereunder. Said deposit shall be in addition to any rental payments required hereunder, and the Department shall be entitled to apply such payment to any debt of the Lessee to the Department that may then exist, as permitted by law, including but not limited to the rentals required hereunder. In lieu of the security deposit being made in cash, the Department, in its sole discretion, may authorize the Lessee to provide an irrevocable Letter of Credit, in a form provided by the Department, in like amount. The amount of the security deposit is subject to adjustment by the Department at any time there is a change in the annual or monthly rentals pursuant to the terms of this Agreement.

3.04 <u>Rental Rate Review</u>:

In the event the Lessee is in possession of the Premises by virtue of this Agreement on October 1, 2018 (or any annual anniversary thereafter), the rental rates stated in Article 3.01 (Annual Rental) above shall be subject to review and adjustment as set forth hereafter. When such rental rate adjustments are approved by the Board of County Commissioners, and new or revised rental rates applicable in whole or in part to the Premises are established by said Board, the Department shall notify the Lessee in writing of such rates and this Agreement shall be considered and deemed to have been administratively amended to incorporate the revised rental rates effective as of such October 1st date. Payments for any retroactive rental adjustments shall be due upon billing by the Department and payable within ten calendar days of same. If Lessee is not notified in writing of any adjustment to the rental rates within ninety (90) days following October 1 of any year during the term of this Agreement, the County waives its right to adjust the rental rates for that year.

3.05 Double Rental:

In the event that the Lessee remains in possession of the Premises beyond the expiration or termination of this Agreement, the Lessee shall be bound by all of the terms and conditions of this Agreement to the same extent as if this Agreement were in full force and effect during the time beyond the expiration date of this Agreement. However, during any such possession of the Premises as a holdover tenant after the County has demanded the return of the Premises, the Lessee shall be liable for double rentals for so long as the Lessee remains in possession after such demand, such rentals to be based upon the rental rates applicable from time to time in whole or in part to the Premises.

3.06 Address for Payments:

The Lessee shall pay, by mail, all rentals, fees and charges required by this Agreement to the following:

By mail:

Miami-Dade County Aviation Department Accounting Division P.O. Box 526624 Miami, Florida 33152-6624

By hand delivery to the offices of the Department during normal working hours to the following:

Miami-Dade County Aviation Department 4200 N.W. 36 Street Miami, Florida 33122

By electronic funds transfer for immediate credit via wire transfer to:

Bank: Bank of America Miami, Florida ABA Number: 026009593 Swift Code Number: BofAUS3N Account Name: Miami-Dade Aviation Department Bank Account Number: 001180000120

Note: Transaction must include the Aviation Department invoice number(s) of charges to be paid.

By electronic funds transfer for next day credit via the ACH (Automatic Clearing House) to:

Bank: Bank of America Miami, Florida ABA Number: 063100277 Account Number: Miami-Dade Aviation Department Bank Account Number: 001180000120

Note: Transaction must include the Aviation Department invoice number(s) of charges to be paid.

3.07 Late Payment Charge:

In the event the Lessee fails to make any payments, as required to be paid under the provisions of this Agreement, within ten days after same shall become due, interest at the rates established from time to time by the Board of County Commissioners of Miami-Dade County, Florida (currently set at 1½% per month), shall accrue against the delinquent payment(s) from the original due date until the Department actually receives payment. The right of the County to require payment of such interest and the obligation of the Lessee to pay same shall be in addition to and not in lieu of the rights of the County to enforce other provisions herein, including termination of this Agreement, and to pursue other remedies provided by law.

3.08 Dishonored Check or Draft:

In the event that the Lessee delivers a dishonored check or draft to the County in payment of any obligation arising under the terms of this Agreement, the Lessee shall

incur and pay a service fee of TWENTY-FIVE DOLLARS, if the face value of the dishonored check or draft is \$50.00 or less, THIRTY DOLLARS, if the face value of the dishonored check or draft is more than \$50.00 and less than \$300.00, or FORTY DOLLARS, if the face value of the dishonored check or draft is \$300.00 or more, or Five Percent of the face value of such dishonored check or draft, whichever is greater. Further, in such event, the Department may require that future payments required pursuant to this Agreement be made by cashier's check or other means acceptable to the Department.

3.09 <u>Utilities</u>:

Unless the Premises are separately metered and billed directly to the Lessee by the utility company or included in the rental rates and noted in Article 3.01, the Lessee hereby agrees to pay monthly, upon billing by the Department, for utilities consumption in the Premises. This monthly charge will be based on a survey conducted by or on behalf of the Department of the consumption by the Lessee and current nondiscriminatory rates charged others at the Airport. This monthly charge may also be adjusted and billed retroactively, from time to time, based on changes in consumption and rates and limited to a retroactive period that is reasonable under the circumstances and no greater than the retroactive period used by Lessor for all similarly situated tenants. The Lessee shall pay for all other utilities used by it. In the event the premises are metered and billed to the Department, the Lessee shall pay for utility consumption based on the metered charge. The County shall have no obligation to provide utilities to the Premises other than those existing as of the effective date of this Agreement.

3.10 Gross Revenues:

The term "Gross Revenues", as used in this Agreement means all monies paid or payable to, or considerations of determinable value received by the Lessee for sales made, transactions had, or services rendered in providing cargo handling and cargo related services to others as defined in Article 2.04 under this Agreement and from the subleasing of office space or other portions of the Premises, regardless of when or where the order therefor is received, whether paid or unpaid, whether on a cash or credit basis or in consideration of any other thing of value; provided, however, that any taxes imposed by law which are separately stated to and actually paid by a customer and directly payable by the Lessee to a taxing authority and sales refunds shall be excluded therefrom.

3.11 <u>Records and Reports</u>:

The Lessee shall keep in Miami-Dade County, during the term of this Agreement, all books of account, records and reports used in its operation necessary to report Gross Revenues and to calculate the Opportunity Fees payable hereunder and as may, from time to time, be required by the Department to document its activities pursuant to this Agreement; provided, however, that Lessee may request the Department to review all or a portion of such items at a location outside of the county, in which case Lessee shall pay the Department all costs associated with travel, lodging and meals of the Department's personnel and auditors to review the documents at such location. All Gross Revenues hereunder shall be accounted for in accordance with generally accepted accounting principles. The form of all such books of account, records and reports shall be subject to the approval of the Department and/or the auditors of the County (one or more of the following: the designated external auditing firm or other certified public accounting firm selected by the Department, the County's Department of and Management Services, or auditors of the State of Florida). Audit Recommendations for changes, additions, or deletions to such books of account, records and reports by the auditors of the County shall be complied with by the Lessee

when requested by the Department. The auditors of the County shall be permitted, during normal working hours, to audit and examine all books of account, records, and reports relating to the operations of the Lessee hereunder, including, but not limited to, balance sheets, profit and loss statements, deposit receipts, Florida State Sales Tax Reports, and such other documents as may be determined by the Department to be necessary and appropriate; provided, however, that the Lessee shall not be required to retain such records in Miami-Dade County, Florida, for more than three years following termination of this Agreement.

3.12 Monthly Report of Gross Revenues:

For each month in which Lessee provides services that are subject to the Opportunity Fee in Article 3.02, on or before the tenth business day following the last day of each such month throughout the term of this Agreement, Lessee shall furnish to the Department a statement of Gross Revenues received or accrued during the preceding calendar month and certify as to the accuracy of such Gross Revenues in the form prescribed by the Department. In the event Lessee fails to submit the monthly report by the tenth (10) business day following the last day of the month, a penalty fee of Fifty Dollars (\$50.00) per day shall be imposed for each calendar day following the report due date until the report is received by the Department, up to a maximum of Seven Hundred Fifty dollars (\$750.00) per violation.

3.13 Annual Audit Required:

Within sixty days of each anniversary of the commencement date of this Agreement and within sixty days following termination of this Agreement, the Lessee shall, at its sole cost and expense, provide to the Department on an annual (or portion thereof) basis an audit report of monthly Gross Revenues, containing an unqualified opinion, prepared and attested to by an independent certified public accounting firm. licensed in the State of Florida. The report shall include a schedule of Gross Revenues and percentage opportunity fees paid to the County under this Agreement, prepared in accordance with the comprehensive basis of accounting defined under terms of this Agreement and reported in the format as subsequently prescribed by the Department. The audit shall be conducted in accordance with generally accepted auditing standards and include issuance of a management letter, which will contain the findings discovered during the course of the examination, such as recommendations to improve internal controls and other significant matters related to this Agreement. In addition, the audit shall include comprehensive compliance procedures to determine whether the books of account, records and reports were kept in accordance with the terms of this Agreement for the period of examination. The auditor shall report such procedures and findings in a separate letter report to the Department. The last such report shall include the last day(s) of operations.

3.14 Waiver of Annual Audit:

Notwithstanding the provisions of Article 3.13 (Annual Audit Required) above, Lessee shall have the option to provide a certification, rather than the audit required in Article 3.13 above, if the Annual Gross Revenues under this Agreement for any year of this Agreement are less than \$250,000.00. Such certification shall be in a form specified or approved by the Department. Lessee acknowledges that if it is determined by the Department as a result of an audit or inspection of the Lessee's books and records, as authorized pursuant to Article 3.15 (Right to Inspect) below, that the Lessee has understated its Gross Revenues and that the certification is incorrect, (a) Lessee shall be liable to all remedies provided by law for such false certification, including charges under Section 837.012, Florida Statutes, for having filed a false oath, (b) such certification may be considered by the Department as a breach of the Agreement and the Department may take appropriate action as a result, (c) the Department shall be entitled to assess a 50% surcharge on the difference between the Opportunity Fees stated on the certification and the Opportunity Fees shown by the audit to be due and payable, and (d) the annual audit waiver pursuant to the provisions of this Article 3.14 shall no longer be applicable for the remaining term of this Agreement.

3.15 Right to Inspect:

The Department and the auditors of the County shall have the right, without limitation, to enter upon the Premises at anytime during normal operating hours of the Lessee to: (1) inspect, review, verify and check all or any portion(s) of the Lessee's procedures for recording or compiling Gross Revenue information by day or month; and (2) audit, check, inspect and review all books of account, records, financial reports, financial statements, operating statements, inventory records, copies of State sales tax returns, and work papers relating to the operation of the Lessee, and other pertinent information as may be determined to be needed or desirable by the Department.

3.16 Monthly Cargo Statistics Report:

The Lessee, on or before the 10th day of every month, shall provide to the Accounting Offices of the Department, indicated in Article 3.07 (Address for Payments) hereof, a monthly cargo statistics report, in a form specified by the Department.

3.17 Utilities:

Unless the Premises are separately metered and billed directly to the Lessee by the utility company or included in the rental rates and noted in Article 3.01, the Lessee hereby agrees to pay monthly, upon billing by the Department, for utility consumption in the Premises. This monthly charge will be based on a survey conducted by or on behalf of the Department of the utility consumption by the Lessee and/or current nondiscriminatory rates charged others at the Airport. This monthly charge may also be adjusted and billed retroactively, from time to time, based on the changes in consumption and rates. The Lessee shall pay for all utilities used by it. In the event the Premises are metered and billed to the Department, the Lessee shall pay for utility consumption based on the metered charge. The County shall have no obligation to provide utilities to the Premises other than those existing as of the effective date of this Agreement.

ARTICLE 4 Maintenance And Repair by Lessee

4.01 <u>Cleaning</u>:

The Lessee shall, at its sole cost and expense, perform or cause to be performed, services which will at all times keep the Premises and, to the extent in Lessee's control, the adjacent non-leased aircraft ramp clean, neat, orderly, sanitary and presentable. Lessee's obligation as specified above with regard to the ramp areas will be discharged to the extent within Lessee's control; it being acknowledged that persons other than Lessee have access to and use of the ramp areas.

4.02 <u>Removal of Trash</u>:

The Lessee shall, at its sole cost and expense, remove from the Premises all trash and refuse which might accumulate and arise from its use of the Premises and the business operations of the Lessee under this Agreement. Such trash and refuse shall be stored temporarily and disposed of in a manner approved by the Department.

4.03 Maintenance and Repairs:

The Lessee shall repair and maintain in good condition the Premises and all improvements or alterations thereto, except for those items for which the County is responsible pursuant to Article 5 (Maintenance by County). Such repair and maintenance shall include, but not be limited to, interior painting, floor coverings, doors, windows, pavement (landside and airside), equipment, dock levelers, protection bumpers attached to building, furnishings, skylights, fixtures, appurtenances, replacement of light bulbs, ballasts and tubes and the replacement of all broken glass, and shall at all times be based on a standard of care reflecting prudent property management. Maintenance and repairs shall be in guality and class equal to or better than the original work to preserve the Premises in good order and condition. The Lessee shall repair all damage caused by the Lessee and its employees, agents, independent contractors, patrons, servants or invitees. Prior to or at termination of this Agreement, injury done by the installation or removal of furniture and personal property of the Lessee shall be repaired so as to restore the Premises to their original state, except as the Premises may have been altered by the Lessee with the approval of the Department pursuant to Article 7.01 (Alteration), and to guit and surrender up the Premises in the same good order and condition as it was at the commencement of this Agreement, reasonable wear and tear and damage caused by an Act of God excepted; provided however, that such return of the Premises under this Article 4.03 shall not relieve the Lessee of its obligations for damages to the Premises that may be specifically provided elsewhere in this Agreement.

4.04 Excavation of Land:

No excavation of any of the land shall be made, no soil or earth shall be removed from the Premises, and no well of any nature shall be dug, constructed or drilled on the Premises, except as may be required for environmental monitoring purposes pursuant to Article 8 (Environmental Compliance).

4.05 <u>Water and Sewerage System</u>:

The Lessee shall, at its sole cost and expense, operate and maintain all the components of the existing water, sanitary sewerage and storm drainage facilities within the boundaries of the Premises. The Lessee shall not make any alterations or modifications to such facilities without the advance written approval of the Department.

4.06 Industrial Waste Facilities:

The Lessee shall be fully responsible for all industrial wastes exiting the Premises and in response thereto shall provide, operate and maintain adequate facilities on the Premises for separating, neutralizing and treating industrial waste and foreign materials and the proper disposal thereof, in accordance with applicable laws, rules and regulations.

4.07 Modifications or Access to Roof:

The Lessee covenants that it shall not install, attach, suspend or in any manner modify the roof, its members or structures nor shall it permit any person to walk on the roof or its members without the prior written consent of the Department. In the event the Lessee violates this covenant, the County shall not have any responsibility for any damages to the property of the Lessee or others inside the leased Premises caused by rain or other hazard in any way related to the roof.

4.08 Loading Dock/Platform:

If the Premises include a loading dock/platform, the Lessee shall keep such loading dock/platform clean and clear at all times unless actively loading or unloading, and shall not use the loading dock/platform for the storage of cargo, equipment or any other materials.

4.09 Inspections:

The Department and/or its designated representatives shall have the right, during normal working hours, to inspect the Premises to identify those items of maintenance, repair, replacement, modification and refurbishment required of the Lessee or the County, pursuant to Article 5 (Maintenance by County), to keep the Premises in good order and condition. The Lessee shall perform all corrective work required of it, identified in such inspection(s) within 30 days of receipt of written notice from the Department, or if such corrective work cannot reasonably be completed within such 30 day period, Lessee shall commence such work within 30 days of receipt of written notice and complete such work as expeditiously thereafter as reasonably possible. Trash and debris problems shall be corrected within 24 hours following receipt of either oral or written notice from the Department.

4.10 Failure to Maintain:

If it is determined by the Department that the Lessee has failed to properly clean, remove trash and refuse, maintain, repair, replace and refurbish the Premises as required by this Article 4 (Maintenance and Repair by Lessee), the Department shall provide to the Lessee a list of deficiencies, reflecting the amount of time to be reasonably allowed for the Lessee to correct same. If the Lessee fails to correct such deficiencies within the time allowed and has not registered an objection as to its obligation to do so, the Department, following 10 days further notice to the Lessee, may enter upon the Premises and perform all work, which, in the judgment of the Department, may be necessary, and the County shall add the cost of such work, plus 25% for administrative costs, to the rent due hereunder on the first day of the month following the date of such work, and such cost shall be and constitute a part of the rent. Subsequent to receipt of the further notice of intent to perform repairs or cleanup from the Department, the Lessee shall not undertake performance of such repairs or cleanup without specific prior written authorization from the Department, not to be unreasonably withheld if the Department has not yet contracted for or commenced the needed repairs or cleanup.

ARTICLE 5 Maintenance by County

5.01 <u>County Maintenance</u>:

The County shall operate and maintain in good condition all components of the existing water, sanitary sewerage and storm water drainage facilities that lie outside the boundaries of the Premises. The County shall maintain the roof, its structural supports and exterior walls of the building. The County shall maintain all common areas including parking areas, hallways, restrooms, elevators, landscaping, irrigation system and the Building's central air conditioning system. The County shall have no maintenance responsibility within the Premises. If any of such facilities are damaged or destroyed by the operations of the Lessee, the Department shall make the necessary repairs or replacements and shall bill the Lessee for the costs of same, plus 25% for administrative costs, in the manner specified in Article 4.06 (Failure to Maintain) hereof.

5.02 County Maintenance Subject to Certain Conditions:

Such maintenance by the County may be subject to interruption caused by repairs, strikes, lockouts, labor controversies, inability to obtain, fuel, power or parts, accidents, breakdowns, catastrophes, national or local emergencies, acts of God, and other conditions beyond the control of the County. Upon any such happening, the Lessee shall have no claim for damages for the County's failure to furnish or to furnish in a timely manner any such maintenance; provided, however, that the Department, in its sole discretion, may provide a rent abatement for that portion of the Premises rendered unusable for the period of time that the County is unable to make the repairs required by Article 5.01 (County Maintenance). The County shall exercise reasonable diligence to remedy and/or cure any such interruptions; to the extent such interruptions are within the County's control.

ARTICLE 6 Regulations, Licenses and Permits

6.01 <u>Rules and Regulations - General</u>

A) Rules and Regulations:

1. The Lessee shall comply with all Ordinances of the County, including the Rules and Regulations of the Department, Chapter 25, Code of Miami-Dade County, Florida, as the same may be amended from time to time, Operational Directives issued thereunder, all additional laws, statutes, ordinances, regulations and rules of the Federal, State and County Governments, and any and all plans and programs developed in compliance therewith, which may be applicable to its operations or activities under this Agreement, specifically including, without limiting the generality hereof, Federal air and safety laws and regulations and Federal, State and County environmental laws.

2. During the renovation period of the work to be performed by Lessee hereunder, and at any time Lessee performs any work on the Premises, Lessee shall comply with all design and construction requirements of MDAD, including, but not limited to, MDAD's Tenant Airport Construction (Reimbursable or Non-Reimbursable) requirements ("TAC"), as they may be amended from time to time.

All design and construction work must be approved in advance by MDAD, to the extent set forth in such TAC.

B) Permits and Licenses:

1. The Lessee, at its sole cost and expense, shall be liable and responsible for obtaining, paying for, maintaining on a current basis, and fully complying with, any and all permits, licenses and other governmental authorizations, however designated, as may be required, at any time throughout the entire term of this Agreement, by any Federal, State, or County governmental entity or any judicial body having jurisdiction over the Lessee or the Lessee's operations and activities, for any activity of the Lessee conducted on the Premises and for any and all operations conducted by the Lessee, including insuring that all legal requirements, permits and licenses necessary for or resulting, directly or indirectly, from the Lessee's operations and activities on the Premises have been obtained and are being fully complied with.

2. Such permits and licenses shall include, but not be limited to, a Certificate of Use and Occupancy and any required Industrial Waste or Operating Permits from

Department of Regulatory and Economic Resources. To the extent that may be required, at either the inception of this Agreement, or within a permissible time period following the commencement of tenant's use of the facility for its intended purpose, the Lessee shall provide to the Department evidence that it has obtained the Certificate of Use and Occupancy and, as applicable, the appropriate operating Waste Permit(s). Upon written request of the Department, the Lessee shall provide to the Department evidence that it has obtained the Certificate of Use and Occupancy and, as applicable, the appropriate operating Waste Permit(s). Upon written request of the Department, the Lessee shall provide to the Department evidence that it has obtained to the Department and licenses, and applications therefor, which the Department may request.

C) Violations of Rules and Regulations:

Subject to the County's obligations as confirmed in Article 8 (Environmental Compliance), Lessee agrees to pay on behalf of the County any penalty, assessment or fine issued against the County, or to defend in the name of the County any claim, assessment or civil action, which may be presented or initiated by any agency or officer of the Federal, State or County governments, based upon a claim or allegation that the Lessee, its agents, employees, invitees, or trespassers have violated any law, ordinance, regulation, rule or directive described in Article 6.01 above or any plan or program developed in compliance therewith. The Lessee further agrees that the substance of this Article 6.01 (Rules and Regulations - General) above shall be included in every sublease, contract and other agreement, which the Lessee may enter into related to its operations and activities under this Agreement and that any such sublease, contract and other agreement shall specifically provide that "Miami-Dade County, Florida is a third party beneficiary of this and related provisions." This provision shall not constitute a waiver of any other conditions of this Agreement prohibiting or limiting assignments, subletting or subcontracting.

6.02 Aircraft Noise Abatement Regulations Compliance:

The Lessee hereby specifically acknowledges its awareness that the noise generated by aircraft, while on the ground and in flight, may cause annoyance to residential areas in proximity to the Airport and/or under or near flight corridors serving the Airport and that the County, as proprietor of the Airport, may, therefore, from time to time adopt certain lawful policies, procedures and/or regulations, not inconsistent with the Airport Noise and Capacity Act of 1990, unless otherwise approved by the Federal Aviation Administration, and not inconsistent with aircraft safety or the Lessee's rights hereunder, intended to abate the effects of aircraft noise. Said policies, procedures and/or regulations may deal with, but are not necessarily limited to, nighttime engine run-ups, preferential runway usage, aircraft arrival and departure patterns, use of displaced runway thresholds, and the like. The Lessee specifically understands and agrees that a violation of such noise abatement policies, procedures and/or regulations of such noise abatement policies, procedures and/or regulations of such noise abatement policies, procedures and/or regulations may result in the arrest or citation of the offending party, with the imposition of fines, and that a violation of such pursuant to the provisions hereof.

ARTICLE 7 Alteration of Premises and Erection of Signs

7.01 <u>Alteration</u>:

The Lessee shall not alter the Premises in any manner whatsoever without the prior written approval of the Department. In the event the Lessee is given approval to make alterations to the Premises, the Lessee shall fully comply with the terms and conditions of such approval, as contained in the Department's approval letter. Failure to do so shall constitute a default pursuant to Article 13.03 (Other Defaults) hereof, unless cured by removal of the unapproved alterations and restoration of the premises to its

original condition within the notice period pursuant to said Article 13.03.

The Lessee acknowledges that the Premises have been provided by the County with a basic level of "tenant finishes" and that, therefore, any additional or special finishes desired by the Lessee shall be at the Lessee's sole cost and expense and that rentals, pursuant to Article 3.01 (Annual Rental), shall be payable during the period the Lessee is installing such additional or special finishes.

7.02 <u>Signage</u>:

The Lessee shall not erect, maintain or display any identifying signs or any advertising matter, of any type or kind which is visible to the public, without prior written approval of the Department. In the event the Department changes the graphics system for the identification of lessees at the Airport, the Lessee agrees, if required by the Department, to change, at its sole cost, any of its identification signs necessary to comply with such graphics system.

ARTICLE 8 Environmental Compliance

8.01 <u>Definitions:</u>

For purposes of this Agreement, the following additional definitions apply:

- A) "Baseline Environmental Conditions" means the presence or release of Hazardous Materials, at, on, under, or from the Premises prior to Lessee's Occupancy Date, the presence or release of which was not caused by Lessee or Lessee's agents, employees, contractors, invitees or trespassers. Solely for purposes of this Agreement, it shall be presumed that the Baseline Environmental Conditions consist of the conditions identified in any existing (as of the date of this Agreement) Miami-Dade County maintained records, including contamination assessment reports and any other technical reports, data bases, remedial action plans, the Baseline Audit or the presence, discharge, disposal or release of any other Hazardous Materials originating prior to the Occupancy Date that comes to be located on the Premises and not caused by Lessee or Lessee's agents, employees, contractors, invitees or trespassers.
- B) "Environmental Claim" means any investigative, enforcement, cleanup, removal, containment, remedial or other private, governmental or regulatory action at any time threatened, instituted or completed pursuant to any applicable Environmental Requirement, against Lessee with respect to its operations at Miami International Airport or against or with respect to the Premises or any condition, use or activity on the Premises (including any such action against Lessee with respect to its operations at Miami International Airport or against or made by any person against County), and any claim at any time threatened or made by any person against Lessee with respect to its operations at Miami International Airport or against or with respect to the Premises or any condition, use or activity on the Premises or any condition, use or activity on the Premises (including any such claim against County), relating to damage, contribution, cost recovery, compensation, loss or injury resulting from or in any way arising in connection with any Hazardous Material or any applicable Environmental Requirement.
- C) "Environmental Law" means any applicable federal, state or local law, statute, ordinance, code, rule, or regulation, or license, authorization, decision, order, injunction, or decree, any of which may be issued by a judicial or regulatory body of competent jurisdiction, or rule of common law including, without limitation, actions in nuisance or trespass, and any judicial or agency interpretation of any

of the foregoing, which pertains to health, safety, any Hazardous Material, or the environment (including but not limited to ground or air or water or noise pollution or contamination, and underground or aboveground tanks) and shall include without limitation, the Solid Waste Disposal Act, 42 U.S.C. § 6901 et seq.; the Comprehensive Environmental Response, Compensation and Liability Act of 1980, 42 U.S.C. § 9601 et seq. ("CERCLA"), as amended by the Superfund Amendments and Reauthorization Act of 1986 ("SARA"); the Hazardous Materials Transportation Act 49 U.S.C. § 1801 et seq.; the Federal Water Pollution Control Act, 33 U.S.C. § 1251 et seq.; the Clean Air Act 42 U.S.C. § 7401 et seq.; the Toxic Materials Control Act 15 U.S.C. § 2601 et seq.; the Safe Drinking Water Act, 42 U.S.C. § 300f et seq.; Chapters 403, 376 and 373, Florida Statutes; Chapters 24 and 25 of the Miami-Dade County Code, and any other applicable local, state or federal environmental statutes, codes, or ordinances, and all rules, regulations, orders and decrees now or hereafter promulgated under any of the foregoing, as any of the foregoing now exist or may be changed or amended or come into effect in the future.

- D) "Environmental Requirement" means any Environmental Law, or any agreement or restriction entered into or applicable by law (including but not limited to any condition or requirement imposed by any insurance or surety company), as the same now exists or may be changed or amended or come into effect in the future, which pertains to health, safety, any Hazardous Material, or the environment, including but not limited to ground or air or water or noise pollution or contamination, and underground or aboveground tanks. With regard to any environmental agreements or restrictions entered into by MDAD, MDAD agrees to use reasonable efforts to apprise the Miami Airport Affairs Committee (the "MAAC"), which represents the interests of the airlines at the Airport, of any such agreements or restrictions and to obtain the MAAC's input into such items.
- E) "Hazardous Material" means any substance, whether solid, liquid or gaseous, which is listed, defined or regulated as a hazardous substance, a hazardous waste or pesticide, or otherwise classified as hazardous or toxic, in or pursuant to any applicable Environmental Requirement; or which is or contains asbestos, radon, any polychlorinated biphenyl, urea formaldehyde foam insulation, explosive or radioactive material, or motor fuel or other petroleum hydrocarbons; or which causes or poses a threat to cause contamination or a nuisance on the Premises ,any adjacent Premises or a hazard to the environment or to the health or safety of persons on the Premises or Other Airport Property.
- F) "Initial Construction Period" means for any lease which contemplates construction or renovation for Premises not previously occupied in whole or in part by Lessee under this Agreement and/or any previous Agreement, a period of time not to exceed six (6) months commencing with the date on which Lessee breaks ground on the Premises for construction of foundations or commences such renovation.
- G) "Occupancy Date" means the date Lessee first entered, occupied or took possession of the Premises under any written or verbal agreement.
- H) "On" or "in" when used with respect to the Premises or any premises adjacent to the Premises, means "on, in, under, above or about."
- I) "Other Airport Property" means property on the Airport occupied or used by Lessee, or upon which Lessee performs operations, but which is not subject to a

lease, sublease or other legal agreement governing the terms of Lessee's occupation, use or operations at such property.

- J) "Recognized Environmental Condition" shall have the meaning set forth in ASTM E 1527-05, Section 1.1.1, as such provision may be amended or superseded from time to time.
- K) "Release" means any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping or disposing into the environment.
- L) "Remediation" means any investigation, clean-up, removal action, remedial action, restoration, repair, response action, corrective action, monitoring, sampling and analysis, installation, reclamation, closure, or post-closure in connection with the suspected, threatened or actual release of Hazardous Materials.
- M) "Trespassers" means third parties who have entered the Premises and whose actions while on the Premises have resulted in a Release of Hazardous Materials directly onto the Premises. Notwithstanding the foregoing, for purposes of this Agreement, Trespassers shall not include those third parties whose actions took place off of the Premises and which resulted in the presence of Hazardous Materials on the Premises due to the migration of Hazardous Materials from that off-Premises location.

8.02 Lessee's Industrial Classification:

Lessee represents and warrants to County that Lessee's Standard Industrial Classification ("SIC") code number, as published in the most recent SIC Manual from the United States Office of Management and Budget, and as used on Lessee's Federal Tax Return is 4512.

8.03 Lessee's Acceptance of the Risks and Condition of Premises As-Is:

Lessee agrees that the Premises shall be leased and delivered to Lessee in its current "as-is/with all faults" condition (but it is not intended by this provision that County be relieved from its duties expressly set forth in this Agreement or any other applicable agreement). Lessee hereby, warrants, covenants, agrees, and acknowledges that:

- A) Hazardous Materials may be present on the Premises and Other Airport Property. The County is currently engaged in a significant environmental remediation program at MIA.
- B) Under Article 8.06 below, Lessee is provided the opportunity to conduct an independent investigation of the Premises and the physical condition thereof, including the potential presence of any Hazardous Materials on or about the Premises. Lessee's report on the investigation, if any such report has been prepared, has been provided to the County and is listed in Exhibit K attached to this Agreement. Whether Lessee has conducted such an investigation or not, Lessee is willing to proceed with this Agreement notwithstanding the environmental conditions of the premises or the properties surrounding the premises, subject to Lessee's right to terminate this Agreement as otherwise provided herein.
- C) Because of the possible presence of environmental contaminants on the Premises or other Airport property, County has made no express, implied, or other representations of any kind with respect to the suitability or usability of the Premises or other Airport property, or any improvements appurtenant thereto, including,

without limitation, the suitability or usability of any building materials, building systems, soils or groundwater conditions (due to the presence of Hazardous Materials in, on, under, or about the Premises or other Airport property), for Lessee's proposed or intended use, and Lessee has relied solely on Lessee's own inspection and examination of such matters.

D) Except as to County's obligations set forth in this Article or elsewhere in this Agreement, Lessee expressly assumes the risk that Hazardous Materials that are or may be present on the Premises at the commencement of this Agreement may affect the suitability or usability of the Premises for Lessee's proposed or intended use. Lessee agrees that, except to the extent of County's Remediation obligations provided in this Article 8, or any other discharge, disposal or release of Hazardous Materials or violation of Environmental Requirements, caused by County, its agents, employees or contractors and except with respect to Baseline Environmental Conditions, County shall have no responsibility or liability with respect to any Hazardous Materials on the Premises. Notwithstanding the foregoing, in no event shall County be liable to Lessee for damages relating to physical or personal injury, business interruptions relocation costs or any other cost (other than a cost for which County is liable under this Article 8) resulting from the presence of Hazardous Materials on the Premises at any time during this Agreement.

8.04 <u>Responsibilities for Hazardous Materials:</u>

A) Unless the parties agree otherwise in writing, the County shall conduct response actions mandated by existing Environmental Requirements applicable to the County for (i) Hazardous Materials disclosed in the Lessee Audit to the extent required by Article 8.06 and (ii) Baseline Environmental Conditions, provided however that:

(1) To the extent this Agreement covers Premises not previously occupied by Lessee and if this Agreement contemplates construction or renovation by the Lessee, any Hazardous Material discovered during the Initial Construction Period as defined in Article 8.01(F), shall be presumed to be a Baseline Environmental Condition under this Agreement except to the extent the Aviation Department demonstrates to the satisfaction of Lessee by written notice setting forth the Aviation Department's explanation as to why the Hazardous Material originated from a discharge, disposal or release that was caused by Lessee, Lessee's agents, employees, contractors, invitees or Trespassers. Should Lessee determine that such a demonstration has not been made to Lessee's satisfaction, County may invoke the dispute resolution provision of 8.16. Until such time as the parties reach an agreement or such time as the dispute is otherwise resolved, responsibility for such Hazardous Material shall remain with the Aviation Department.

(2) To the extent this Agreement covers premises previously occupied by Lessee, and except for Baseline Environmental Conditions, Remediation of any Hazardous Material discovered on the Premises shall be the responsibility of the Lessee, except to the extent that Lessee demonstrates to the satisfaction of Aviation Department Management by written notice setting forth Lessee's explanation as to why the Hazardous Material originated from (1) a discharge, disposal or release outside of the Premises, unless such discharge, disposal or release was caused by Lessee, Lessee's agents, employees, contractors or invitees ; (2) a discharge, disposal or release of Hazardous Material on the Premises prior to the date upon which Lessee first occupied the premises and not caused by Lessee or Lessee's agents, employees, contractors, invitees or Trespassers; or (3) a discharge, disposal or release caused by the County, its agents, employees, contractors or any third party. Should the Aviation

Department determine that such a demonstration has not been made to Aviation Department's satisfaction, Lessee may invoke the dispute resolution provision of 8.16. Until such time as the parties reach an agreement or such time as the dispute is otherwise resolved, responsibility for such Hazardous Material shall remain with Lessee.

B) County's responsibility for Remediation under this Article 8.04 shall be limited to the Recognized Environmental Conditions required to be remediated under applicable Environmental Requirements. If County is permitted to leave any Hazardous Material in place under applicable Environmental Requirements, County shall have the option of so doing, unless a governmental authority requires at any time the removal of Hazardous Materials for Lessee to be able to continue with construction or occupancy of the Premises. The County shall notify Lessee of any such decision to leave Hazardous Material in place.

Listing of contamination assessment reports and remedial action.

(1) To the extent they exist, the County has made available to Lessee a listing of contamination assessment reports and remedial action plans regarding any soil and groundwater contamination at the Premises. Such list is found on Exhibit K to this Agreement. The County may have already installed or may have plans to install remediation systems to clean up the contamination described in such reports to the extent they exist. Lessee agrees that during the term of the Agreement, County's authorized representatives shall have the right to enter the Premises in order to operate, inspect, maintain, relocate and replace any such installed systems. Without limiting the generality of the foregoing, the County shall have the right to: (a) install, use, monitor, remove (or, in connection with monitoring wells, abandon in place in accordance with applicable governmental regulations) soil borings, treatment systems, pumps, monitoring wells, and associated equipment; (b) construct, maintain, and ultimately remove various mechanical devices designed to aid in the monitoring and remediating effort; and (c) undertake such related activities as the Aviation Department or other governmental authorities may require or recommend, utilizing such methods as the Aviation Department or the applicable governmental authorities may elect in order to remediate the contamination described in any such reports.

(2) County shall utilize reasonable efforts to minimize any disturbance of the Lessee's use of the Premises caused by any Remediation it undertakes and shall provide Lessee prior written notice of such Remediation. Lessee agrees that it shall not unreasonably interfere with or obstruct such Remediation. County and Lessee each agree to take such action as may be reasonable to coordinate their operations so as to minimize any interference with the other party. If vehicles, equipment, or materials belonging to the Lessee have to be temporarily relocated to permit the Remediation to be performed, the Lessee will effect such relocation at no expense to the County. Attached to Exhibit K is a site sketch of the Premises describing any existing or currently planned Remediation equipment and depicting the current and proposed future location of such equipment.

(3) If Remediation equipment or materials need to be temporarily stored in a secure location on the Premises, the Lessee will provide reasonable storage inside the building on the Premises for such equipment and materials at no expense to the County, provided, however, that Lessee shall bear no liability and otherwise shall have no responsibility for any theft of and/or damage to such equipment or materials so stored, to the extent Lessee took reasonable measures to prevent, such theft and/or damage and such theft and/or damage was not caused by Lessee or Lessee's employees. To the extent that water and electrical service within the

Premises are not metered and the Lessee does not pay for such services directly, the Lessee will provide the County with water and electrical service from the Premises in connection with the Remediation, without charge. The Lessee acknowledges the Remediation may be conducted at the locations depicted on the site sketch attached to Exhibit K at any time during the term of the Agreement and may continue until such time as a no further action letter is obtained from the appropriate regulatory authorities.

8.05 <u>Baseline Audit</u>:

The County has provided Lessee with a copy of an environmental audit of the Premises, conducted to identify any Recognized Environmental Conditions associated with the Premises, which audit may include analyses of soil and groundwater samples (the initial "Baseline Audit"). Except to the extent Lessee previously occupied the Premises, the County shall be responsible for any Recognized Environmental Conditions within the meaning of ASTM E 1527-05, or most recent version, disclosed by the Baseline Audit. Except to the extent Lessee previously occupied the Premises, Lessee may terminate this Agreement within sixty (60) days of receipt of the Baseline Audit if Lessee, in its sole discretion, determines that the Recognized Environmental Conditions disclosed in such Baseline Audit are unacceptable. To the extent Lessee previously occupied the Premises, Lessee, subject to its right to invoke the dispute resolution provision of 8.16, shall be responsible for all Recognized Environmental Conditions disclosed in the Baseline Audit, which are not otherwise Baseline Environmental Conditions, unless Lessee demonstrates to the County's satisfaction that the Recognized Environmental Conditions originated from (1) a discharge, disposal or release outside of the Premises, unless such discharge, disposal or release was caused by Lessee, Lessee's agents employees, contractors or invitees; or (2) a discharge, disposal or release of Hazardous Material on the Premises prior to Lessee's first occupancy of the Premises and not caused by Lessee, Lessee's agents, employees, contractors, invitees, or Trespassers.

8.06 Lessee Audit:

Lessee, at its sole cost and expense, shall have the right to conduct, within sixty (60) days from the receipt of the Baseline Audit, an environmental inspection of the Premises (the "Lessee Audit"), through an independent environmental consultant approved in writing by County, such approval not to be unreasonably withheld or delayed. If Lessee elects to conduct a Lessee Audit, it shall furnish County a copy of the Lessee Audit within thirty (30) days of Lessee's receipt of the Lessee Audit. The purpose of the Lessee Audit is to determine whether there are present on the Premises any Recognized Environmental Conditions not identified in the Baseline Audit, any previous audits, or any contamination assessment reports or remedial action plans, to the extent any such documents exist. Within thirty (30) days of receipt of such Lessee Audit, the County shall notify Lessee if it disputes the Recognized Environmental Conditions or the delineation of any subsurface conditions described in the Lessee If the Lessee Audit reveals any Recognized Environmental Conditions or Audit. delineates any subsurface contamination not disclosed in any contamination assessment reports, remedial action plans, or the Baseline Audit, and which are not otherwise considered Baseline Environmental Conditions under the terms of this Agreement, then, except to the extent that Lessee previously occupied the Premises, the County, at its option, shall: (i) allow Lessee to terminate the Agreement, without penalty, within sixty (60) days of receipt of such notice of dispute from the County; or (ii) notify Lessee that it has agreed to be responsible for such Recognized Environmental Conditions and delineated subsurface contamination to the same extent as the County is responsible for the Recognized Environmental Conditions and subsurface contamination disclosed in any contamination assessment reports, remedial action

plans and the Baseline Audit. If the County allows Lessee to terminate the Agreement and Lessee elects not to terminate. Lessee's failure to terminate shall constitute a waiver of 1) Lessee's rights to terminate its obligations under this Agreement as to any findings in such Lessee Audit, except as to its right to cancel the lease on thirty (30) days' notice under Article 1.01 (B) and, 2) as provided in Article 8.04, any claim it may have against the County with respect either to Recognized Environmental Conditions and subsurface contamination disclosed in such Lessee Audit. To the extent the Lessee previously occupied the Premises, Lessee shall be responsible for all Recognized Environmental Conditions disclosed in the Lessee Audit that are not Baseline Environmental Conditions unless Lessee demonstrates to the satisfaction of Aviation Department Management by written notice setting forth Lessee's explanation why the Recognized Environmental Conditions originated from (1) a discharge, disposal or release outside of the Premises, unless such discharge, disposal or release was caused by Lessee, Lessee's agents, employees, contractors, or invitees; (2) a discharge, disposal or release of Hazardous Material on the Premises prior to the Occupancy Date and not caused by Lessee or Lessee's agents, employees, contractors, invitees or Trespassers ; or (3) a discharge, disposal or release caused by Should the Aviation Department determine that such a County or third party. demonstration has not been made to Aviation Department's satisfaction, Lessee may invoke the dispute resolution provision of 8.16. Until such time as the parties reach an agreement or until such time as the dispute is otherwise resolved, responsibility for such Recognized Environmental Condition shall remain with Lessee.

8.07 Environmental Maintenance of Premises:

Except for the obligations of the County under this Article 8, Lessee shall, at its sole cost and expense, keep, maintain and use the Premises, and operate within the Premises at all times, in compliance with all applicable Environmental Laws, and shall maintain the Premises in good and sanitary order, condition, and repair.

8.08 Lessee's Use of Hazardous Materials:

Exhibit K-08 is a complete list of all Hazardous Materials which Lessee currently intends to use on the Premises or Other Airport Property during the term of the Agreement which have been approved by the County, and the use, storage and transportation of which on or about the Premises shall not be subject to County's approval or objections. Except for those Hazardous Materials listed on Exhibit K-08, Lessee shall not use, store, generate, treat, transport, or dispose of any Hazardous Material on the Premises or Other Airport Property without first providing the County thirty (30) days written notice prior to bringing such Hazardous Material upon the premises. To the extent certain Hazardous Materials are be needed to be used by Lessee on a non-routine basis, such as for emergency repairs, Lessee may provide such notice within twenty-four (24) hours of bringing such Hazardous Material upon the premises. Notwithstanding the foregoing, County may object to the use of any previously approved Hazardous Material should County reasonably determine that the continued use of the Hazardous Material by Lessee presents a material increased risk of site contamination, damage or injury to persons, Premises, resources on or near the Premises or Other Airport Property, or noncompliance due to a change in regulation of such Hazardous Material under applicable Environmental Law. Upon County's objection, Lessee shall immediately remove the Hazardous Material from the site. This section 8.08 shall not apply to Hazardous Materials which are not used, generated, treated or disposed of by Lessee but which are otherwise transported by Lessee solely in the course of Lessee's business, such as cargo operations, and for which Lessee has no knowledge as to the identity of such hazardous materials prior to such transport. County's objection or failure to object to the use, storage, generation, treatment, transportation, or disposal of Hazardous Material under this paragraph, or the exclusion
of certain Hazardous Materials under this paragraph, shall not limit or affect Lessee's obligations under this Agreement, including Lessee's duty to remedy or remove releases or threatened releases; to comply with applicable Environmental Law and/or Environmental Requirements relating to the use, storage, generation, treatment, transportation, and/or disposal of any such Hazardous Materials; or to indemnify County against any harm or damage caused thereby. Lessee shall promptly and completely answer periodic questionnaires from the County concerning Lessee's practices regarding the generation, use, storage, and disposal of Hazardous Materials under this Agreement.

- 8.09 Entry by County:
- A) Notwithstanding any other right of entry granted to County under this Agreement, and subject to the requirements set forth in Article 8.09(B), MDAD shall have the right, at its own expense and upon reasonable notice, to enter the Premises or to have consultants enter the Premises throughout the Term of this Agreement for the purposes of: (1) determining whether the Premises are in conformity with applicable Environmental Law; (2) conducting an environmental review or investigation of the Premises; (3) determining whether Lessee has complied with the applicable environmental requirements of this Agreement; (4) determining the corrective measures, if any, required of Lessee to ensure the safe use, storage, and disposal of Hazardous Materials; or (5) removing Hazardous Materials (except to the extent used, stored, generated, treated, transported, or disposed of by Lessee in compliance with applicable Environmental Requirements and the terms of this Agreement). Lessee agrees to provide access and reasonable assistance for such inspections. MDAD shall use its best efforts to reasonably minimize interruptions of business operations on the Premises.
- B) Such inspections may include, but are not limited to, entering the Premises or adjacent property with drill rigs or other machinery for the purpose of obtaining laboratory samples of environmental conditions or soil or groundwater conditions. Lessee shall have the right to collect split samples of any samples collected by MDAD. MDAD shall not be limited in the number of such inspections during the Term of this Agreement. MDAD will conduct such inspections during Lessee's normal business hours, but MDAD may conduct such inspections in other than normal business hours if the circumstances so require. For inspections conducted by MDAD, MDAD agrees to provide Lessee with reasonable notice (not less than twenty four (24) hours) prior to inspecting the Premises; provided however, that such notice period shall not apply under circumstances in which MDAD reasonably determines that there exists an immediate threat to the health, safety, or welfare of any persons. Based on the results of such inspections, should MDAD reasonably determine that Hazardous Materials have been released, discharged, stored, or used on the Premises in violation of the terms of this Agreement, Lessee shall, in a timely manner, at its expense, remove such Hazardous Materials in a manner not inconsistent with applicable Environmental Law and otherwise comply with the reasonable recommendations of MDAD and any regulatory authorities related to the The right granted to MDAD herein to inspect the results of such inspections. Premises shall not create a duty on MDAD's part to inspect the Premises, nor liability of MDAD for Lessee's use, storage, or disposal of Hazardous Materials, it being understood that Lessee shall be solely responsible for all liability in connection therewith. MDAD shall provide the results of such inspections to the Lessee in a timely manner if requested to do so in writing. Nothing herein shall be construed to limit, restrain, impair or interfere with County's regulatory authority to conduct inspections and/or the manner in which it conducts such inspections. Lessee shall not be liable or otherwise responsible for any property damage to the Premises or

injury to any person caused by County, its agents or consultants during County's inspection under this Section 8.09.

8.09 <u>Permits and Licenses</u>:

The Lessee warrants that it will secure at the times required by issuing authorities all applicable permits or approvals that are required by any governmental authority having lawful jurisdiction to enable Lessee to conduct its obligations under this Agreement. Upon written request, Lessee shall provide to County copies of all permits, licenses, certificates of occupancy, approvals, consent orders, or other authorizations issued to Lessee under applicable Environmental Requirements, as they pertain to the Lessee's operations on or use of the Premises or Other Airport Property.

8.11 Notice of Discharge to County:

- A) In the event of: (i) the happening of any material event involving the spill, release, leak, seepage, discharge or clean up of any Hazardous Material on the Premises or Other Airport Property in connection with Lessee's operation thereon; or (ii) any written Environmental Claim affecting Lessee from any person or entity resulting from Lessee's use of the Premises or Other Airport Property, then Lessee shall immediately notify County orally within twenty-four (24) hours and in writing within three (3) business days of said notice. If County is reasonably satisfied that Lessee is not promptly commencing the response to either of such events, County shall have the right but not the obligation to enter onto the Premises or to take such other actions as it shall deem reasonably necessary or advisable to clean up, remove, resolve or minimize the impact of or otherwise deal with any such Hazardous Material or Environmental Claim following receipt of any notice from any person or any entity having jurisdiction asserting the existence of any Hazardous Material or an Environmental Claim pertaining to the Premises, which if true, could result in an order, suit or other action against the County. If Lessee is unable to resolve such action in a manner which results in no liability on the part of County, all reasonable costs and expenses incurred by County shall be deemed additional rent due County under this Agreement and shall be payable by Lessee upon demand, except to the extent they relate to a Baseline Environmental Condition.
- B) With regard to any reporting obligation arising out of Lessee's operations or during the Agreement, Lessee shall timely notify the State of Florida Department of Environmental Protection, Miami-Dade County Department of Environmental Resources Management, and the United States Environmental Protection Agency, as appropriate, with regard to any and all applicable reporting obligations while simultaneously providing written notice to County.
- C) Within sixty (60) days of execution of this Agreement, Lessee shall submit to County an emergency action plan/contingency plan setting forth in detail Lessee's procedures for responding to spills, releases, or discharges of Hazardous Materials. The emergency action plan/contingency plan shall identify Lessee's emergency response coordinator and Lessee's emergency response contractor.

8.12 <u>Reports to County</u>:

For any year in which any Hazardous Materials have been used, generated, treated, stored, transported or otherwise been present on or in the Premises, (or on or in other Airport property for purposes related to Lessee's operations on the Premises), Lessee shall provide County with a written report listing: the Hazardous Materials which were present on the Premises or other Airport property; all releases of Hazardous Material that occurred or were discovered on the Premises or other Airport property and

which were required to be reported to regulatory authorities under applicable Environmental Laws; all enforcement actions related to such Hazardous Materials, including all consent agreements or other non-privileged documents relating to such enforcement actions during that time period. In addition, Lessee shall provide County with copies of any reports filed in accordance with the Emergency Planning and Community Right to Know Act (EPCRA) and shall make available for review upon request by County copies of all manifests for hazardous wastes generated from operations on the Premises. Lessee shall provide the report required under this section to the County by April 1 of each year for the preceding calendar year.

8.13 Periodic Environmental Audits:

Lessee shall establish and maintain, at its sole expense, a system to assure and monitor its continued compliance on the Premises with all applicable Environmental Laws, which system shall include, no less than once each year a detailed review of such compliance (the "Environmental Audit") by such consultant or consultants as County may approve, which approval shall not be unreasonably withheld, delayed or conditioned. Alternatively, if the MDAD approves, which approval shall not be unreasonably withheld, delayed, or conditioned, such Environmental Audit may be conducted by Lessee's personnel but in either case Lessee shall provide County with a copy or summary of its report of its annual Environmental Audit, which shall be consistent with ASTM's "Practice for Environmental Regulatory Compliance Audits". If the Environmental Audit indicates any unresolved violation of any applicable Environmental Law and/or Environmental Requirements, Lessee shall, at the request of County, provide a detailed review of the status of any such violation within thirty (30) days of the County's request.

8.14 <u>Remediation of Hazardous Material Release</u>:

If Lessee or Lessee's agents, employees, contractors, invitees or trespassers cause any Hazardous Materials to be released, discharged, or otherwise located on or about the Premises or Other Airport Property during the term of this Agreement ("Hazardous Material Release"), Lessee shall promptly take all actions, at its sole expense and without abatement of rent, as are reasonable and necessary to return the affected portion of the Premises or Other Airport Property and any other affected soil or groundwater to their condition existing prior to the Hazardous Material Release in a manner not inconsistent with applicable Environmental Law. County shall have the right to approve all such remedial work, including, without limitation: (i) the selection of any contractor or consultant Lessee proposes to retain to investigate the nature or extent of such Hazardous Material Release or to perform any such remedial work; (ii) any reports or disclosure statements to be submitted to any governmental authorities prior to the submission of such materials; and (iii) any proposed remediation plan or any material revision thereto prior to submission to any governmental authorities. The County's approvals shall not be unreasonably withheld, conditioned or delayed. Notwithstanding the foregoing, County's prior consent shall not be necessary if a Hazardous Material Release poses an immediate threat to the health, safety, or welfare of any persons and, despite Lessee's best efforts, it is not practicable to obtain County's consent before taking remedial action to abate such immediate threat, provided that: (a) Lessee shall notify County as soon as possible and shall thereafter obtain County's consent as otherwise provided in this paragraph; and (b) Lessee shall take only such action as may be necessary or appropriate to abate such immediate threat and shall otherwise comply with the provisions of this paragraph. In addition to any rights reserved by County in this Agreement, County shall have the right, but not the obligation, to participate with Lessee, Lessee's consultants and Lessee's contractors in any meetings with representatives of the governmental authorities and Lessee shall provide County reasonable notice of any such meetings. All remedial work shall be performed in

compliance with all applicable Environmental Laws. The County's consent to any remedial activities undertaken by Lessee shall not be withheld so long as County reasonably determines that such activities will not cause any material adverse long-term or short-term effect on the Premises, or other adjoining property owned by County. Lessee's obligations in this section do not apply to Baseline Environmental Conditions.

8.15 Indemnity:

Lessee shall indemnify, defend (with counsel reasonably satisfactory to County), and hold County, its directors, officers, employees, agents, assigns, and any successors to County's interest in the Premises, harmless from and against any and all loss, cost, damage, expense (including reasonable attorneys' fees), claim, cause of action, judgment, penalty, fine, or liability, directly or indirectly, relating to or arising from the use, storage, release, discharge, handling, or presence of Hazardous Materials on. under, or about the Premises or Other Airport Property and caused by Lessee, Lessee's agents, employees, contractors, invitees or trespassers. This indemnification shall include without limitation: (a) personal injury claims; (b) the payment of liens; (c) diminution in the value of the Premises or Other Airport Property; (d) damages for the loss or restriction on use of the Premises or Other Airport Property; (e) sums paid in settlement of claims; (f) reasonable attorneys' fees, consulting fees, and expert fees, (g) the cost of any investigation of site conditions, and (h) the cost of any repair, cleanup, remedial, removal, or restoration work or detoxification if required by any governmental authorities or deemed necessary in County's reasonable judgment, but shall not extend to such claims, payment, diminution, damages, sums, fees or costs to the extent caused (i) solely by an act of God or (ii) by the negligent or willful misconduct of the County, its officers, employees, contractors or agents. For any legal proceedings or actions initiated in connection with the Hazardous Materials Release, County shall have the right at its expense but not the obligation to join and participate in such proceedings or actions in which the County is a named party, and control that portion of the proceedings in which it is a named party. County may also negotiate, defend, approve, and appeal any action in which County is named as a party taken or issued by any applicable governmental authorities with regard to a Hazardous Materials Release; provided, however, claims for which Lessee may be liable pursuant to this Article 8.15 shall not be settled without Lessee's consent. Any costs or expenses incurred by County for which Lessee is responsible under this paragraph or for which Lessee has indemnified County: (i) shall be paid to County on demand, during the term of this Agreement as additional rent; and (ii) from and after the expiration or earlier termination of the Agreement shall be reimbursed by Lessee on demand. Lessee's obligations pursuant to the foregoing indemnity shall survive the expiration or termination of this Agreement and shall bind Lessee's successors and assignees and inure to the benefit of County's successors and assignees. Notwithstanding any other provision of this Agreement, this section 8.15 does not apply to Baseline Environmental Conditions or a discharge, disposal or release caused by the County, its officers, employees, contractors or agents.

- A) This indemnity specifically includes the direct obligation of Lessee to perform, at its sole cost and expense, any remedial or other activities required or ordered by a court or agency having competent jurisdiction over the subject matter, or otherwise necessary to avoid or minimize injury or liability to any person, or to prevent the spread of Hazardous Materials.
- B) In addition, because costs of Hazardous Materials discharges are passed on to airlines through the residual rate charging mechanism at Miami International Airport and the charges to Lessee will be increased to offset such costs, Lessee agrees in order to minimize its obligations in this regard to use best efforts to assist the Aviation Department in responding to Hazardous

Materials spills in or Airport property reasonably close to the Premises used by Lessee by making Lessee's remediation equipment and personnel available for such emergency remediation activity. However, Lessee may provide such assistance only at the direct request of the Aviation Department and only if Lessee's remediation equipment is intended to be utilized for the Hazardous Material spill at issue and only if Lessee's personnel have been trained to respond to the Hazardous Material spill at issue. If Lessee is directed to perform any remedial work under this Article 8.15(B) for which it is later determined that Lessee is not responsible, the Aviation Department shall reimburse Lessee for all costs associated with or arising out of Lessee's performance of such remedial work. Lessee shall cooperate with the Aviation Department in any subsequent effort by the Aviation Department to recover from the responsible parties all costs involved with the remediation effort that utilized Lessee's equipment and personnel. Lessee shall perform all such work in its own name in accordance with applicable laws Lessee acknowledges that the County's regulatory power in this regard is independent of the County's contractual undertakings herein, and nothing herein shall affect the County's right in its regulatory capacity to impose its environmental rules, regulations, and authorities upon the Lessee in accordance with the law.

- C) In the event Lessee fails to perform its obligations in Article 8.15(A) above, and without waiving its rights hereunder, County may, at its option, perform such remedial work as described in Article 8.15(A) above, and thereafter seek reimbursement for the costs thereof. In accordance with this Article 8, Lessee shall permit County or its designated representative access to the Premises areas to perform such remedial activities.
- D) Whenever County has incurred costs described in this section as a result of the failure of Lessee to perform its obligations hereunder, Lessee shall, within thirty (30) days of receipt of notice thereof, reimburse County for all such expenses together with interest at the rate of 1 ½ % per month on the outstanding balance commencing on the thirty-first date following Lessee's receipt of such notice until the date of payment.
- E) To the extent of Lessee's responsibility under this Article and without limiting its obligations under any other paragraph of this Agreement, and except to the extent of County's responsibility for environmental conditions set forth in this Article 8, Lessee shall be solely and completely responsible for responding to and complying with any administrative notice, order, request or demand, or any third party claim or demand relating to potential or actual Hazardous Materials contamination on the Premise. Lessee's responsibility under this paragraph includes but is not limited to responding to such orders on behalf of County and defending against any assertion of County's financial responsibility or individual duty to perform under such orders. Lessee shall assume, pursuant to the Indemnity provision set forth in this Article 8, any liabilities or responsibilities which are assessed against County in any action described under this paragraph.

8.16 <u>Dispute Resolution</u>:

County and Lessee agree that any dispute between them relating to this Article 8 will first be submitted, by written notice, to a designated representative of both County and Lessee who will meet at County's place of business or other mutually agreeable location, or by teleconference, and confer in an effort to resolve such dispute. Any decision of the representatives will be final and binding on the parties. In the event the representatives are unable to resolve any dispute within ten (10) days after submission

to them, either party may refer the dispute to mediation, or institute any other available legal or equitable proceeding in order to resolve the dispute.

8.17 <u>Waiver and Release</u>:

Lessee, on behalf of itself and its heirs, successors and assigns, hereby waives, releases, acquits and forever discharges County, its principals, officers, directors, employees, agents, representatives and any other person acting on behalf of the County, and the successors and assigns of any of the preceding, of and from any and all claims, actions, causes of action, demands, rights, damages, costs, expenses or compensation whatsoever, direct or indirect, known or unknown, foreseen or unforeseen, which Lessee or any its heirs, successors, or assigns now has or which may arise in the future on account of or in any way related to or in connection with any past, present or future physical characteristic or condition of the Premises , including, without limitation, any Hazardous Material, in, at, on, under or related to the Premises , or any violation or potential violation of any Environmental Law applicable thereto; provided, however, this Article 8.17 shall not constitute a waiver or release of any obligation of County under this Article 8. Lessee acknowledges that County would not enter into this Agreement without Lessee's agreement to the waiver and release provided herein.

8.18 No Waiver of Rights, Causes of Actions or Defenses:

Notwithstanding any language in this Agreement, including without limitation Articles 8.03, 8.04, 8.05, 8.06, 8.14, 8.15 and 8.16, Lessee does not agree to waive or release any rights, causes of action or defenses it may have against Miami-Dade County or any other party related to allegations made by the County in (i) Case No. 01-8758 CA 25 which has been filed by the County in the Florida Circuit Court of the Eleventh Judicial Circuit, and (ii) a letter dated April 9, 2001, to Lessee and others (who are referred to as "responsible parties" or "RPs"). Nothing herein shall be construed to limit or expand upon any releases previously granted to or exchanged between the parties as a result of judgments or settlements obtained in proceedings between the parties, including, without limitation, settlements in bankruptcy or settlements entered under Case No. 01-8758 CA 25 which has been filed by the County in the Florida Circuit Court of the parties in bankruptcy or settlements entered under Case No. 01-8758 CA 25 which has been filed by the County in the Florida Circuit Court of the parties in bankruptcy or settlements entered under Case No. 01-8758 CA 25 which has been filed by the County in the Florida Circuit Court of the Eleventh Judicial Circuit.

8.19 <u>Surrender of Premises</u>:

Lessee shall surrender the Premises used by Lessee to County upon the expiration or earlier termination of this Agreement free of debris, waste, and Hazardous Materials used, stored, or disposed of by Lessee or its agents, employees, contractors, invitees or Trespassers, or otherwise discharged on the Premises or Other Airport Property for which Lessee is responsible during the term of this Agreement. The Premises shall be surrendered in a condition that complies with all applicable Environmental Requirements, and such other reasonable environmental requirements as may be imposed by County. Lessee shall not be responsible under this section 8.19 to the extent of County's obligations under this Article 8.

8.20 Breach:

Any breach by Lessee of any provision of this Article 8 shall, after notice and a reasonable opportunity for Lessee to cure, constitute a default of the Agreement and shall entitle County to exercise any and all remedies provided in the Agreement, or as otherwise permitted by law.

8.21 <u>Survivability of Terms</u>:

The terms and conditions of this Article 8, including the indemnity, waiver, and release, shall survive the termination of this Agreement.

8.22 <u>Right to Regulate</u>:

As provided for in Article 20.12 of this Agreement, nothing within this Article 8 shall be construed to waive or limit, restrain, impair or interfere with the County's regulatory authority.

ARTICLE 9 Indemnification and Hold Harmless

Lessee shall indemnify and hold harmless the County and its officers, employees, agents and instrumentalities from any and all liability, losses or damages, including reasonable attorneys' fees and costs of defense, which the County or its officers, employees, agents or instrumentalities may incur as a result of claims, demands, suits, causes of actions or proceedings of any kind or nature arising out of, relating to or resulting from the performance of this Agreement or the use of the Premises by the Lessee, its employees, agents, servants, partners, contractors, subcontractors, invitees, or trespassers except to the extent such losses or damages are caused by the negligence or willful misconduct of the County, its officers, employees or agents. Lessee shall pay all claims and losses in connection therewith and shall investigate and defend all claims, suits or actions of any kind or nature in the name of the County, where applicable, including appellate proceedings, and shall pay costs, judgments and attorney's fees which may issue thereon. Lessee expressly understands and agrees that any insurance protection required by this Agreement or otherwise provided by Lessee shall in no way limit the responsibility to indemnify, keep and save harmless and defend the County or its officers, employees, agents and instrumentalities as herein provided.

ARTICLE 10

Assignment and Subletting

The Lessee shall not assign, transfer, pledge, or otherwise encumber this Agreement without the express prior written consent of the Department, provided, however, this Agreement may be assigned to any successor airline resulting from a merger with or acquisition of Lessee. Further, the Lessee shall not sublet all or any portion of the Premises, nor allow others to use the Premises without the express prior written consent of the Department.

ARTICLE 11

<u>Insurance</u>

11.01 <u>Insurance Required</u>: In addition to such insurance as may be required by law, the Lessee shall maintain, without lapse or material change, for so long as it occupies the Premises, the following insurance:

A) Commercial General Liability Insurance on a comprehensive basis, including Contractual Liability, to cover the Lessee's Premises and operations, in an amount not less than \$5,000,000 combined single limit per occurrence for bodily injury and property damage. The County must be shown as an additional insured with respect to this coverage.

- B) Automobile Liability Insurance covering all owned, non-owned and hired vehicles used by the Lessee in connection with its operations under this Agreement in an amount not less than:
 - (1) \$5,000,000 combined single limit per occurrence for bodily injury and property damage.
 - (2) \$300,000 combined single limit per occurrence for bodily injury and property damage covering vehicles when being used by the Lessee off of the AOA.
- C) Pollution Legal Liability Insurance for Lessee's Premises and other airport property in an amount not less than \$2,000,000 per pollution incident and in the aggregate covering site assessment, site clean-up, third party claims, including, but not limited to governmental claims, legal defense costs, charges and expenses arising from any on-site and off-site loss, damage, expense or claim related to the release or any threatened release of Hazardous Material.
- D) Workers' Compensation as required by Chapter 440, Florida Statutes.

The insurance coverages required shall include those classifications, as listed in standard liability insurance manuals, which most nearly reflect the operations of the Lessee under this Agreement. All insurance policies required pursuant to the terms of this Agreement shall be issued in companies approved to do business under the laws of the State of Florida. Such companies must be rated no less than "A-" as to management, and no less than "VII" as to strength in accordance with the latest edition of "Best's Insurance Guide", published by A.M. Best Company, Inc., or its equivalent, subject to approval of the County Risk Management Division.

11.02 Insurance Certificates Required:

Prior to the commencement of operations hereunder and annually thereafter, the Lessee shall furnish or cause to be furnished certificates of insurance to the Department which certificates shall clearly indicate that:

- (A) The Lessee has obtained insurance in the types, amounts and classifications as required for strict compliance with this Article;
- (B) The policy cancellation notification provisions specify at least 30 days advance written notice of cancellation to the County; and
- (C) The County is named as an additional insured with respect to the Lessee's public liability policies.

On said insurance certificates, unless specifically shown to be excluded thereon, comprehensive public liability coverage shall include contractual liability, and notification of cancellation shall include notification of material changes in the policies.

The County reserves the right to require the Lessee to provide such reasonably amended insurance coverage as it deems necessary or desirable, upon issuance of notice in writing to the Lessee, which notice shall automatically amend this Agreement effective 30 days after such notice.

11.03 Compliance:

Compliance with the requirements of this Article 11 (Insurance) shall not relieve the Lessee of its liability under any other portion of this Agreement or any other agreement between the County and the Lessee.

11.04 Right to Examine:

The Department reserves the right, upon reasonable notice, to examine the original or true copies of policies of insurance (including but not limited to binders, amendments, exclusions, riders and applications) to determine the true extent of coverage. The Lessee agrees to permit such inspection at the offices of the Department.

11.05 Personal Property:

Any personal property of the Lessee or of others placed in the Premises and Airport shall be at the sole risk of the Lessee or the owners thereof, and the County shall not be liable for any loss or damage, except to the extent such loss or damage was caused by the sole active negligence of the County, as limited by Section 768.28, Florida Statutes.

ARTICLE 12 Use of Public Facilities

The County grants to the Lessee, in common with all others desiring to use The Airport and only to the extent necessary or reasonably desirable, so long as such use does not conflict with the County's operation of the Airport, to carry out the rights granted the Lessee hereunder, the nonexclusive privilege to use the roads of egress and ingress, service roads and such other facilities and improvements as may be now in existence or hereafter constructed for the use of persons lawfully using The Airport; provided, however, that such usage shall be subject to the payment of nondiscriminatory fees and other charges established by the County. Nothing herein contained shall grant to the Lessee the right to use any leasable space or area improved or unimproved which is leased to a third party, or which the County has not leased herein.

ARTICLE 13

<u>Termination</u>

13.01 Payment Defaults:

Failure of the Lessee to make all payments of rentals, fees and charges required to be paid herein when due shall constitute a default, and the County may, at its option terminate this Agreement after seven calendar days notice in writing to the Lessee unless the default be cured within the notice period.

13.02 Insurance Defaults:

The County shall have the right, upon seven calendar days written notice to the Lessee, to terminate this Agreement if the Lessee fails to provide evidence of insurance coverage in strict compliance with Article 11 hereof prior to commencement of operations, or fails to provide a renewal of said evidence upon its expiration; provided, however, that such termination shall not be effective if the Lessee provides the required evidence of insurance coverage within the notice period.

13.03 Other Defaults:

The County shall have the right, upon 30 calendar days written notice to the Lessee, to terminate this Agreement upon the occurrence of any one or more of the

following, unless the same shall have been corrected within such period, or, if correction cannot reasonably be completed within such 30-day period, in the sole discretion of the Department, the Lessee has commenced substantial corrective steps within such 30-day period and diligently pursues same to completion:

- (A) Failure of the Lessee to comply with any covenants of this Agreement, other than the covenants to pay rentals, fees and charges when due, and the covenants to provide required evidence of insurance coverage.
- (B) The conduct of any business, the performance of any service, or the merchandising of any product or service not specifically authorized herein, by the Lessee.
- (C) Failure of the Lessee to comply with any Environmental Law or Environmental Requirement as those terms are defined in Article 8.01 of this Agreement.

13.04 <u>Habitual Default</u>:

Notwithstanding the foregoing, in the event that the Lessee has frequently, regularly or repetitively defaulted in the performance of or breached any of the terms, covenants and conditions required herein to be kept and performed by the Lessee, in the sole opinion of the County and regardless of whether the Lessee has cured each individual condition of breach or default as provided in Articles 13.01 (Payment Defaults), 13.02 (Insurance Defaults) and 13.03 (Other Defaults) hereinabove, the Lessee shall be determined by the Director to be an "habitual violator." At the time that such determination is made, the Department shall issue to the Lessee a written notice advising of such determination and citing the circumstances therefor. Such notice shall also advise the Lessee that there shall be no further notice or grace periods to correct any subsequent breach(es) or default(s) and that any subsequent breach(es) or default(s), of whatever nature, taken with all previous breaches and defaults, shall be considered cumulative and, collectively, shall constitute a condition of non-curable default and grounds for immediate termination of this Agreement. In the event of any such subsequent breach or default, the County may cancel this Agreement upon the giving of written notice of termination to the Lessee, such termination to be effective upon the tenth day following the date of receipt thereof and all payments due hereunder shall be payable to said date, and the Lessee shall have no further rights hereunder.

13.05 Termination by Abandonment:

This Agreement shall be automatically terminated upon the abandonment by the Lessee of the Premises or the voluntary discontinuance of operations at the Airport for any period of time exceeding 15 consecutive calendar days, unless such abandonment or discontinuance has been caused by strike, labor disturbance, acts of God, civil disturbance or governmental order that prevents the Lessee's use of the Premises for the purposes authorized in Article 2 (Use of Premises) hereof. Such termination shall not relieve the Lessee of its rental payment obligation for the remaining term of the agreement nor does it constitute a waiver by the Lessor of its rights to recover damages for rental payments for the remaining term of the agreement and loss of future rentals.

13.06 Actions at Termination:

The Lessee shall vacate, quit, surrender up and deliver the Premises to the County on or before the termination date of this Agreement, whether by lapse of time or otherwise. The Lessee shall surrender the Premises in the condition required under Article 4.03 (Maintenance and Repairs) herein. All repairs for which the Lessee is responsible shall be completed prior to surrender. The Lessee shall deliver to the

Department all keys to the Premises upon surrender. On or before the termination date of this Agreement, except in the instance of termination pursuant to Article 13.05 (Termination by Abandonment), in which event the Lessee shall be allowed up to five calendar days from date of termination, and provided that the Lessee is not in default in the payment of any rentals, fees or charges required to be paid herein, the Lessee shall remove all of its personal property from the Premises. Any personal property of the Lessee not removed in accordance with this Article may be re-

moved by the Department for storage at the cost of the Lessee. Failure on the part of the Lessee to reclaim its personal property within 30 days from the date of termination shall constitute a gratuitous transfer of title thereof to the County for whatever disposition is deemed to be in the best interest of the County.

The Lessee shall, at its expense, take all actions required by Federal, State and local laws, regulations or codes to remove from the Premises any hazardous substance or environmental contaminant, whether stored in drums, or found in vats, containers, distribution pipe lines, or the like. All such substances and contaminants shall be removed by the Lessee in a manner approved and authorized by such Federal, State or local laws, regulations or codes.

If the County advises the Lessee that it has reasonable grounds to believe that any hazardous substance or environmental contaminant has been released within the Premises or into the ground under the Premises, during the term of this Agreement or during the terms of any prior leases between the Lessee and the County for the same or substantially the same Premises, then the Lessee at its sole cost and expense shall retain an approved environmental consultant to perform whatever environmental assessment may be required to determine the extent of such release. The Lessee shall comply with the recommendations and conclusions, contingent upon County approval, of such consultant regarding environmental cleanup efforts that may be required, and shall comply with any other clean up requirements imposed on the Lessee by Federal, State or local law, regulations or codes.

13.07 Lien Upon Personal Property:

In the event of termination for default or upon termination of this Agreement by its term, the County shall have a lien upon all personal property of the Lessee to secure the payment of any unpaid rentals, fees and charges accruing under the terms of this Agreement.

13.08 Right to Show Premises:

At any time after the Lessee has been given notice of termination or default, pursuant to Article 13 (Termination) hereof, the County shall have the right to enter on the Premises for the purpose of showing the Premises to prospective tenants or users.

13.09 County Defaults:

This Agreement shall be subject to termination by the Lessee in the event of a default by the County in the performance of any covenant or agreement herein required to be performed by the County and the failure of the County to remedy same within a reasonable period of time following receipt of written notice from the Lessee of such default.

13.10 Other Terminations:

This Agreement shall be subject to termination by the County or the Lessee in the event of any one or more of the following:

- (A) The permanent abandonment of the Airport.
- (B) The lawful assumption by the United States Government or any authorized agency thereof, of the operation, control or use of the Airport, or any substantial part or parts thereof, in such a manner as to substantially restrict the Lessee from operating therefrom for a period in excess of 90 consecutive days, provided that nothing contained herein shall be deemed to constitute a waiver by the Lessee of any right it may have against the United States to just compensation in the event of any such assumption.
- (C) The issuance by any court of competent jurisdiction of any injunction in any way substantially preventing or restraining the use of the Airport, and the remaining in force of such injunction for a period in excess of 90 days.

ARTICLE 14 Special Conditions

14.01 <u>Quality of Services</u>:

The Lessee shall furnish the services required and authorized, pursuant to Article 2 (Use of Premises) hereof, on a good, prompt and efficient basis and on a fair, equal and not unjustly discriminatory basis to all users thereof.

14.02 Nondiscriminatory Prices:

The Lessee shall charge fair, reasonable, customary and not unjustly discriminatory prices for each unit of sale or service; provided, however, that the Lessee may make reasonable, customary and nondiscriminatory discounts, rebates or similar types of price reductions to volume purchasers of the Lessee's services.

14.03 County's Obligations:

The Lessee, in recognition of the County's obligation, pursuant to Section 22 of Part V of the Federal Aviation Administration's standard grant assurances, to enforce the provisions of Articles 14.01 (Quality of Services) and 14.02 (Nondiscriminatory Prices) above, agrees that the Department may, from time to time, promulgate standards, methods and procedures for and monitor and test the provision of services hereunder and may require the Lessee to provide copies of schedules of service charges and the bases for discounts, rebates and similar types of price reductions. Should the Department determine that the Lessee is not in compliance with the provisions of Articles 14.01 (Quality of Services) and 14.02 (Nondiscriminatory Prices) above, the first such occurrence shall be considered a curable default, pursuant to Article 13.03 (Other Defaults) hereof, and subsequent occurrence(s) shall be considered a material breach of this Agreement, entitling the County to the remedies provided in this Agreement or by law.

ARTICLE 15

Equal Employment Opportunity, Nondiscrimination and Affirmative Action

15.01 Equal Employment Opportunity:

In accordance with Title 14 Code of Federal Regulation (CFR) Part 152 (Affirmative Action Employment Program), the Lessee shall not discriminate against any employee or applicant for employment because of age, sex, race, color, religion, marital status, place of birth or national origin, ancestry, in accordance with the Americans with

Disabilities Act, discriminate against any otherwise qualified employees or applicants for employment with disabilities who can perform the essential functions of the job with or without reasonable accommodation. The Lessee shall take affirmative actions' to ensure that applicants are employed and that employees are treated during their employment without regard to age, sex, race, color, religion, marital status, place of birth or national origin, ancestry, or disability. Such actions include, but not limited to, the following: Employment, upgrading, transfer or demotion, recruitment, recruitment advertising, layoff or termination, rates of pay or other forms of compensation, selection for training including apprenticeship.

The Lessee agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided by the County setting forth the provisions of this Equal Employment Opportunity clause. The Lessee shall comply with all applicable provisions of the Civil Rights Act of 1964, Executive Order 11246 issued September 24, 1965, as amended by Executive Order 113155, revised order No. 4 issued December 1, 1951, as amended, and the Americans with Disabilities Act. The Age Discrimination in Employment Act effective June 12, 1968, Executive Order 13166 issued August 11, 2000, Improving Access to Services for persons with Limited English Proficient (LEP), the rules, regulations and relevant orders of the Secretary of Labor, Florida Statues §112.041, §112.042, §112.043 and the Miami-Dade County Code Section 11A1 through 13A1, Articles 3 and 4.

The Lessee shall assign responsibility to one of its officials to develop procedures that will assure that the policies of Equal Employment Opportunity and Affirmative Action are understood and implemented.

15.02 Nondiscriminatory Access to Premises:

The Lessee, for itself, its sublessees, successors, in interest, assigns, and contractors (herein collectively for this Section 15.02 the "Lessee"), as part of the consideration hereof, does hereby covenant and agree that (1) no person on the grounds of race, color, or national origin, will be excluded from participation in, denied the benefits of, or be otherwise subjected to discrimination in the use of the Premises and improvements hereunder, (2) that in the construction of any improvements on, over, or under the Premises hereunder, and the furnishing of services therein or thereon, no person the ground of race, color, or national origin, will be excluded from participation in, denied the benefits of, or otherwise be subjected to discrimination, (3) that the Lessee will use the Premises and improvements in compliance with all other requirements imposed by or pursuant to the List of Nondiscrimination Acts and Authorities attached hereto as Exhibit R-02.

15.03 Breach of Nondiscrimination Covenants:

In the event it has been determined that the Lessee has breached any enforceable nondiscrimination covenants contained in Section 15.01 Equal Employment Opportunity and Section 15.02 Nondiscriminatory Access to Premises above, pursuant to the complaint procedures contained in the applicable Federal Regulations, and the Lessee fails to comply with the sanctions and/or remedies which have been prescribed, the County shall have the right to terminate this Agreement pursuant to the Termination of the Agreement section hereof.

15.04 Nondiscrimination:

During the performance of this Agreement, the Lessee agrees as follows: The Lessee shall, in all solicitations or advertisements for employees placed by or on behalf of the Lessee, state that all qualified applicants will receive consideration for

employment without regard to age, sex, race, color, religion, marital status, place of birth or national origin, ancestry, physical handicap or disability. The Lessee shall furnish all information and reports required by Executive Order 11246 issued September 24, 1965, as amended by Executive Order 113155, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to the Lessee books, records, accounts by the County and Compliance Review Agencies for purposes of investigation to ascertain the compliance with such rules, regulations, and orders. In the event of the Lessee's noncompliance with the nondiscrimination clauses of this Agreement or with any of the said rules, regulations, and orders, this Agreement may be canceled, terminated, or suspended in whole or in part in accordance with the Termination of Agreement section hereof and the Lessee may be declared ineligible for further contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, as amended by Executive Order 113155 and such sanctions as may be imposed and remedies invoked as provided in Executive Order 113155 and such sanctions as may be imposed and remedies invoked as provided in Executive Order 11246 as amended or by rules, regulations, and orders of the Secretary of Labor, or as otherwise provided by law.

The Lessee will include Section 15.01 Equal Employment Opportunity and Section 15.02 Nondiscriminatory Access to Premises of this Article in the Lessee subcontracts in excess of \$10,000.00, unless exempted by rules, regulations or orders of the Secretary of Labor issued pursuant to Section 204 of Executive Order 11246 issued September 24, 1965, as amended by Executive Order 113155, so that such provisions will be binding upon each sub-consultant. The Lessee shall take such action with respect to any sub-contract as the County may direct as a means of enforcing such provisions, including sanctions for noncompliance; provided, however, that in the event the Lessee becomes involved in, or is threatened with, litigation with a sub-consultant as the result of such direction by the County or by the United States, the Lessee may request the United States to enter into such litigation to protect the interests of the United States.

15.05 Disability Nondiscrimination Affidavit:

By entering into this Agreement with the County and signing the Disability Nondiscrimination Affidavit, the Lessee attests that this is not in violation of the Americans with Disabilities Act of 1990 (and related Acts) or Miami-Dade County Resolution No. R-385-95. If the Lessee or any owner, subsidiary or other firm affiliated with or related to the Lessee is found by the responsible enforcement officer of the Courts or the County to be in violation of the Act or the Resolution, such violation shall render this Contract terminable in accordance with the Termination of Agreement section hereof. This Contract shall be void if the Lessee submits a false affidavit pursuant to this Resolution or the Lessee was not in violation at the time it submitted its affidavit.

15.06 <u>Affirmative Action/Nondiscrimination of Employment Promotion and</u> <u>Procurement Practices:</u>

(County Code Section 2-8.1.5): In accordance with the requirements of County Code Section 2-8.1.5, all firms with annual gross revenues in excess of \$5 million seeking to contract with Miami-Dade County shall, as a condition of award, have a written Affirmative Action Plan and Procurement Policy on file with the County's Department of Procurement Management. Said firms must also submit, as a part of their Lease to be filed with the Clerk of the Board, an appropriately completed and signed Affirmative Action Plan/Procurement Policy Affidavit.

Firms whose Boards of Directors are representative of the population make-up of the nation are exempt from this requirement and must submit, in writing, a detailed listing of their Boards of Directors, showing the race or ethnicity of each board member, to the County's Department of Procurement Management. Firms claiming exemption must submit, as part of their Lease to be filed with the Clerk of the Board, an appropriately completed and signed Exemption Affidavit in accordance with County Code Section 2-8.1.5. These submittals shall be subject to periodic reviews to assure that the entities do not discriminate in their employment and procurement practices against minorities and women/owned businesses.

It will be the responsibility of each firm to provide verification of their gross annual revenues to determine the requirement for compliance with the County Code Section. Those firms that do not exceed \$5 million annual gross revenues must clearly state so in their Lease.

15.07 <u>Title VI Clauses Applicable to this Agreement</u>:

(A) The Lessee, for itself, its successors in interest, and assigns, as a part of the consideration hereof, does hereby covenant and agree, as a covenant running with the land, that (1) no person on the ground of race, color, or national origin, will be excluded from participation in, denied the benefits of, or be otherwise subjected to discrimination in the use of said facilities, (2) that in the construction of any improvements on, over, or under such land, and the furnishing of services thereon, no person on the ground of race, color, or national origin, will be excluded from participation in, denied the benefits of, or otherwise be subjected to discrimination, (3) that the Lessee will use the premises in compliance with all other requirements imposed by or pursuant to the Title VI List of Pertinent Nondiscrimination Acts and Authorities in Article 15.08.

(B) With respect to licenses, leases, permits, etc., in the event of breach of any of the above Nondiscrimination covenants, the County will have the right to terminate the Lease and to enter, re-enter, and repossess said lands and facilities thereon, and hold the same as if the Lease had never been made or issued.

15.08 Title VI List of Pertinent Nondiscrimination Acts and Authorities:

During the performance of this Lease, the Lessee agrees to comply with the following nondiscrimination statutes and authorities; including but not limited to:

- Title VI of the Civil Rights Act of 1964 (42 U.S.C. § 2000d *et seq.*, 78 stat. 252), (prohibits discrimination on the basis of race, color, national origin);
- 49 CFR part 21 (Non-discrimination In Federally-Assisted Programs of The Department of Transportation—Effectuation of Title VI of The Civil Rights Act of 1964);
- The Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970, (42 U.S.C. § 4601), (prohibits unfair treatment of persons displaced or whose property has been acquired because of Federal or Federal-aid programs and projects);
- Section 504 of the Rehabilitation Act of 1973, (29 U.S.C. § 794 *et seq.*), as amended, (prohibits discrimination on the basis of disability); and 49 CFR part 27;
- The Age Discrimination Act of 1975, as amended, (42 U.S.C. § 6101 *et seq.*), (prohibits discrimination on the basis of age);

- Airport and Airway Improvement Act of 1982, (49 USC § 471, Section 47123), as amended, (prohibits discrimination based on race, creed, color, national origin, or sex);
- The Civil Rights Restoration Act of 1987, (PL 100-209), (Broadened the scope, coverage and applicability of Title VI of the Civil Rights Act of 1964, The Age Discrimination Act of 1975 and Section 504 of the Rehabilitation Act of 1973, by expanding the definition of the terms "programs or activities" to include all of the programs or activities of the Federal-aid recipients, sub-recipients and Lessees, whether such programs or activities are Federally funded or not);
- Titles II and III of the Americans with Disabilities Act of 1990, which prohibit discrimination on the basis of disability in the operation of public entities, public and private transportation systems, places of public accommodation, and certain testing entities (42 U.S.C. §§ 12131 – 12189) as implemented by Department of Transportation regulations at 49 CFR parts 37 and 38;
- The Federal Aviation Administration's Non-discrimination statute (49 U.S.C. § 47123 (prohibits discrimination on the basis of race, color, national origin, and sex);
- Executive Order 12898, Federal Actions to Address Environmental Justice in Minority Populations and Low-Income Populations, which ensures nondiscrimination against minority populations by discouraging programs, policies, and activities with disproportionately high and adverse human health or environmental effects on minority and low-income populations;
- Executive Order 13166, Improving Access to Services for Persons with Limited English Proficiency, and resulting agency guidance, national origin discrimination includes discrimination because of limited English proficiency (LEP). To ensure compliance with Title VI, Lessee must take reasonable steps to ensure that LEP persons have meaningful access to its programs (70 Fed. Reg. at 74087 to 74100);
- Title IX of the Education Amendments of 1972, as amended, which prohibits Lessee from discriminating because of sex in education programs or activities (20 U.S.C. 1681 et seq).

ARTICLE 16 Security and Special Provisions

16.01 Security:

The Lessee acknowledges and accepts full responsibility for (i) the security and protection of the Premises, any improvements thereon, its equipment and property on the Airport and (ii) all breaches of federal and Lessor security requirements by Lessee's employees or those persons for whom Lessee has responsibility under Article 16.02, and (iii) control of access to the AOA or any Security Identification Display Area ("SIDA") through the Premises by persons and vehicles. The Lessee fully understands and acknowledges that any security measures deemed necessary by the Lessee for the protection of said Premises, equipment and property shall be the sole responsibility of the Lessee and shall involve no cost to the County. Lessee is responsible for compliance by its employees and all others for whom it is responsible with applicable security requirements relating to access, through Lessee's Premises, to the AOA/SIDA. All such security measures by the Lessee shall be in accordance with 49 CFR Part 1542 and the Airport Security Plan.

16.02 Security Identification Display Areas Access - Identification Badges:

Security Identification Display Areas Access - Identification Badges: The Lessee shall be responsible for (i) assuring that all of Lessee's employees, and all employees

and persons of entities accessing the AOA/SIDA through the Premises on behalf of Lessee (collectively herein, the "SIDA Users"), have appropriate SIDA Identification Badges and comply with all federal and Lessor security requirements applicable to the Premises and SIDAs, (ii) immediately reporting to MDAD all lost or stolen ID badges of Lessee's employees and/or visitors' badges provided by Lessee, and (iii) immediately returning the ID badges of any of Lessee's employees that are transferred from the Airport or terminated from the employ of the Lessee or upon termination of this Agreement. Each employee must complete the SIDA training program conducted by the Department, before an ID badge is issued. The Lessee shall pay, or cause to be paid, to the Department any fines or penalties imposed on Lessor for a violation of the security requirements set forth herein, as well as such nondiscriminatory charges, as may be established from time to time, for lost or stolen ID badges and those not returned to the Department in accordance with this Article. The Department shall have the right to require the Lessee to conduct background investigations and to furnish certain data on such employees before the issuance of ID badges, which data may include the fingerprinting of employee applicants for the badges.

16.03 AOA - Driver Training:

Before the Lessee shall permit any employee to operate a motor vehicle of any kind or type on the AOA, the Lessee shall require such employee to attend and successfully complete the AOA Driver Training Course conducted from time to time by the Department. The privilege of a person to operate a motor vehicle on the AOA may be withdrawn by the Department for any violation of AOA driving rules. Notwithstanding the above, the Lessee shall be responsible for ensuring that all such vehicle operators possess current, valid, appropriate Florida driver's licenses.

16.04 Alcohol and Drug Testing:

The Lessee acknowledges that the County, as a public agency sponsor under the provisions of the Airport and Airway Improvement Act of 1982, as amended (the "Act"), has the obligation to establish a drug free workplace and to establish policies and programs to ensure airport safety and security. The Lessee acknowledges that the Department, on behalf of the County, has the right to require users of the Airport (Lessees, Permittees, Licensees, etc.) to establish reasonable programs to further the achievement of the objectives described herein. Accordingly, the Lessee shall establish programs for pre-employment alcohol and drug screening for all candidates for employment at the Airport who will as a part of their duties (a) be present on the AOA; (b) operate a motor vehicle of any type on the AOA; or (c) operate any equipment, motorized or not, on the AOA and for the same or similar screening based upon a reasonable suspicion that an employee, while on duty on the AOA, may be under the Notwithstanding the above, the Lessee specifically influence of alcohol or drugs. acknowledges that the County, acting through the Department, has the right and obligation to deny access to the AOA and to withdraw AOA driving privileges from any person who it has a reasonable suspicion to believe is under the influence of alcohol or drugs.

16.05 Drug-Free Workplace Default:

The Lessee acknowledges it has provided to the County a Drug-Free Workplace Affidavit certifying that it is providing a drug-free workplace for its employees, as required by County Ordinance No. 92-15, adopted on March 17, 1992 as amended from time to time ("Ordinance"). Based on the provisions of said Ordinance, the County shall have the right, upon 30 days written notice to the Lessee, to terminate this Agreement in the event the Lessee fails to provide, as of each anniversary of the effective date of this Agreement, the annual re-certification affidavit as required by the Ordinance; provided, however, that such termination shall not be effective if the Lessee submits the required Affidavit within the notice period.

Further, this Agreement shall be terminated upon not less than fifteen (15) calendar days written notice to the Lessee and without liability to the County, if the Department or the County Mayor determines any of the following:

- A) That the Lessee has made a false certification in its execution of the Affidavit submitted or in its annual re-certification as required by the Ordinance;
- B) That the Lessee has violated its original or renewal certification by failing to carry out any of the specific requirements of the Ordinance, other than the annual re-certification; or
- C) That such a number of employees of the Lessee have been convicted of violations occurring in its workplace(s) as to indicate that the Lessee has failed to make a good faith effort to provide a drug-free workplace as required by the Ordinance.

16.06 Special Programs:

The Lessee shall ensure that all employees participate in such safety, security and other training and instructional programs, as the Department or appropriate Federal agencies may from time to time require.

16.07 Vehicle Permit and Company Identification:

Motor vehicles and equipment of the Lessee operating on the AOA must have an official motor vehicle identification permit issued pursuant to Operational Directives of the Department. In addition, company identification must be conspicuously displayed thereon.

16.08 Federal Agencies Right to Consent:

The Lessee understands and agrees that all persons entering and working in or around arriving international aircraft and facilities used by the various Federal Inspection Services agencies may be subject to the consent and approval of such agencies. Persons not approved or consented to by the Federal Inspection Services agencies shall not be employed by the Lessee in areas under the jurisdiction or control of such federal inspection agencies.

16.09 AOA - Right to Search:

The Lessee agrees that its vehicles, cargo, goods and other personal property are subject to being searched when attempting to enter or leave and while on the AOA. The Lessee further agrees that it shall not authorize any employee or agent to enter the AOA unless and until such employee or agent has executed a written consent-to-search form acceptable to the Department. Persons not executing such consent-to-search form shall not be employed by the Lessee at the Airport, in any job requiring access to the AOA.

It is further agreed that the Department has the right to prohibit an individual, agent or employee of the Lessee from entering the AOA based upon facts which would lead a person of reasonable prudence to believe that such individual might be inclined to engage in theft, cargo tampering, aircraft sabotage or other unlawful activities. Any person denied access to the AOA or whose prior authorization has been revoked or suspended on such grounds shall be entitled to a hearing before the Director of the Department or his authorized designee within a reasonable time. Prior to such hearing, the person denied access to the AOA shall be advised, in writing, of the reasons for such denial.

The Lessee acknowledges and understands that these provisions are for the protection of all users of the AOA and are intended to reduce the incidence of thefts, cargo tampering, aircraft sabotage and other unlawful activities at the Airport.

16.10 Right of Flight:

There is hereby reserved to the County, its successors and assigns, for the use and benefit of the County and the public, a right of flight for the passage of aircraft in the air space above the surface of the premises herein leased, together with the right to cause in said air space such noise as may be inherent in the operation of aircraft, now known or hereafter used for navigation of or flight in the air, using said air space for landing at, taking off from or operating on Miami International Airport.

16.11 Height Restrictions:

The Lessee expressly agrees for itself, its successors and assigns, to restrict the height of structures, objects of natural growth and other obstructions on the leased premises to such a height so as to comply with Federal Aviation Regulations, Part 77 and with the Code of Miami-Dade County, whichever is more restrictive.

ARTICLE 17 Control of Employees

17.01 Control of Employees:

The Lessee shall properly control the actions of its employees at all times that said employees are working on the Airport, ensuring that they present a neat appearance and discharge their duties in a courteous and efficient manner and that they maintain a high standard of service to the public.

17.02 Lessee's Responsibility for Employee's Violations:

In the event the Lessee is in default of the covenants of Article 17.01 (Control of Employees) for failure to properly control its employees or by permitting its employees to improperly use the facilities provided by the County, the Department shall have the right to require the Lessee to conduct an investigation into any claimed violation of the covenants; if such investigation substantiates a violation, Lessee agrees to administer the appropriate discipline up to and including discharge of the offending employee.

ARTICLE 18 Civil Actions

18.01 Governing Law; Venue:

This Agreement shall be governed and construed in accordance with the laws of the State of Florida. The venue of any action on this Agreement shall be laid in Miami-Dade County, Florida, and any action to determine the rights or obligations of the parties hereto shall be brought in the courts of the State of Florida.

18.02 Notice of Commencement of Civil Action:

In the event that the County or the Lessee commence a civil action where such action is based in whole or in part on an alleged breach of this Agreement, the County and the Lessee agree that service of process shall be made pursuant to the rules of Civil Procedure in the court in which the action has been filed.

18.03 Registered Office/Agent; Jurisdiction:

Notwithstanding the provisions of Article 18.02 (Notice of Commencement of Civil Action), and in addition thereto, the Lessee, if a corporation, shall designate a registered office and a registered agent, as required by Section 48.091, Florida Statutes, such designations to be filed with the Florida Department of State in accordance with Section 607.0501, Florida Statutes. If the Lessee is a natural person, he and his personal representative hereby submit themselves to the jurisdiction of the Courts of this State for any cause of action based in whole or in part on an alleged breach of this Agreement.

ARTICLE 19

Trust Agreement

19.01 Incorporation of Trust Agreement by Reference:

Notwithstanding any of the terms, provisions and conditions of this Agreement, it is understood and agreed by the parties hereto that, to the extent of any inconsistency with or ambiguity relating to the terms and conditions of the Agreement, and the levels of rents, fees or charges required hereunder and their modification or adjustment as may be required by terms and conditions of this Agreement, and the level of rents, fees or charges required hereunder and their periodic modification or adjustment as may be required by the provisions of the Amended and Restated Trust Agreement dated as of the 15th day of December, 2002 as amended by and between the County and the JP Morgan Chase Bank as Trustee and Wachovia Bank, National Association as Co-trustee, and specifically the terms of Section 501 thereof, shall prevail and govern at all times during the term of this Agreement. Copies of the Trust Agreement are available for inspection in the offices of the Department during normal working hours.

19.02 Adjustment of Terms and Conditions:

If, at any time during the term of this Agreement, a Federal agency or court of competent jurisdiction shall determine that any of the terms and conditions of this Agreement, including the rentals, fees and charges required to be paid hereunder to the County by the Lessee or by other Lessees under other Agreements of the County for the lease or use of facilities used for similar purposes, are unjustly discriminatory, the County shall have the right to modify such terms and conditions and to increase or otherwise adjust the rentals, fees and charges required to be paid under this Agreement in such a manner as the County shall determine is necessary and reasonable so that the rentals, fees and charges payable by the Lessee and others shall not thereafter be unjustly discriminatory to any user of like facilities and shall not result in any violation of the Trust Agreement and/or bond resolution or in any deficiency in revenues necessary to comply with the covenants of the Trust Agreement and/or bond resolution. In the event the County has modified the terms and conditions of this Agreement. including any adjustment of the rentals, fees and charges required to be paid to the County pursuant to this provision, this Agreement shall be amended to incorporate such modification of the terms and conditions including the adjustment of rentals, fees and charges upon the issuance of written notice from the Department to the Lessee.

ARTICLE 20 Other Provisions

20.01 No Representation:

The County makes no representation, warranty, guarantee, or averment of any nature whatsoever concerning the physical condition of the Premises, and it is agreed that County will not be responsible for any loss, damage or costs which may be incurred by the Lessee by reason of any such physical condition.

20.02 <u>Headings</u>:

Any headings preceding the text of any articles, paragraphs or sections of this Agreement shall be solely for convenience of reference and shall not constitute a part of this Agreement, nor shall they affect its meaning, construction or effect.

20.03 Interference:

The Lessee further expressly agrees to prevent any use of the Premises, which would interfere with or adversely affect the operation or maintenance of the Airport or otherwise constitute an airport hazard.

20.04 Authorized Uses Only:

The Lessee shall not use or permit the use of the Premises for any illegal or unauthorized purpose or for any purpose which would increase the premium rates paid by the County on, or invalidate, any insurance policies of the County or any policies of insurance written on behalf of the Lessee under this Agreement.

20.05 Binding Effect:

The terms, conditions and covenants of this Agreement shall inure to the benefit of and be binding upon the parties hereto and their successors and assigns. This provision shall not constitute a waiver of any conditions prohibiting assignment or subletting.

20.06 Federal Subordination:

This Agreement shall be subordinate to the provisions of any existing or future agreements between the County and the United States of America relative to the operation and maintenance of the Airport, the execution of which has been or may be required as a condition precedent to the expenditure of Federal funds for the development of the Airport. All provisions of this Agreement shall be subordinate to the right of the United States of America to lease or otherwise assume control over the Airport, or any part thereof, during time of war or national emergency for military or naval use and any provisions of this Agreement inconsistent with the provisions of such lease to, or assumption of control by, the United States of America shall be suspended.

20.07 <u>Notices</u>:

All notices required or permitted to be given under the terms and provisions of this Agreement by either party to the other shall be in writing and shall be hand delivered or sent by registered or certified mail, return receipt requested, to the parties as follows:

As to the County or Aviation Department:

Director Miami-Dade County Aviation Department Post Office Box 0255504 Miami, Florida 33102-5504

As to the Lessee:

President National Airport Services, Inc. 6025 NW 18th Street Bldg #716-E, Suite Miami, Florida 33126

or to such other address as may hereafter be provided by the parties in writing. Notices by registered or certified mail shall be deemed received on the delivery date indicated by the U.S. Postal Service on the return receipt. Hand delivered notices shall be deemed received by the Lessee when presented to the local management representative of the Lessee.

20.08 Rights Reserved:

Rights not specifically granted the Lessee by this Agreement are reserved to the County.

20.09 Rights of County at Airport:

The County shall have the absolute right, without limitation, to make any repairs, alterations and additions to any structures and facilities at the Airport. The County shall, in the exercise of such right, be free from any and all liability to the Lessee for business damages occasioned during the making of such repairs, alterations and additions, except those occasioned by the sole active negligence of the County, its employees, or agents.

20.10 Rights to be Exercised by Department:

Wherever in this Agreement rights are reserved to the County, such rights may be exercised by the Department.

20.11 No Waiver:

There shall be no waiver of the right of either party to demand strict performance of any of the provisions, terms and covenants of this Agreement nor shall there be any waiver of any breach, default or non-performance hereof by either party, unless such waiver is explicitly made in writing by the other party. Any previous waiver or course of dealing shall not affect the right of either party to demand strict performance of the provisions, terms and covenants of this Agreement with respect to any subsequent event or occurrence of any subsequent breach, default or non-performance hereof by the other party.

20.12 Right to Regulate:

Except to the extent prohibited by Federal law, nothing in this Agreement shall be construed to waive or limit the governmental authority of the County, as a political subdivision of the State of Florida, to regulate the Lessee or its operations. Notwithstanding any provision of this Agreement, nothing herein shall bind or obligate the County, the Zoning Appeals Board, the Building and Zoning Department (as it may be renamed from time to time), the Planning Department, or any department, board or agency of the County to agree to any specific request of the Lessee that is related in any way to the regulatory or quasi-judicial power of the County; and the County shall be released and held harmless by the Lessee from any liability, responsibility, claims, consequential damages or other damages, or losses resulting from the denial or withholding of such requests, provided, however, that this provision shall not preclude any appeal from County action wherein the sole remedy sought is reversible of the County's action or injunctive relief; nor shall it preclude any action based on the County's bad faith, capricious behavior or arbitrary action.

20.13 Severability:

If any provision of this Agreement or the application thereof to either party to this Agreement is held invalid by a court of competent jurisdiction, such invalidity shall not affect other provisions of this Agreement which can be given effect without the invalid provision, and to this end, the provisions of this Agreement are severable.

20.14 Inspections:

The authorized employees and representatives of the County and of any applicable Federal or State agency having jurisdiction hereof shall have the right of access to the Premises at all reasonable times for the purposes of inspection and testing to determine compliance with the provisions of this Agreement. This right of inspection and testing shall impose no duty on the County to inspect and shall impart no liability upon the County should it not make any such inspections.

20.15 Payment of Taxes:

The Lessee shall pay all taxes and other costs lawfully assessed against its leasehold interests in the Premises, its improvements and its operations under this Agreement; provided, however, the Lessee shall not be deemed to be in default of its obligations hereunder for failure to pay such taxes pending the outcome of any legal proceedings instituted to determine the validity of such taxes. Failure to pay the taxes upon the adverse ultimate conclusion of such legal proceedings against the Lessee shall constitute a default.

20.16 Quiet Enjoyment of Others:

The Lessee shall control the actions of its employees, agents, invitees and those doing business with it, so as to not annoy, disturb or be offensive to others and to provide the service hereunder so as to not unreasonably create a nuisance or thing which may disturb the quiet enjoyment of any other users of the Airport.

20.17 Radon Disclosure:

In accordance with Section 404.056, Florida Statutes, the following disclosure is hereby made:

<u>Radon Gas</u>: Radon is a naturally occurring radioactive gas that, when it has accumulated in a building in sufficient quantities, may present health risks to persons who are exposed to it over time. Levels of radon that exceed federal and state guidelines have been found in buildings in Florida. Additional information regarding radon and radon testing may be obtained from your county public health unit.

20.18 Force Majeure:

The terms and conditions of the Agreement (with the exception of the obligation of the Lessee to pay the amounts required by the terms of this Agreement) shall be subject to "Force Majeure". Neither the County nor the Lessee shall be considered in default in the performance of its obligations hereunder, if such performance is prevented or delayed because of war, hostilities, revolution, civil commotion, strike, lock-out, epidemic, fire, wind, flood, hurricane, tornado, lightning, wind damage, or because of any law, order, proclamation, regulation or ordinance of any government or of any subdivision thereof because of any act of god or any other cause whether of similar or dissimilar nature beyond the reasonable control of the party affected, provided that notice of such force majeure is given by the affected party to the other within ten (10) days of the beginning of said force majeure. Should one or both of the parties be prevented from fulfilling their contractual obligation by a state of force majeure lasting continuously for a period of six (6) months, the parties shall consult with each other regarding the implementation of the Agreement.

20.19 Destruction of Premises:

In the event the Premises shall be destroyed or so damaged or injured by fire, windstorm, flood or other casualty (and in each such event the Lessee was not at fault in whole or in part) during the life of this Agreement that the Premises or any portion thereof are rendered untenantable, the County shall have the right, but not the obligation, to render said Premises or damaged portion thereof tenantable by repairs completed within a reasonable period of time.

- (A) Total Destruction: In the event the County elects not to render the Premises tenantable, if destroyed or damaged in their entirety, the Lessee shall be so notified in writing by the Department, and this Agreement shall be deemed terminated as of the date of the casualty, with the Lessee being liable only for payment of rentals on a pro rata basis as to whatever portion(s) of the Premises which were tenantable and used by the Lessee following the casualty. In such event, the Department shall endeavor to find adequate replacement premises for the Lessee in existing facilities on the Airport.
- (B) If the damaged portion of the Premises is not rendered tenantable by the County within a reasonable period of time, and the Lessee shall determine that: 1) the loss of the damaged portion of the Premises shall have a materially adverse impact on the ability of the Lessee to utilize the Premises for the purposes described in Article 2; or 2) would require the Lessee to obtain other space off the Premises in order to substantially conduct the operations of the Lessee originally conducted within the Premises, then, in either such event, upon written notice to the County, the Lessee may cancel this Agreement as of a date which shall be not later than three months from the giving of such notice, if the repairs are not completed within 90 days following such written notice of the intent to cancel, or if the County has not commenced repairs within such notice period for repairs which cannot be reasonably completed within such 90-day period. In the event of cancellation, the rent for the untenantable portion of the Premises shall be paid only to the date of such fire, windstorm, flood, or other casualty. If the Agreement is not canceled following any such casualty, the rent shall be abated as to the portion of the Premises rendered untenantable.

If the casualty was caused in whole or in part by the Lessee, its officers, employees, agents, contractors or trespassers, then the Lessee shall not have the right to terminate this Agreement and shall be responsible under other provisions of this Agreement for

payment to the County of all damage to the Premises, plus the loss of rentals attributable to the damaged or destroyed premises.

20.20 Quiet Enjoyment:

Subject to the terms of this Agreement, specifically including, but not limited to, environmental remediation steps to be taken under Article 8, the County's right and obligation to make certain repairs, alterations, and additions under Articles 5 (Maintenance by County) and 20.09 (Rights of County at Airport), which, for purposes of this clause, includes any and all demolition, in whole or in part, of buildings and runways, and roadway systems on or off the Airport, and the reservation of easement rights to the airspace under Article 16.10 (Right of Flight), all of which provisions and others in this Agreement, the Lessee acknowledges may cause disruption and disturbance to the Lessee, and upon the observance by the Lessee of all the terms, provisions, covenants, and conditions imposed upon the Lessee hereunder, the Lessee shall peaceably and quietly hold and enjoy the Premises for the term of this Agreement; provided, however, that the County shall not be liable for any violation of this clause or for any disturbance or disruption in or to the Lessee's business, for acts or omissions of tenants, users of the Airport, third parties or when any department or agency of the County is acting in its governmental capacity or by Acts of God.

20.21 Interpretation of Agreement:

This Agreement is the result of negotiation between the parties hereto and has been typed/printed by one party for the convenience of both parties, and the parties covenant that this Agreement shall not be construed in favor of or against any of the parties hereto.

20.22 Entirety of Agreement:

The parties hereto agree that this Agreement sets forth the entire agreement between the parties, and there are no promises or understandings other than those stated herein. None of the provisions, terms and conditions contained in this Agreement may be added to, modified, superseded or otherwise altered, except as may be specifically authorized herein or by written instrument executed by the parties hereto.

(This space is intentionally blank)

EXHIBIT K-08 Listing of Hazardous Materials Usage (ARTICLE 8.08)

Pursuant to Section 8.08, the tenant intends to use the following hazardous materials in the operations within their leasehold, which are subject to approval from the Miami-Dade Aviation Department (MDAD). Note this exhibit, including MDAD's approval, shall be part of the final Lease Agreement documentation.

PRODUCT NAME

- NAME OF MANUFACTURER ______
- CHEMICAL ABSTRACTS SERVICE (CAS) REGISTRY NUMBER (IF AVAILABLE) FOR EACH HAZARDOUS INGREDIENT IN THE PRODUCT

MDAD APPROVAL (CIVIL ENVIRONMENTAL ENGINEERING DIVISION)

EXHIBIT – K-13

(ARTICLE 8.13: Periodic Environmental Audits)

Miami-Dade Aviation Department Civil Environmental Engineering Division

TENANT AUDIT SUMMARY MINIMUM REQUIREMENTS

Environmental Compliance Audit:

The audits shall conform with the most current ASTM standard E-2107. At a minimum the audit scope shall include the areas listed below, detailed visual inspection of the tenant leasehold and operations; a review of documents made available by the facility; interviews with knowledgeable site representatives; the completion of a detailed compliance audit questionnaire; photographic documentation of selected site conditions; and any follow-up visits.

Areas of Concern:

- Environmental Management Systems
- Air Emissions
- Asbestos and Polychlorinated Biphenyls (PCBs)
- Hazardous Materials and Waste
- Oil Pollution Management
- Pesticides
- Solid Waste
- Storage Tanks
- Water Supply and Wastewater

Documentation:

- Permits, Licenses, Certifications, etc.
- All regulatory activities by any environmental agency.
- Standard Operating Procedures (SOPs)
- Best Management Practices (BMPs)
- Emergency Response (spills, etc.)
- SPCC Plan, if required
- Dangerous Goods
- Employees Training Programs and Records.

EXHIBIT R-02 FAA List of non-discrimination federal statutes (ARTICLE 15.02)

A6.3.6 Title VI List of Pertinent Nondiscrimination Acts and Authorities

Title VI List of Pertinent Nondiscrimination Acts and Authorities

During the performance of this contract, the contractor, for itself, its assignees, and successors in interest (hereinafter referred to as the "contractor") agrees to comply with the following non-discrimination statutes and authorities; including but not limited to:

- Title VI of the Civil Rights Act of 1964 (42 U.S.C. § 2000d et seq., 78 stat. 252), (prohibits discrimination on the basis of race, color, national origin);
- 49 CFR part 21 (Non-discrimination In Federally-Assisted Programs of The Department of Transportation—Effectuation of Title VI of The Civil Rights Act of 1964);
- The Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970, (42 U.S.C. § 4601), (prohibits unfair treatment of persons displaced or whose property has been acquired because of Federal or Federal-aid programs and projects);
- Section 504 of the Rehabilitation Act of 1973, (29 U.S.C. § 794 *et seq.*), as amended, (prohibits discrimination on the basis of disability); and 49 CFR part 27;
- The Age Discrimination Act of 1975, as amended, (42 U.S.C. § 6101 *et seq.*), (prohibits discrimination on the basis of age);
- Airport and Airway Improvement Act of 1982, (49 USC § 471, Section 47123), as amended, (prohibits discrimination based on race, creed, color, national origin, or sex);
- The Civil Rights Restoration Act of 1987, (PL 100-209), (Broadened the scope, coverage and applicability of Title VI of the Civil Rights Act of 1964, The Age Discrimination Act of 1975 and Section 504 of the Rehabilitation Act of 1973, by expanding the definition of the terms "programs or activities" to include all of the programs or activities of the Federal-aid recipients, sub-recipients and contractors, whether such programs or activities are Federally funded or not);
- Titles II and III of the Americans with Disabilities Act of 1990, which prohibit discrimination on the basis of disability in the operation of public entities, public and private transportation systems, places of public accommodation, and certain testing entities (42 U.S.C. §§ 12131 – 12189) as implemented by Department of Transportation regulations at 49 CFR parts 37 and 38;
- The Federal Aviation Administration's Non-discrimination statute (49 U.S.C. § 47123) (prohibits discrimination on the basis of race, color, national origin, and sex);
- Executive Order 12898, Federal Actions to Address Environmental Justice in Minority
 Populations and Low-Income Populations, which ensures non-discrimination against minority
 populations by discouraging programs, policies, and activities with disproportionately high and
 adverse human health or environmental effects on minority and low-income populations;
- Executive Order 13166, Improving Access to Services for Persons with Limited English Proficiency, and resulting agency guidance, national origin discrimination includes discrimination because of limited English proficiency (LEP). To ensure compliance with Title VI, you must take reasonable steps to ensure that LEP persons have meaningful access to your programs (70 Fed. Reg. at 74087 to 74100);
- Title IX of the Education Amendments of 1972, as amended, which prohibits you from discriminating because of sex in education programs or activities (20 U.S.C. 1681 et seq).

Issued on January 29, 2016 Airports (ARP) Page 23

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their appropriate officials as of the date first above written.

	BOARD OF COUNTY COMMISSIONERS OF MIAMI-DADE COUNTY, FLORIDA
	By: <u>K. A. Ayatt</u> Deputy Aviation Director
	Date://///9
	ATTEST: Harvey Ruvin, Clerk By:
	(CLERK SEAL)
Approved as to form and laged sufficiency	LESSEE: NATIONAL AIRPORT SERVICES, INC. By: Title:
ereality Pase.ney	Jorge J. Alvarez Print Name
	Date:
	ATTEST:
	Corporate Secretary
	Print Name

(CORP. SEAL)





	N C	TAUNA S
ST.		TAXIWAY "Q" RUNWAY 12:30
NIW BZINA ALE		12-30 (J) (J) (J) (J) (J) (J) (J) (J) (J) (J)
	WI58 WI BLDG 716	
	Airfield A	
CODE: SPACE CLASS	<u>SQ. FT.</u>	MIAMI DADE AVIATION DEPARTMENT MIAMI INTERNATIONAL AIRPORT
(As assigned by Airside Operations)		EXHIBIT P
SCALE: N.T.S. EFS #: 6230	DATE: 12/04/2018	NATIONAL AIRPORT SERVICES

MIAMI-DADE COUNTY, FLORIDA FACILITIES DIVISION AVIATION DEPARTMENT -



MDAD PROCEDURES FOR TENANT AIRPORT CONSTRUCTION REIMBURSABLE PROJECTS

PURPOSE

To provide details for the initiation and management of a Tenant Airport Construction Program reimbursable project.

DEFINITIONS

FAA	Federal Aviation Administration A/E Tenant's State Registered Architect or Engineer
	responsible for the design of the project.
BCC	Board of County Commissioners
GSA	General Services Administration
MARC	Miscellaneous Asbestos Recovery Contract
MCC/TAC	Miscellaneous Construction Contract/Tenant Airport construction
MDAD	Miami Dade Airport Aviation
NTP	Notice to Proceed
PM	Project Manager
TAC-N	Tenant Airport Construction Non- reimbursable projects
Tenant	Business Partner, Lessee

INSTRUCTION

GENERAL INFORMATION

Summary of Department Process for Design and Construction of TAC-R Projects

When an airport tenant wishes to improve or expand a leasehold area, the tenant must contact the MDAD Manager, Properties and Commercial Operations to discuss the proposed improvement or expansion. The Manager, Properties and Commercial Operations or designee determines whether the proposed design and construction will be a reimbursable or non-reimbursable project.

If the determination is that the proposed design and construction are reimbursable, the tenant must complete and submit MDAD Form FD1-001 requesting approval to design and construct the project, detailing the proposed construction and providing a proposed schedule and cost estimate.

The Manager, Properties and Commercial Operations or designee prepares an instruction and forwards it to the MDAD Manager, Planning, the MDAD Design Chief, the MDAD Manager, Maintenance Engineering, the MDAD Manager, Terminal Facilities, and others as appropriate, for review and tentative approval. The following documents, at a minimum, are attached to the form: MDAD Form FD1-001 "TAC-R Projects Request for Approval" shall include the following

attachments:

- Conceptual drawings/sketches
- > Completed Tenant Project Information Sheet providing the following information:
 - ✓ Project Name
 - ✓ Scope of Work (project description)
 - Project Cost (tenant's estimate) broken down between design and construction
 - ✓ Tenant's Name and Contact Person's name and telephone numbers
 - ✓ A/E of Record Name and Contact Person's name and telephone numbers
 - ✓ Contractor Name and Contact Person's name and telephone numbers
 - Verification that the Tenant has been instructed on insurance responsibilities and MDAD TAC-N procedures and requirements

FP0-110-P REV 9/2016

Page 1 of 2

MIAMI-DADE COUNTY, FLORIDA FACILITIES DIVISION

AVIATION DEPARTMENT -



If tentative approval in denied, the tenant is notified through MDAD Manager, Properties and Commercial Operations.

If tentative approval is given, the tenant is so advised through MDAD Manager, Properties and Commercial Operations and is invited to a meeting with a project conceptual review team which includes the MDAD Manager, Planning the MDAD Design Chief, the TAC-R chief to discuss the design process, bid and award process, construction and close out process. These discussions will include cost and schedule, and the identification of a TAC-R Project Manager.

There are standard MDAD procedures for the above named processes. However, dependent on the complexity, size, location and cost of the project, the conceptual review team may, through the TAC-R chief, waive certain aspects of these standard procedural requirements.

Procedures for Design and Construction

1. The design of the project will entail the Selection of an Architect/Engineer. The procedures for this activity are outlined in MDAD Procedure FD1-050-P and FD2-010

2. Upon selection, the A/E will be required to produce the following:

- (a) A Planning Book in accordance with MDAD Procedure FD1-040-P
- (b) A Project Book in accordance to MDAD Procedure FD1-050-P
- (c) Design Documents in accordance with MDAD Procedure FD3-007-P
- (d) Design includes Contract Formation in accordance with MDAD Procedure FD4-009-P

3. Upon satisfactory completion of the design process, the project shall be bid and awarded for construction in accordance with MDAD Procedure FD4-011-P

4. Satisfactory completion of the project is dependent on inspections consistent with MDAD Procedures for Substantial Completion and Beneficial Occupancy as outlined in MDAD Procedures FD5-195-P

5. Upon satisfactory completion of item (4) above the project will be closed out accordance with MDAD Procedures FD5-245-P and tenant will get reimbursed after form FD5-265 "TAC-R Project Reimbursement Approval Check List" is processed and reimbursement requirements are met.

6. The Project Manager shall be mindful of the fact that the above core MDAD Procedures with references to other Procedures together form the broad spectrum of management structures for the project.

FP0-110-P REV 9/2016

Page 2 of 2

Lse. No. <u>C009584</u> Cust. No. <u>NATP39</u> Doc. Name <u>NATP9584SE</u>

LEASE AGREEMENT BETWEEN **MIAMI-DADE COUNTY**, FLORIDA, AS LESSOR, AND **NATIONAL AIRPORT SERVICES,** INC., AS LESSEE, MIAMI INTERNATIONAL AIRPORT

26th THIS LEASE AGREEMENT ("Agreement"), is made and entered into as of the day of <u>April</u>, 2016, by and between MIAMI-DADE COUNTY, a political subdivision of the State of Florida ("County") and NATIONAL AIRPORT SERVICES, INC. a Florida corporation ("Lessee").

WITNESSETH:

For and in consideration of the premises and of the mutual covenants and agreements herein contained, the parties hereto agree as follows:

ARTICLE 1 Term and Premises

1.01 <u>Term</u>: The County hereby leases to the Lessee, and the Lessee hereby leases from the County, the premises described in Article 1.02 (Premises) hereof, for the purposes and uses set forth in Article 2 (Use of Premises) hereof, for a term of month to month, not to exceed five years, commencing on May 1, 2016, and terminating on April 30, 2021 unless sooner terminated as provided for herein.

1.02 <u>Premises</u>: The premises leased herein are located in Building 716 F in the Westside Cargo Area of Miami International Airport ("Airport"), and are more particularly described as follows and as shown on Exhibits A, and P, dated March 1, 2016, attached hereto and made a part hereof ("Premises"):

Building 716E & F:

Air-conditioned Office (716-F, 1st level)1,659 square feetNon-air-conditioned warehouse space7,100 square feetNon-air-conditioned warehouse space (GSE)1,282 square feetLand10,041 square feet

1.03 <u>Suitability of Premises</u>: The Lessee acknowledges that the Premises are suitable for the Lessee's proposed use and that the County has no obligation to perform or cause to be performed any maintenance, repairs, clean-ups, painting, or the like. The Lessee's obligation under this Agreement, such as in Article 6.01(B) (Permits and Licenses), to obtain all operating permits required of the Lessee, shall not require the County to take any action or perform any tasks within the Premises to enable the Lessee to obtain such permits, including, but not limited to, certificates of occupancy, which shall remain the Lessee's exclusive obligation to perform in order to obtain such permits.

1

1.04 Adjustment of Premises:

(A) The Premises leased hereunder are subject to relocation, or modification, by addition or deletion, with the mutual consent of the Lessee and the County Aviation Department (the "Department"), and Articles 1.02 (Premises) and 3.01 (Annual Rental) hereof and the exhibits to this Agreement shall then be administratively revised to reflect such relocation, modification, or deletion upon 30 days written notice to the Lessee by the Department. Relocated space may not be similar in size, configuration, or location to the Premises leased herein.

(B) In the event the Lessee requests approval to handle cargo for others within and from the Premises, pursuant to Article 2.04 (Cargo Handling of Others) hereof, if the Department, in its sole judgment, determines that the Lessee is leasing more space than it requires for its own operations, the Department may modify the Premises by deleting such excess space from the Premises and, in such event, Articles 1.02 (Premises) and 3.01 (Annual Rental) hereof and accompanying exhibits to this Agreement shall thereafter be administratively adjusted to reflect such deletion, upon not less than 30 days written notice to the Lessee.

1.05 <u>County Right to Terminate for Airport Development</u>: The County shall have the right, at any time during the term of this Agreement, to terminate the Agreement upon not less than six months advance written notice to the Lessee, if the Premises are needed by the County for Airport development projects as approved by the Board of County Commissioners. Such termination shall not entitle the Lessee to any relocation costs, and in no event shall the County be responsible to the Lessee for any costs, damages, expenses, moving costs, loss of revenues, business interruption damages, or the like, caused by or arising out of such termination, nor shall such termination be deemed an eminent domain taking for any purpose.

Aircraft Parking: The Lessee shall be entitled to use aircraft parking 1.06 position(s), as shown on Exhibit P, dated March 1, 2015, attached hereto and made a part hereof, as designated from time to time by the Department, for the parking and around handling of the Lessee's regularly scheduled all-cargo flights, or for all-cargo flights routed to the Airport due to weather or emergency conditions, for time periods and at use fees as may be established from time to time by the Department. It shall be the responsibility of the Lessee to request use and assignment of each parking position in advance to secure approval of such aircraft parking from the Department, in accordance with procedures established from time to time by the Department, and to pay the established rates and fees required for such use and assignment. The Department shall make reasonable efforts to assign the Lessee aircraft parking position(s) most convenient to the Premises leased by the Lessee hereunder, subject to availability and within the assignment procedures established by the Department; provided, however, the County shall have no liability to the Lessee if such parking positions are not available for assignment to the Lessee. Use of the aircraft parking position(s) by the Lessee shall be solely for aircraft operated by or on behalf of the Lessee and shall not be transferred or assigned by the Lessee to others.

ARTICLE 2 Use of Premises

2.01 <u>Authorized Use of Premises</u>: The Lessee shall use the Premises for the following purposes only:

(A) As to the Ground Floor operational and office space:
- (1) Receiving, storing, consolidating and packaging of air cargo for airlines. Handling and pre-cooling of flowers and other perishable fresh produce cargo for third parties (Lessee's customers) including pickup and delivery, loading and unloading of aircraft (prohibited in the Terminal Building area), refrigeration, storage, consolidation and dispatch of flowers and other perishable cargo transported or to be transported by air or land.
- (2) Transportation, from and to other designated premises, of air cargo of the Lessee's customers to and from the leased premises.
- (3) Transportation of air cargo to and from aircraft, including the loading and unloading of such aircraft, except that such loading and unloading of aircraft is not authorized in the Terminal Building Area of the Airport. The maintenance and storage of cargo handling vehicles and equipment is prohibited except in the areas specifically leased by the Lessee hereunder, unless otherwise approved in writing by the Department.
- (4) Preparation of manifests and other documentation and the performance of administrative functions directly related to the services provided pursuant Sections (1) through (3) above.
- (5) Operation of bonded and non-bonded warehouses.
- (6) For normal office operations associated with the Lessee's cargo handling business.
- (7) Such other activities as are directly related or incidental to the authorized cargo handling use of the Premises.
- (8) All activities listed above related to the Lessee's cargo handling business shall be performed by employees of the Lessee. There shall be no right of the Lessee to employ agents to perform activities hereunder unless such agents have authorization from the Department through permit or lease to operate on the Airport.
- (B) Paved ramp space, immediately adjacent to the leased Premises, may be used for the following purposes only:
 - (1) Staging, loading and unloading of air cargo shipped or to be shipped by the Lessee.
 - (2) Parking of operable aircraft service equipment of the Lessee related to its cargo activities.

The Lessee shall not permit these activities to interfere with designated service roads or lanes or so as to interfere with the activities of others.

2.02 <u>Common Use Aircraft Parking Positions</u>: Common use aircraft parking positions, adjacent or proximate to the leased Premises, may be used for the following purposes only:

- (A) For loading and unloading of all-cargo aircraft owned, operated, wetleased, dry-leased or chartered by the Lessee.
- (B) For the performance of transit or turnaround aircraft maintenance on allcargo aircraft owned, operated, wet-leased, dry-leased or chartered by the Lessee.
- (C) For the fueling and servicing of all-cargo aircraft owned, operated, wetleased, dry-leased, or chartered by the Lessee.

Common use aircraft parking positions shall not be used for the parking or storage of ground service equipment, containers or like equipment.

- 2.03 Vehicular Parking:
- (A) Vehicular parking on the landside, immediately adjacent to the Premises, may be used only as truck loading and unloading stalls and for business guest parking. The Lessee shall not park its own vehicles nor permit its officers or employees to park in such stalls.
- (B) Common use public parking areas adjacent to or in the vicinity of Building 716 may be used by officers, employees and business guests of the Lessee and of other tenants of the Building and nearby buildings. Vehicles of the Lessee which are not required for day-to-day operations of the Lessee or which are not operable shall not be parked at, in or near the Building in which the Premises are located nor elsewhere on the Airport.
- (C) Parking of any type (i.e. employees, public or business guest, trucks, etc.) on any landscaped or grassed area is prohibited.

2.04 <u>Cargo Handling of Others</u>: The handling of air cargo of other airlines and/or the cargo servicing of aircraft of others on or from the leased Premises is specifically prohibited, without the express, prior written approval of the Department, which approval, if granted, may be rescinded by the Department upon 30 days written notice. If such cargo handling or aircraft cargo servicing is so approved, the Lessee shall be required to pay the County a percentage (currently established at 7%) of the gross revenues received by the Lessee from such activities, in accordance with procedures (which procedures include a definition of gross revenues, payment and reporting requirements, audit requirements and the like) to be transmitted to the Lessee in connection with such approval. Failure of the Lessee to comply with such procedures shall constitute a default pursuant to Article 13.03 (Other Defaults) hereof.

2.05 <u>Non-Flyable Aircraft</u>: Aircraft in non-flyable condition shall not be parked or stored on the leased Premises for a period in excess of 60 days without the prior written approval of the Department, which shall not be unreasonable withheld or delayed.

ARTICLE 3 Rentals and Payments

3.01 <u>Annual Rental</u>: As annual rental for the lease of the Premises, the Lessee shall pay to the County, commencing on May 1, 2016, the sum of **\$143,040.75** payable in twelve equal monthly installments of **\$11,920.06**, in U.S. funds, on the first day of each and every month in advance and without billing, at the offices of the Department

as set forth in Article 3.06 (Address for Payments). Said rental is computed as follows:

TOTAL				\$143,040.75	\$11,920.06
Land		10,041	\$1.75	\$17,571.75	\$1,464.31
Non-A/C Warehouse GSE	107161695	1,282	\$12.00	\$15,384.00	\$1,282.00
Non-A/C Warehouse	117161840	7,100	\$12.00	\$85,200.00	\$7,100.00
A/C Office (1st level)	87161820	1,659	\$15.00	\$24,885.00	\$2,073.75
Building 716-F					
Description	ID#	SQ. FT.	RATE	ANNUAL	MONTHLY

TABLE 1

Plus applicable State sales taxes, as required by law

Rental rates exclude electricity, water and sewer and storm drainage utilities. Rental rate includes basic usage levels of electricity, water and sewer and janitorial services to common areas.

3.02 <u>Opportunity Fee</u>: As additional consideration for the rights and privileges granted herein, the Lessee shall pay to the County the amount by which 7% of the monthly Gross Revenues, as defined in Article 3.10 (Gross Revenues), generated from its activities under this Agreement, exceeds the monthly rentals, as adjusted from time to time, under Articles 3.01 (Annual Rental) and 3.05 (Rental Rate Review). The Lessee shall pay such amount to the County by the tenth day of the month following the month in which the Gross Revenues were received or accrued. The percentage fees payable on any unreported Gross Revenues, determined by the annual audit required pursuant to Article 3.13 (Annual Audit Required) are considered, for the purposes of Article 3.08 (Late Payment Charge), as having been due on the tenth day of the month following the month during which such unreported Gross Revenues were received or accrued. Such opportunity fee is not a payment for the lease or license to use the Premises, but rather payment for the Lessee's privilege of doing business on the Airport.

3.03 Security Deposit: Unless previously deposited with the County, within three days after the date this Agreement is fully executed, Lessee will deposit with the County an amount equal to two times the required total monthly rental of occupied space or all space to which Lessee has access to as determined pursuant to Article 3.01 above, plus applicable State sales tax thereon. The County will hold the Security Deposit as security for the payment of the Lessee's obligations hereunder. Said Security Deposit shall be in addition to any rental payments required hereunder, and the Department shall be entitled to apply such payment to any debt of the Lessee to the Department that may then exist, as permitted by law, including but not limited to the rent In lieu of the Security Deposit being made in cash, the required hereunder. Department, in its sole discretion, may authorize the Lessee to provide an irrevocable Letter of Credit, in a form provided by the Department, in like amount. The amount of the Security Deposit is subject to adjustment by the Department at any time there is a change in the annual or monthly rentals pursuant to the terms of this Agreement; provided further, that the Department shall have the right to demand an increase in the Security Deposit requirement of up to an additional four months rental to provide the Department with adequate assurance of the Lessee's payment of its obligations, if such assurance is reasonably required because of the Lessee's defaults in the timely payment of rents, fees and charges due hereunder, or because the Department has reason to believe, based on published reports, that the Lessee's future ability to pay such rentals, fees and charges, on a timely basis, is in jeopardy.

3.04 <u>Common Use Service Charges</u>: In addition to the monthly rentals required herein, the Lessee shall pay each month during the term of this Agreement an applicable portion of the costs incurred by the County in providing certain common use services for the benefit of the Lessee, including, but not necessarily limited to, security, servicing of dumpsters provided by the Department pursuant to Article 4.02 (Removal of Trash), compressed air, emergency power, and industrial waste system, as applicable and actually used by or provided to the Lessee. Such costs, including administrative costs, shall be determined by the Department and billed periodically. These service charges shall be adjusted and billed retroactively from time to time based on changes in usages and in costs to the County.

3.05 <u>Rental Rate Review</u>: In the event the Lessee is in possession of the Premises by virtue of this Agreement on October 01, 2015 (and each annual anniversary thereafter), the rental rates stated in Article 3.01 (Annual Rental) above shall be subject to review and adjustment as set forth hereafter. When such rental rate adjustments are approved by the Board of County Commissioners, and new or revised rental rates applicable in whole or in part to the Premises and to all other similarly situated tenants at the airport are established by said Board, the Department shall notify the Lessee in writing of such rates and this Agreement shall be considered and deemed to have been administratively amended to incorporate the revised rental rates effective as of such October 1 date. Payments for any retroactive rental adjustments shall be due upon billing by the Department and payable within ten calendar days of same.

3.06 <u>Double Rental</u>: In the event that the Lessee remains in possession of the Premises beyond the expiration or termination of this Agreement, the Lessee shall be bound by all of the terms and conditions of this Agreement to the same extent as if this Agreement were in full force and effect during the time beyond the expiration date of this Agreement. However, during any such possession of the Premises as a holdover tenant after the County has demanded the return of the Premises, the Lessee shall be liable for double rentals for so long as the Lessee remains in possession after such demand, such rentals to be based upon the rental rates applicable from time to time in whole or in part to the Premises. The parties acknowledge that the Lessee was not a holdover tenant with respect to the Prior Lease.

3.07 <u>Address for Payments</u>: The Lessee shall pay, by mail, all rentals, fees and charges required by this Agreement to the following:

Miami-Dade County Aviation Department Accounting Division P. O. Box 526624 Miami, Florida 33152-6624

Payments may be made by hand-delivery to the offices of the Department during normal working hours.

3.08 Late Payment Charge: In the event the Lessee fails to make any payments, as required to be paid under the provisions of this Agreement, within ten days after same shall become due, interest at the rates established from time to time by the Board of County Commissioners of Miami-Dade County, Florida (currently set at 1½% per month), shall accrue against the delinquent payment(s) from the original due date until the Department actually receives payment. Such interest rate shall apply unless the Board of County Commissioners has established a different rate or a specific provision of federal or state law requires otherwise. The right of the County to require

payment of such interest and the obligation of the Lessee to pay same shall be in addition to and not in lieu of the right of the County to enforce other provisions herein, including termination of this Agreement, and to pursue other remedies provided by law.

3.09 <u>Dishonored Check or Draft</u>: In the event that the Lessee delivers a dishonored check or draft to the County in payment of any obligation arising under the terms of this Agreement, the Lessee shall incur and pay a service fee of TWENTY-FIVE DOLLARS, if the face value of the dishonored check or draft is \$50.00 or less, THIRTY DOLLARS, if the face value of the dishonored check or draft is more than \$50.00 and less than \$300.00, or FORTY DOLLARS, if the face value of the fa

3.10 <u>Gross Revenues</u>: The term "Gross Revenues", as used in this Agreement means all monies paid or payable to, or considerations of determinable value received by the Lessee for sales made, transactions had, or services rendered in providing cargo handling and cargo related services to others as defined in Article 2.04 under this Agreement and from the subleasing of office space or other portions of the Premises, regardless of when or where the order therefor is received, whether paid or unpaid, whether on a cash or credit basis or in consideration of any other thing of value; provided, however, that any taxes imposed by law which are separately stated to and actually paid by a customer and directly payable by the Lessee to a taxing authority and sales refunds shall be excluded therefrom. See 2.04

3.11 <u>Records and Reports</u>: The Lessee shall keep in Dade County, during the term of this Agreement, all books of account, records and reports used in its operation necessary to report Gross Revenues and to calculate the percentage opportunity fees pavable hereunder and as may, from time to time, be required by the Department to document its activities pursuant to this Agreement. All Gross Revenues hereunder shall be accounted for in accordance with generally accepted accounting principles. The form of all such books of account, records and reports shall be subject to the approval of the Department and/or the auditors of the County (one or more of the following: the designated external auditing firm or other certified public accounting firm selected by the Department, the County's Department of Audit and Management Services or auditors of the State of Florida). Recommendations for changes, additions or deletions to such books of account, records and reports by the auditors of the County shall be complied with by the Lessee when requested by the Department. The auditors of the County shall be permitted, during normal working hours, to audit and examine all books of account, records and reports relating to the operations of the Lessee hereunder, including, but not limited to, balance sheets, profit and loss statements, deposit receipts, Florida State Sales Tax Reports and such other documents as may be determined by the Department to be necessary and appropriate; provided, however, that the Lessee shall not be required to retain such records in Dade County, Florida, for more than three years following termination of this Agreement.

3.12 <u>Monthly Report of Gross Revenues</u>: On or before the tenth day following the end of each calendar month throughout the term of this Agreement, the Lessee shall furnish to the Department a statement of monthly Gross Revenues in the form prescribed by the Department. In the event there are no Gross Revenues a monthly report will be submitted stating such.

3.13 Annual Audit Required: Within sixty days of each anniversary of the

commencement date of this Agreement and within sixty days following termination of this Agreement, the Lessee shall, at its sole cost and expense, provide to the Department on an annual (or portion thereof) basis an audit report of monthly Gross Revenues, containing an unqualified opinion, prepared and attested to by an independent certified public accounting firm, licensed in the State of Florida. The report shall include a schedule of Gross Revenues and percentage opportunity fees paid to the County under this Agreement, prepared in accordance with the comprehensive basis of accounting defined under terms of this Agreement and reported in the format as The audit shall be conducted in subsequently prescribed by the Department. accordance with generally accepted auditing standards and include issuance of a management letter, which will contain the findings discovered during the course of the examination, such as recommendations to improve internal controls and other significant matters related to this Agreement. In addition, the audit shall include comprehensive compliance procedures to determine whether the books of account, records and reports were kept in accordance with the terms of this Agreement for the period of examination. The auditor shall report such procedures and findings in a separate letter report to the Department. The last such report shall include the last day(s) of operations.

3.14 <u>Right to Inspect</u>: The Department and the auditors of the County shall have the right, without limitation, to enter upon the Premises at anytime during normal operating hours of the Lessee to: (1) inspect, review, verify and check all or any portion(s) of the Lessee's procedures for recording or compiling Gross Revenue information by day or month; and (2) audit, check, inspect and review all books of account, records, financial reports, financial statements, operating statements, inventory records, copies of State sales tax returns, and work papers relating to the operation of the Lessee, and other pertinent information as may be determined to be needed or desirable by the Department.

3.15 <u>Other Fees and Charges</u>: The Lessee acknowledges that the Board of County Commissioners has or will establish or direct the establishment, from time to time, of various fees and charges for the use of various facilities, equipment and services provided by the County and not leased to or specifically provided to the Lessee hereunder, and procedures relating to the payment of same. The Lessee shall pay, upon billing, for its use of such facilities, equipment and services those fees and charges which are billed monthly. For other fees and charges which are based on usage, the Lessee shall, unless otherwise directed by the Department in writing, report it uses of applicable facilities, equipment and services and pay the applicable fees and charges at such frequency and in such manner as may be prescribed by the Department.

3.16 <u>Monthly Cargo Statistics Report</u>: The Lessee, on or before the 10th day of every month, shall provide to the Accounting Offices of the Department, indicated in Article 3.07 (Address for Payments) hereof, a monthly cargo statistics report, in a form specified by the Department.

3.17 <u>Utilities</u>: Unless the Premises are separately metered and billed directly to the Lessee by the utility company or included in the rental rates and noted in Article 3.01, the Lessee hereby agrees to pay monthly, upon billing by the Department, for utility consumption in the Premises. This monthly charge will be based on a survey conducted by or on behalf of the Department of the utility consumption by the Lessee and/or current nondiscriminatory rates charged others at the Airport. This monthly charge may also be adjusted and billed retroactively, from time to time, based on the changes in consumption and rates. The Lessee shall pay for all utilities used by it. In

the event the Premises are metered and billed to the Department, the Lessee shall pay for utility consumption based on the metered charge. The County shall have no obligation to provide utilities to the Premises other than those existing as of the effective date of this Agreement.

ARTICLE 4

Maintenance And Repair by Lessee

4.01 <u>Cleaning</u>: The Lessee shall, at its sole cost and expense, perform or cause to be performed, services which will at all times keep the Premises and, to the extent in Lessee's control, the adjacent non-leased aircraft ramp clean, neat, orderly, sanitary and presentable. Lessee's obligation as specified above with regard to the ramp areas will be discharged to the extent within Lessee's control; it being acknowledged that persons other than Lessee have access to and use of the ramp areas.

4.02 <u>Removal of Trash</u>: The Lessee shall, at its sole cost and expense, remove from the Premises all trash and refuse which might accumulate and arise from its use of the Premises and the business operations of the Lessee under this Agreement. Such trash and refuse shall be disposed of only in the common use dumpsters provided by the Department or in such other manner approved by the Department.

4.03 Maintenance and Repairs: The Lessee shall repair and maintain in good condition the Premises and all improvements or alterations thereto, except for those items for which the County is responsible pursuant to Article 5 (Maintenance by County). Such repair and maintenance shall include, but not be limited to, refrigerated cooler room panels, cooler doors, cooler system compressor, cooler evaporator, cooler condensers, expansion device and controls, piping system and all cooler room support equipment including dryers, solenoid valves and components for control and safety, fence enclosures, protective guardrails, protective bollards, painting of the interior of the Premises, overhead and personnel doors, windows, assigned pavement (landside and airside) within or appurtenant to the Premises as shown on Exhibit A, equipment, dock levelers, protection bumpers attached to building, furnishings, skylights, fixtures, appurtenances, replacement of light bulbs, ballasts and tubes and the replacement of all broken glass, and shall at all times be based on a standard of care reflecting prudent property management. Maintenance and repairs shall be in quality and class equal to or better than the original work to preserve the Premises in good order and condition. The Lessee shall repair all damage caused by the Lessee and its employees, agents, independent contractors, patrons, servants or invitees. Prior to or at termination of this Agreement, injury done by the installation or removal of furniture and personal property of the Lessee shall be repaired so as to restore the Premises to their original state, except as the Premises may have been altered by the Lessee with the approval of the Department pursuant to Article 7.01 (Alterations), and to guit and surrender up the Premises in the same good order and condition as it was at the commencement of this Agreement, reasonable wear and tear and damage caused by an Act of God excepted; provided however, that such return of the Premises under this Article 4.03 shall not relieve the Lessee of its obligations for damages to the Premises during the term of this Agreement or as provided in Article 13.06 (C) or that may be specifically provided elsewhere in this Agreement.

4.04 <u>Excavation of Land</u>: No excavation of any of the land shall be made, no soil or earth shall be removed from the Premises, and no well of any nature shall be

dug, constructed or drilled on the Premises, except as may be required for environmental monitoring purposes pursuant to Article 8 (Environmental Compliance).

4.05 <u>Water and Sewerage System</u>: The Lessee shall, at its sole cost and expense, operate and maintain all the components of the existing water, sanitary sewerage and storm drainage facilities within the boundaries of the Premises. The Lessee shall not make any alterations or modifications to such facilities without the advance written approval of the Department.

4.06 <u>Industrial Waste Facilities</u>: The Lessee shall be fully responsible for all industrial wastes exiting the Premises and in response thereto shall provide, operate and maintain adequate facilities on the Premises for separating, neutralizing and treating industrial waste and foreign materials and the proper disposal thereof, in accordance with applicable laws, rules and regulations.

4.07 <u>Modifications or Access to Roof</u>: The Lessee covenants that it shall not install, attach, suspend or in any manner modify the roof, its members or structures nor shall it permit any person to walk on the roof or its members without the prior written consent of the Department. In the event the Lessee violates this covenant, the County shall not have any responsibility for any damages to the property of the Lessee or others inside the leased Premises caused by rain or other hazard in any way related to the roof.

4.08 <u>Loading Dock/Platform</u>: If the Premises include a loading dock/platform, the Lessee shall keep such loading dock/platform clean and clear at all times and shall not use the loading dock/platform for the storage of cargo, equipment or any other materials.

4.09 <u>Inspections</u>: The Department and/or its designated representatives shall have the right, during normal working hours, to inspect the Premises to identify those items of maintenance, repair, replacement, modification and refurbishment required of the Lessee or the County, pursuant to Article 5 (Maintenance by County), to keep the Premises in good order and condition. The Lessee shall perform all corrective work required of it, identified in such inspection(s) within 30 days of receipt of written notice from the Department, or if such corrective work cannot reasonably be completed within such 30 day period, Lessee shall commence such work within 30 days of receipt of written notice and complete such work as expeditiously thereafter as reasonably possible. Trash and debris problems shall be corrected within 24 hours following receipt of either oral or written notice from the Department.

4.10 <u>Failure to Maintain</u>: If it is determined by the Department that the Lessee has failed to properly clean, remove trash and refuse, maintain, repair, replace and refurbish the Premises as required by this Article 4 (Maintenance and Repair by Lessee), the Department shall provide to the Lessee a list of deficiencies, reflecting the amount of time to be reasonably allowed for the Lessee to correct same. If the Lessee fails to correct such deficiencies within the time allowed and has not registered an objection as to its obligation to do so, the Department, following 10 days further notice to the Lessee, may enter upon the Premises and perform all work, which, in the judgment of the Department, may be necessary, and the County shall add the cost of such work, plus 25% for administrative costs, to the rent due hereunder on the first day of the month following the date of such work, and such cost shall be and constitute a part of the rent. Subsequent to receipt of the further notice of intent to perform repairs or cleanup from the Department, the Lessee shall not undertake performance of such repairs or cleanup without specific prior written authorization from the Department, not

to be unreasonably withheld if the Department has not yet contracted for or commenced the needed repairs or cleanup.

ARTICLE 5

Maintenance by County

5.01 <u>Maintenance of Water, Sewer and Drainage Systems</u>: The County shall operate and maintain, in good condition, all components of the existing water, sanitary sewerage and storm water drainage facilities that lie outside the boundaries of the Premises.

5.02 <u>Building Maintenance</u>: The County shall maintain, in good condition, the roof, exterior walls and the building's structural support system. The County shall maintain all common areas, including, hallways, vestibules, restrooms, elevators, in said areas, landscaping, irrigation system and the Building's central air conditioning system. If any of such facilities are damaged or destroyed by the operations of the Lessee, the Department shall require Lessee to make all necessary repairs or replacements or else the Department shall make the necessary repairs or replacements and bill the Lessee for the costs of same, plus 25% for administrative costs in the manner specified in Article 4.10 (Failure to Maintain) hereof. The County shall provide janitorial services to all common areas. The County shall not maintain any doors, including personnel, overhead and cargo doors, or any windows within or directly leading into the leased Premises.

5.03 <u>County Maintenance Subject to Certain Conditions</u>: Such maintenance by the County may be subject to interruption caused by repairs, strikes, lockouts, labor controversies, inability to obtain fuel or power or parts, accidents, breakdowns, catastrophes, national or local emergencies, acts of God, and other conditions beyond the control of the County. Upon any such happening, the Lessee shall have no claim for damages for the County's failure to furnish or to furnish in a timely manner any such maintenance; provided, however, that the Department, in its reasonable discretion, may provide a rent abatement for that portion of the Premises rendered unusable for the period of time that the County is unable to make the repairs required by Articles 5.01 (Maintenance of Water, Sewer and Drainage Systems) and 5.02 (Building Maintenance). The County shall exercise reasonable diligence to remedy and/or cure any such interruptions, to the extent such interruptions are within the County's control.

ARTICLE 6 Regulations, Licenses and Permits

6.01 Rules and Regulations - General:

(A) (1) The Lessee shall comply with all Ordinances of the County, including the Rules and Regulations of the Department, Chapter 25, Code of Miami-Dade County, Florida, as the same may be amended from time to time, Operational Directives issued thereunder, all additional laws, statutes, ordinances, regulations and rules of the Federal, State and County Governments, and any and all plans and programs developed in compliance therewith, which may be applicable to its operations or activities under this Agreement, specifically including, without limiting the generality hereof, Federal air and safety laws and regulations and Federal, State and County environmental laws.

(2) During the renovation period of the work to be performed by Lessee hereunder, and at any time Lessee performs any work on the Premises, Lessee shall comply with all design and construction requirements of MDAD, including, but not limited to, MDAD's Tenant Airport Construction (Reimbursable or Non-Reimbursable) requirements ("TAC"), as they may be amended from time to time. All design and construction work must be approved in advance by MDAD, to the extent set forth in such TAC requirements.

- (B) <u>Permits and Licenses</u>:
 - (1) The Lessee, at its sole cost and expense, shall be liable and responsible for obtaining, paying for, maintaining on a current basis, and fully complying with, any and all permits, licenses and other governmental authorizations, however designated, as may be required, at any time throughout the entire term of this Agreement, by any Federal, State, or County governmental entity or any judicial body having jurisdiction over the Lessee or the Lessee's operations and activities, for any activity of the Lessee conducted on the Premises and for any and all operations conducted by the Lessee, including insuring that all legal requirements, permits and licenses necessary for or resulting, directly or indirectly, from the Lessee's operations and activities on the Premises have been obtained and are being fully complied with.
 - (2) Such permits and licenses shall include, but not be limited to, a Certificate of Use and Occupancy and any required Industrial Waste or Operating Permits from Department of Regulatory and Economic Resources. At the inception of this Agreement, the Lessee shall provide to the Department evidence that it has obtained the Certificate of Use and Occupancy and, as applicable, the appropriate operating Waste Permit(s). Upon written request of the Department, the Lessee shall provide to the Department copies of any permits and licenses, and applications therefor, which the Department may request.
- Violations of Rules and Regulations: Subject to the County's obligations (C) as confirmed in Article 8 (Environmental Compliance), Lessee agrees to pay on behalf of the County any penalty, assessment or fine issued against the County, or to defend in the name of the County any claim, assessment or civil action, which may be presented or initiated by any agency or officer of the Federal, State or County governments, based upon a claim or allegation that the Lessee, its agents, employees, invitees, or trespassers have violated any law, ordinance, regulation, rule or directive described in Article 6.01 above or any plan or program developed in compliance therewith. The Lessee further agrees that the substance of this Article 6.01 (Rules and Regulations - General) above shall be included in every sublease, contract and other agreement, which the Lessee may enter into related to its operations and activities under this Agreement and that any such sublease, contract and other agreement shall specifically provide that "Miami-Dade County, Florida is a third party beneficiary of this and related provisions." This provision shall not

constitute a waiver of any other conditions of this Agreement prohibiting or limiting assignments, subletting or subcontracting.

Aircraft Noise Abatement Regulations Compliance: The Lessee hereby 6.02 specifically acknowledges its awareness that the noise generated by aircraft, while on the ground and in flight, may cause annoyance to residential areas in proximity to the Airport and/or under or near flight corridors serving the Airport and that the County, as proprietor of the Airport, may, therefore, from time to time adopt certain lawful policies. procedures and/or regulations, not inconsistent with the Airport Noise and Capacity Act of 1990, unless otherwise approved by the Federal Aviation Administration, and not inconsistent with aircraft safety or the Lessee's rights hereunder, intended to abate the effects of aircraft noise. Said policies, procedures and/or regulations may deal with, but are not necessarily limited to, nighttime engine run-ups, preferential runway usage, aircraft arrival and departure patterns, use of displaced runway thresholds, and the like. The Lessee specifically understands and agrees that a violation of such noise abatement policies, procedures and/or regulations may result in the arrest or citation of the offending party, with the imposition of fines, and that a violation of same shall constitute a material breach hereunder and may result in termination of this Agreement pursuant to the provisions hereof.

ARTICLE 7

Alteration of Premises and Erection of Signs

7.01 <u>Alterations</u>: The Lessee shall not alter the Premises in any manner whatsoever without prior written approval of the Department. In the event the Lessee is given approval to make any alterations to the Premises, the Lessee shall fully comply with the terms and conditions of the approval document from the Department's Tenant Airport Construction Program in effect, and Article 6 (Regulations, Licenses and Permits). The Lessee's failure to do so shall constitute a default pursuant to Article 13.03 (Other Defaults) hereof.

The Lessee acknowledges that the Premises have been provided by the County with a basic level of "tenant finishes" and that, therefore, any additional or special finishes desired by the Lessee shall be at the Lessee's sole cost and expense and that rentals, pursuant to Article 3.01 (Annual Rental), shall be payable during the period the Lessee is installing such additional or special finishes.

7.02 <u>Signage</u>: The Lessee shall not erect, maintain or display any identifying signs or any advertising matter, of any type or kind which is visible to the public, on the exterior of the building. All signage within the building leading into the Premises shall require the advance written approval of the Department.

ARTICLE 8

Environmental Compliance

- 8.01 <u>Definitions:</u> For purposes of this Agreement, the following additional definitions apply:
- (A) "Baseline Environmental Conditions" means the presence or release of Hazardous Materials, at, on, under, or from the Premises prior to Lessee's Occupancy Date, the presence or release of which was not caused by Lessee or Lessee's agents, employees, contractors, invitees or Trespassers. Solely for purposes of this Agreement, it shall be presumed that the Baseline Environmental Conditions consist of the conditions

identified in any existing (as of the date of this Agreement) Miami-Dade County maintained records, including contamination assessment reports and any other technical reports, data bases, remedial action plans, the Baseline Audit or the presence, discharge, disposal or release of any other Hazardous Materials originating prior to the Occupancy Date that comes to be located on the Premises and not caused by Lessee or Lessee's agents, employees, contractors, invitees or Trespassers.

- (B) "Environmental Claim" means any investigative, enforcement, cleanup, removal, containment, remedial or other private, governmental or regulatory action at any time threatened, instituted or completed pursuant to any applicable Environmental Requirement, against Lessee with respect to its operations at Miami International Airport or against or with respect to the Premises or any condition, use or activity on the Premises (including any such action against County), and any claim at any time threatened or made by any person against Lessee with respect to its operations at Miami International Airport or against or with respect to its operations at Miami International Airport or against or with respect to its operations at Miami International Airport or against or with respect to the Premises or any condition, use or activity on the Premises (including any such claim against County), relating to damage, contribution, cost recovery, compensation, loss or injury resulting from or in any way arising in connection with any Hazardous Material or any applicable Environmental Requirement.
- "Environmental Law" means any applicable federal, state or local law, (C) statute, ordinance, code, rule, or regulation, or license, authorization, decision, order, injunction, or decree, any of which may be issued by a judicial or regulatory body of competent jurisdiction, or rule of common law including, without limitation, actions in nuisance or trespass, and any judicial or agency interpretation of any of the foregoing, which pertains to health, safety, any Hazardous Material, or the environment (including but not limited to ground or air or water or noise pollution or contamination, and underground or aboveground tanks) and shall include without limitation, the Solid Waste Disposal Act, 42 U.S.C. § 6901 et seq.; the Comprehensive Environmental Response, Compensation and Liability Act of 1980, 42 U.S.C § 9601 et seq. ("CERCLA"), as amended by the Superfund Amendments and Reauthorization Act of 1986 ("SARA"); the Hazardous Materials Transportation Act 49 U.S.C. § 1801 et seq.; the Federal Water Pollution Control Act, 33 U.S.C.§ 1251 et seq.; the Clean Air Act 42 U.S.C. § 7401 et seq.; the Toxic Materials Control Act 15 U.S.C. § 2601 et seq.; the Safe Drinking Water Act, 42 U.S.C. § 300f et seq.; Chapters 403, 376 and 373, Florida Statutes; Chapters 24 and 25 of Miami-Dade County Code, and any other applicable local, state or federal environmental statutes, codes, or ordinances, and all rules, regulations, orders and decrees now or hereafter promulgated under any of the foregoing, as any of the foregoing now exist or may be changed or amended or come into effect in the future.
- (D) "Environmental Requirement" means any Environmental Law, or any agreement or restriction entered into or applicable by law (including but not limited to any condition or requirement imposed by any insurance or surety company), as the same now exists or may be changed or amended or come into effect in the future, which pertains to health, safety, any Hazardous Material, or the environment, including but not limited to

ground or air or water or noise pollution or contamination, and underground or aboveground tanks.

With regard to any environmental agreements or restrictions entered into by MDAD, MDAD agrees to use reasonable efforts to apprise the Miami Airport Affairs Committee (the "MAAC"), which represents the interests of the airlines at the Airport, of any such agreements or restrictions and to obtain the MAAC's input into such items.

- (E) "Hazardous Material" means any substance, whether solid, liquid or gaseous, which is listed, defined or regulated as a hazardous substance, a hazardous waste or pesticide, or otherwise classified as hazardous or toxic, in or pursuant to any applicable Environmental Requirement; or which is or contains asbestos, radon, any polychlorinated biphenyl, urea formaldehyde foam insulation, explosive or radioactive material, or motor fuel or other petroleum hydrocarbons; or which causes or poses a threat to cause contamination or a nuisance on the Premises, any adjacent Premises or a hazard to the environment or to the health or safety of persons on the Premises or Other Airport Property.
- (F) "Initial Construction Period" means for any lease which contemplates construction or renovation for Premises not previously occupied in whole or in part by Lessee under this Agreement and/or any previous Agreement, a period of time not to exceed six (6) months commencing with the date on which Lessee breaks ground on the Premises for construction of foundations or commences such renovation.
- (G) "Occupancy Date" means the date Lessee first entered, occupied or took possession of the Premises under any written or verbal agreement.
- (H) "On" or "in" when used with respect to the Premises or any premises adjacent to the Premises, means "on, in, under, above or about."
- (I) "Other Airport Property" means property on the Airport occupied or used by Lessee, or upon which Lessee performs operations, but which is not subject to a lease, sublease or other legal agreement governing the terms of Lessee's occupation, use or operations at such property.
- (J) "Recognized Environmental Condition" shall have the meaning set forth in ASTM E 1527-05, Section 1.1.1, as such provision may be amended or superseded from time to time.
- (K) "Release" means any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping or disposing into the environment.
- (L) "Remediation" means any investigation, clean-up, removal action, remedial action, restoration, repair, response action, corrective action, monitoring, sampling and analysis, installation, reclamation, closure, or post-closure in connection with the suspected, threatened or actual release of Hazardous Materials.
- (M) "Trespassers" means third parties who have entered the Premises and whose actions while on the Premises have resulted in Release of

Hazardous Materials directly onto the Premises or onto other Airport Property accessed by such persons through the Premises. Notwithstanding the foregoing, for purposes of this Agreement, Trespassers shall not include those third parties whose actions took place off of the Premises and which resulted in the presence of Hazardous Materials on the Premises due to the migration of Hazardous Materials from that off-Premises location.

Lessee's Industrial Classification: Lessee represents and warrants to 8.02 County that Lessee's Standard Industrial Classification ("SIC") code number, as published in the most recent SIC Manual from the United States Office of Management Federal Return is used on Lessee's Tax and Budget, and as 47 2070117

8.03 <u>Lessee's Acceptance of the Risks and Condition of Premises As-Is</u>: Lessee agrees that the Premises shall be leased and delivered to Lessee in its current "as-is/with all faults" condition (but it is not intended by this provision that County be relieved from its duties expressly set forth in this Agreement or any other applicable agreement). Lessee hereby, warrants, covenants, agrees, and acknowledges that:

- (A) Hazardous Materials may be present on the Premises and Other Airport Property. The County is currently engaged in a significant environmental remediation program at MIA.
- (B) Under Article 8.06 below, Lessee is provided the opportunity to conduct an independent investigation of the Premises and the physical condition thereof, including the potential presence of any Hazardous Materials on or about the Premises. Lessee's report on the investigation, if any such report has been prepared, has been provided to the County and is listed in Exhibit K attached to this Agreement. Whether Lessee has conducted such an investigation or not, Lessee is willing to proceed with this Agreement notwithstanding the environmental conditions of the premises or the properties surrounding the premises, subject to Lessee's right to terminate this Agreement as otherwise provided herein.
- (C) Because of the possible presence of environmental contaminants on the Premises or other Airport property, County has made no express, implied, or other representations of any kind with respect to the suitability or usability of the Premises or other Airport Property, or any improvements appurtenant thereto, including, without limitation, the suitability or usability of any building materials, building systems, soils or groundwater conditions (due to the presence of Hazardous Materials in, on, under, or about the Premises or other Airport property), for Lessee's proposed or intended use, and Lessee has relied solely on Lessee's own inspection and examination of such matters.
- (D) Except as to County's obligations set forth in this Article or elsewhere in this Agreement, Lessee expressly assumes the risk that Hazardous Materials that are or may be present on the Premises at the commencement of this Agreement may affect the suitability or usability of the Premises for Lessee's proposed or intended use. Lessee agrees that, except to the extent of County's Remediation obligations provided in this Article 8, or any other discharge, disposal or release of Hazardous Materials or violation of Environmental Requirements, caused by County,

its agents, employees or contractors and except with respect to Baseline Environmental Conditions, County shall have no responsibility or liability with respect to any Hazardous Materials on the Premises. Notwithstanding the foregoing, in no event shall County be liable to Lessee for damages relating to physical or personal injury, business interruptions relocation costs or any other cost (other than a cost for which County is liable under this Article 8) resulting from the presence of Hazardous Materials on the Premises at any time during this Agreement.

8.04 Responsibilities for Hazardous Materials.

(A) Unless the parties agree otherwise in writing, the County shall conduct response actions mandated by existing Environmental Requirements applicable to the County for (i) Hazardous Materials disclosed in the Lessee Audit to the extent required by Article 8.06 and (ii) Baseline Environmental Conditions, provided however that:

1) To the extent this Agreement covers Premises not previously occupied by Lessee and if this Agreement contemplates construction or renovation by the Lessee, any Hazardous Material discovered during the Initial Construction Period as defined in Article 8.01(F), shall be presumed to be a Baseline Environmental Condition under this Agreement except to the extent the Aviation Department demonstrates to the satisfaction of Lessee by written notice setting forth the Aviation Department's explanation as to why the Hazardous Material originated from a discharge, disposal or release that was caused by Lessee, Lessee's agents, employees, contractors, invitees or Trespassers. Should Lessee determine that such a demonstration has not been made to Lessee's satisfaction, County may invoke the dispute resolution provision of 8.16. Until such time as the parties reach an agreement or such time as the dispute is otherwise resolved, responsibility for such Hazardous Material shall remain with the Aviation Department.

2) To the extent this Agreement covers premises previously occupied by Lessee, and except for Baseline Environmental Conditions, Remediation of any Hazardous Material discovered on the Premises shall be the responsibility of the Lessee, except to the extent that Lessee demonstrates to the satisfaction of Aviation Department Management by written notice setting forth Lessee's explanation as to why the Hazardous Material originated from (1) a discharge, disposal or release outside of the Premises, unless such discharge, disposal or release was caused by Lessee, Lessee's agents, employees, contractors or invitees (2) a discharge, disposal or release of Hazardous Material on the Premises prior to the date upon which Lessee first occupied the premises and not caused by Lessee or Lessee's agents, employees, contractors or invitees; or (3) a discharge, disposal or release caused by the County, its agents, employees, contractors or any third party. Should the Aviation Department determine that such a demonstration has not been made to Aviation Department's satisfaction, Lessee may invoke the dispute resolution provision of 8.16. Until such time as the parties reach an agreement or such time as the dispute is otherwise resolved, responsibility for such Hazardous Material shall remain with Lessee.

(B) County's responsibility for Remediation under this Article 8.04 shall be

limited to the Recognized Environmental Conditions required to be remediated under applicable Environmental Requirements. If County is permitted to leave any Hazardous Material in place under applicable Environmental Requirements, County shall have the option of so doing, unless a governmental authority requires at any time the removal of Hazardous Materials for Lessee to be able to continue with construction or occupancy of the Premises. The County shall notify Lessee of any such decision to leave Hazardous Material in place.

- (C) (1) To the extent they exist, the County has made available to Lessee a listing of contamination assessment reports and remedial action plans regarding any soil and groundwater contamination at the Premises. Such list is found on Exhibit K to this Agreement. The County may have already installed or may have plans to install remediation systems to clean up the contamination described in such reports to the extent they exist. Lessee agrees that during the term of the Agreement, County's authorized representatives shall have the right to enter the Premises in order to operate, inspect, maintain, relocate and replace any such installed systems. Without limiting the generality of the foregoing, the County shall have the right to: (a) install, use, monitor, remove (or, in connection with monitoring wells, abandon in place in accordance with applicable governmental regulations) soil borings, treatment systems, pumps, monitoring wells, and associated equipment; (b) construct, maintain, and ultimately remove various mechanical devices designed to aid in the monitoring and remediating effort; and (c) undertake such related activities as the Aviation Department or other governmental authorities may require or recommend, utilizing such methods as the Aviation Department or the applicable governmental authorities may elect in order to remediate the contamination described in any such reports.
 - (2) County shall utilize reasonable efforts to minimize any disturbance of the Lessee's use of the Premises caused by any Remediation it undertakes and shall provide Lessee prior written notice of such Remediation. Lessee agrees that it shall not unreasonably interfere with or obstruct such Remediation. County and Lessee each agree to take such action as may be reasonable to coordinate their operations so as to minimize any interference with the other party. If vehicles, equipment, or materials belonging to the Lessee have to be temporarily relocated to permit the Remediation to be performed, the Lessee will effect such relocation at no expense to the County. Attached to Exhibit K is a site sketch of the Premises describing any existing or currently planned Remediation equipment and depicting the current and proposed future location of such equipment.
 - (3) If Remediation equipment or materials need to be temporarily stored in a secure location on the Premises, the Lessee will provide reasonable storage inside the building on the Premises for such equipment and materials at no expense to the County, provided, however, that Lessee shall bear no liability and otherwise shall have no responsibility for any theft of and/or damage to such equipment or materials so stored, to the extent Lessee took reasonable measures to prevent, such theft and/or damage and such theft and/or damage was not caused by Lessee or Lessee's employees. To the extent that water and electrical service within the Premises are not metered and the Lessee does not pay for such

services directly, the Lessee will provide the County with water and electrical service from the Premises in connection with the Remediation, without charge. The Lessee acknowledges the Remediation may be conducted at the locations depicted on the site sketch attached to Exhibit K at any time during the term of the Agreement and may continue until such time as a no further action letter is obtained from the appropriate regulatory authorities.

8.05 Baseline Audit: The County has provided Lessee with a copy of an environmental audit of the Premises, conducted to identify any Recognized Environmental Conditions associated with the Premises, which audit may include analyses of soil and groundwater samples (the initial "Baseline Audit"). Except to the extent Lessee previously occupied the Premises, the County shall be responsible for any Recognized Environmental Conditions within the meaning of ASTM E 1527-05, or most recent version, disclosed by the Baseline Audit. Except to the extent Lessee previously occupied the Premises, Lessee may terminate this Agreement within sixty (60) days of receipt of the Baseline Audit if Lessee, in its sole discretion, determines that the Recognized Environmental Conditions disclosed in such Baseline Audit are unacceptable. To the extent Lessee previously occupied the Premises, Lessee, subject to its right to invoke the dispute resolution provision of 8.16, shall be responsible for all Recognized Environmental Conditions disclosed in the Baseline Audit, which are not otherwise Baseline Environmental Conditions, unless Lessee demonstrates to the County's satisfaction that the Recognized Environmental Conditions originated from (1) a discharge, disposal or release outside of the Premises, unless such discharge, disposal or release was caused by Lessee, Lessee's agents employees, contractors or invitees or Trespassers; or (2) a discharge, disposal or release of Hazardous Material on the Premises prior to Lessee's first occupancy of the Premises and not caused by Lessee, Lessee's agents, employees, contractors or invitees.

8.06 Lessee Audit: Lessee, at its sole cost and expense, shall have the right to conduct, within sixty (60) days from the receipt of the Baseline Audit, an environmental inspection of the Premises (the "Lessee Audit"), through an independent environmental consultant approved in writing by County, such approval not to be unreasonably withheld or delayed. If Lessee elects to conduct a Lessee Audit, it shall furnish County a copy of the Lessee Audit within thirty (30) days of Lessee's receipt of the Lessee Audit. The purpose of the Lessee Audit is to determine whether there are present on the Premises any Recognized Environmental Conditions not identified in the Baseline Audit, any previous audits, or any contamination assessment reports or remedial action plans, to the extent any such documents exist. Within thirty (30) days of receipt of such Lessee Audit, the County shall notify Lessee if it disputes the Recognized Environmental Conditions or the delineation of any subsurface conditions described in the Lessee Audit. If the Lessee Audit reveals any Recognized Environmental Conditions or delineates any subsurface contamination not disclosed in any contamination assessment reports, remedial action plans, or the Baseline Audit, and which are not otherwise considered Baseline Environmental Conditions under the terms of this Agreement, then, except to the extent that Lessee previously occupied the Premises, the County, at its option, shall: (i) allow Lessee to terminate the Agreement, without penalty, within sixty (60) days of receipt of such notice of dispute from the County; or (ii) notify Lessee that it has agreed to be responsible for such Recognized Environmental Conditions and delineated subsurface contamination to the same extent as the County is responsible for the Recognized Environmental Conditions and subsurface contamination disclosed in any contamination assessment reports, remedial action plans and the Baseline Audit. If the County allows Lessee to terminate the Agreement and Lessee elects not to terminate, Lessee's failure to terminate shall constitute a

waiver of 1) Lessee's rights to terminate its obligations under this Agreement as to any findings in such Lessee Audit, except as to its right to cancel the lease on thirty (30) days notice under Article 1.01 (B) and, 2) as provided in Article 8.04, any claim it may have against the County with respect either to Recognized Environmental Conditions and subsurface contamination disclosed in such Lessee Audit. To the extent the Lessee previously occupied the Premises, Lessee shall be responsible for all Recognized Environmental Conditions disclosed in the Lessee Audit that are not Baseline Environmental Conditions unless Lessee demonstrates to the satisfaction of Aviation Department Management by written notice setting forth Lessee's explanation why the Recognized Environmental Conditions originated from (1) a discharge, disposal or release outside of the Premises, unless such discharge, disposal or release was caused by Lessee, Lessee's agents, employees, contractors, or invitees; (2) a discharge, disposal or release of Hazardous Material on the Premises prior to the Occupancy Date and not caused by Lessee or Lessee's agents, employees, contractors or invitees ; or (3) a discharge, disposal or release caused by County or third party. Should the Aviation Department determine that such a demonstration has not been made to Aviation Department's satisfaction, Lessee may invoke the dispute resolution provision of 8.16 Until such time as the parties reach an agreement or until such time as the dispute is otherwise resolved, responsibility for such Recognized Environmental Condition shall remain with Lessee.

8.07 <u>Environmental Maintenance of Premises</u>: Except for the obligations of the County under this Article 8, Lessee shall, at its sole cost and expense, keep, maintain and use the Premises, and operate within the Premises at all times, in compliance with all applicable Environmental Laws, and shall maintain the Premises in good and sanitary order, condition, and repair.

8.08 Lessee's Use of Hazardous Materials: Exhibit Z is a complete list of all Hazardous Materials which Lessee currently intends to use on the Premises or Other Airport Property during the term of the Agreement which have been approved by the County, and the use, storage and transportation of which on or about the Premises shall not be subject to County's approval or objections. Except for those Hazardous Materials listed on Exhibit Z, Lessee shall not use, store, generate, treat, transport, or dispose of any Hazardous Material on the Premises or Other Airport Property without first providing the County thirty (30) days written notice prior to bringing such Hazardous Material upon the premises. To the extent certain Hazardous Materials are be needed to be used by Lessee on a non-routine basis, such as for emergency repairs, Lessee may provide such notice within twenty-four (24) hours of bringing such Hazardous Material upon the premises. Notwithstanding the foregoing, County may object to the use of any previously approved Hazardous Material should County reasonably determine that the continued use of the Hazardous Material by Lessee presents a material increased risk of site contamination, damage or injury to persons, Premises, resources on or near the Premises or Other Airport Property, or noncompliance due to a change in regulation of such Hazardous Material under applicable Environmental Law. Upon County's objection, Lessee shall immediately remove the Hazardous Material from the site. This section 8.08 shall not apply to Hazardous Materials which are not used, generated, treated or disposed of by Lessee but which are otherwise transported by Lessee solely in the course of Lessee's business, such as cargo operations, provided, however, Lessee shall remain responsible under Article 8.14 and 8.15 for such transported Hazardous Materials . County's objection or failure to object to the use, storage, generation, treatment, transportation, or disposal of Hazardous Material under this paragraph, or the exclusion of certain Hazardous Materials under this paragraph, shall not limit or affect Lessee's obligations under this Agreement, including Lessee's duty to remedy or remove releases or threatened releases; to comply with applicable

JAY.

Environmental Law and/or Environmental Requirements relating to the use, storage, generation, treatment, transportation, and/or disposal of any such Hazardous Materials; or to indemnify County against any harm or damage caused thereby. Lessee shall promptly and completely answer periodic questionnaires from the County concerning Lessee's practices regarding the generation, use, storage, and disposal of Hazardous Materials under this Agreement.

8.09 Entry by County:

(A) Notwithstanding any other right of entry granted to County under this Agreement, and subject to the requirements set forth in Article 8.09(B), MDAD shall have the right, at its own expense and upon reasonable notice, to enter the Premises or to have consultants enter the Premises throughout the Term of this Agreement for the purposes of: (1) determining whether the Premises are in conformity with applicable Environmental Law; (2) conducting an environmental review or investigation of the Premises; (3) determining whether Lessee has complied with the applicable environmental requirements of this Agreement; (4) determining the corrective measures, if any, required of Lessee to ensure the safe use, storage, and disposal of Hazardous Materials; or (5) removing Hazardous Materials (except to the extent used, stored, generated, treated, transported, or disposed of by Lessee in compliance with applicable Environmental Requirements and the terms of this Agreement). Lessee agrees to provide access and reasonable assistance for such inspections. MDAD shall use its best efforts to reasonably minimize interruptions of business operations on the Premises.

Such inspections may include, but are not limited to, entering the (B) Premises or adjacent property with drill rigs or other machinery for the purpose of obtaining laboratory samples of environmental conditions or soil or groundwater conditions. Lessee shall have the right to collect split samples of any samples collected by MDAD, MDAD shall not be limited in the number of such inspections during the Term of this Agreement MDAD will conduct such inspections during Lessee's normal business hours, but MDAD may conduct such inspections in other than normal business hours if the circumstances so require. For inspections conducted by MDAD, MDAD agrees to provide Lessee with reasonable notice (not less than twenty four (24) hours) prior to inspecting the Premises; provided however, that such notice period shall not apply under circumstances in which MDAD reasonably determines that there exists an immediate threat to the health, safety, or welfare of any persons. Based on the results of such inspections, should MDAD reasonably determine that Hazardous Materials have been released, discharged, stored, or used on the Premises in violation of the terms of this Agreement, Lessee shall, in a timely manner, at its expense, remove such Hazardous Materials in a manner not inconsistent with applicable Environmental Law and otherwise comply with the reasonable recommendations of MDAD and any regulatory authorities related to the results of such inspections. The right granted to MDAD herein to inspect the Premises shall not create a duty on MDAD's part to inspect the Premises, nor liability of MDAD for Lessee's use, storage, or disposal of Hazardous Materials, it being understood that Lessee shall be solely responsible for all liability in connection therewith. MDAD shall provide the results of such inspections to the Lessee in a timely manner if requested to do so in writing. Nothing herein shall be construed to limit, restrain, impair or interfere with County's regulatory authority to conduct inspections and/or the manner in which it conducts such inspections. Lessee shall not be liable or otherwise responsible for any property damage to the Premises or injury to any person caused by County, its agents or consultants during County's inspection under this Section 8.09.

8.10 <u>Permits and Licenses</u>: The Lessee warrants that it will secure at the times required by issuing authorities all applicable permits or approvals that are required by any governmental authority having lawful jurisdiction to enable Lessee to conduct its obligations under this Agreement. Upon written request, Lessee shall provide to County copies of all permits, licenses, certificates of occupancy, approvals, consent orders, or other authorizations issued to Lessee under applicable Environmental Requirements, as they pertain to the Lessee's operations on or use of the Premises or Other Airport Property.

8.11 Notice of Discharge to County:

In the event of: (i) the happening of any material event involving the spill, release, (A) leak, seepage, discharge or clean up of any Hazardous Material on the Premises or Other Airport Property in connection with Lessee's operation thereon; or (ii) any written Environmental Claim affecting Lessee from any person or entity resulting from Lessee's use of the Premises or Other Airport Property, then Lessee shall immediately notify County orally within twenty-four (24) hours and in writing within three (3) business days of said notice. If County is reasonably satisfied that Lessee is not promptly commencing the response to either of such events. County shall have the right but not the obligation to enter onto the Premises or to take such other actions as it shall deem reasonably necessary or advisable to clean up, remove, resolve or minimize the impact of or otherwise deal with any such Hazardous Material or Environmental Claim following receipt of any notice from any person or any entity having jurisdiction asserting the existence of any Hazardous Material or an Environmental Claim pertaining to the Premises, which if true, could result in an order, suit or other action against the County. If Lessee is unable to resolve such action in a manner which results in no liability on the part of County, all reasonable costs and expenses incurred by County shall be deemed additional rent due County under this Agreement and shall be payable by Lessee upon demand, except to the extent they relate to a Baseline Environmental Condition.

(B) With regard to any reporting obligation arising out of Lessee's operations or during the Agreement, Lessee shall timely notify the State of Florida Department of Environmental Protection, all Miami-Dade County pertinent regulatory agencies, and the United States Environmental Protection Agency, as appropriate, with regard to any and all applicable reporting obligations while simultaneously providing written notice to County.

(C) Within sixty (60) days of execution of this Agreement, Lessee shall submit to County an emergency action plan/contingency plan setting forth in detail Lessee's procedures for responding to spills, releases, or discharges of Hazardous Materials. The emergency action plan/contingency plan shall identify Lessee's emergency response coordinator and Lessee's emergency response contractor.

8.12 <u>Reports to County</u>: For any year in which any Hazardous Materials have been used, generated, treated, stored, transported or otherwise been present on or in the Premises, (or on or in other Airport property for purposes related to Lessee's operations on the Premises), Lessee shall provide County with a written report listing: the Hazardous Materials which were present on the Premises or other Airport property; all releases of Hazardous Material that occurred or were discovered on the Premises or other Airport property and which were required to be reported to regulatory authorities under applicable Environmental Laws; all enforcement actions related to such Hazardous Materials, including all, consent agreements or other non-privileged documents relating to such enforcement actions during that time period. In addition, Lessee shall provide County with copies of any reports filed in accordance with the Emergency Planning and Community Right to Know Act (EPCRA) and shall make available for review upon request by County copies of all manifests for hazardous wastes generated from operations on the Premises. Lessee shall provide the report required under this section to the County by April 1 of each year for the preceding calendar year.

Periodic Environmental Audits: Lessee shall establish and maintain, at its 8.13 sole expense, a system to assure and monitor its continued compliance on the Premises with all applicable Environmental Laws, which system shall include, no less than once each year a detailed review of such compliance (the "Environmental Audit") by such consultant or consultants as County may approve, which approval shall not be unreasonably withheld, delayed or conditioned. Alternatively, if the Aviation Department approves, which approval shall not be unreasonably withheld, delayed, or conditioned, such Environmental Audit may be conducted by Lessee's personnel but in either case Lessee shall provide County with a copy or summary of its report of its annual Environmental Audit, which shall be consistent with ASTM's "Practice for Environmental Regulatory Compliance Audits" which shall include in its scope the items listed in Exhibit E hereto or other recognized format approved by County. If the Environmental Audit indicates any unresolved violation of any applicable Environmental Law and/or Environmental Requirements, Lessee shall, at the request of County, provide a detailed review of the status of any such violation within thirty (30) days of the County's request.

8.14 Remediation of Hazardous Material Releases: If Lessee or Lessee's agents, employees, contractors, invitees or trespassers cause any Hazardous Materials to be released, discharged, or otherwise located on or about the Premises or Other Airport Property during the term of this Agreement ("Hazardous Material Release"), Lessee shall promptly take all actions, at its sole expense and without abatement of rent, as are reasonable and necessary to return the affected portion of the Premises or Other Airport Property and any other affected soil or groundwater to their condition existing prior to the Hazardous Material Release in a manner not inconsistent with applicable Environmental Law. County shall have the right to approve all such remedial work, including, without limitation: (i) the selection of any contractor or consultant Lessee proposes to retain to investigate the nature or extent of such Hazardous Material Release or to perform any such remedial work; (ii) any reports or disclosure statements to be submitted to any governmental authorities prior to the submission of such materials; and (iii) any proposed remediation plan or any material revision thereto prior to submission to any governmental authorities. The County's approvals shall not be unreasonably withheld, conditioned or delayed. Notwithstanding the foregoing, County's prior consent shall not be necessary if a Hazardous Material Release poses an immediate threat to the health, safety, or welfare of any persons and, despite Lessee's best efforts, it is not practicable to obtain County's consent before taking remedial action to abate such immediate threat, provided that: (a) Lessee shall notify County as soon as possible and shall thereafter obtain County's consent as otherwise provided in this paragraph; and (b) Lessee shall take only such action as may be necessary or appropriate to abate such immediate threat and shall otherwise comply with the provisions of this paragraph. In addition to any rights reserved by County in this Agreement, County shall have the right, but not the obligation, to participate with Lessee, Lessee's consultants and Lessee's contractors in any meetings with representatives of the governmental authorities and Lessee shall provide County reasonable notice of any such meetings. All remedial work shall be performed in compliance with all applicable Environmental Laws. The County's consent to any remedial activities undertaken by Lessee shall not be withheld so long as County reasonably determines that such activities will not cause any material adverse long-term or short-term effect on the Premises, or other adjoining property owned by County.

Lessee's obligations in this section do not apply to Baseline Environmental Conditions.

8.15 Indemnity: Lessee shall indemnify, defend (with counsel reasonably satisfactory to County), and hold County, its directors, officers, employees, agents, assigns, and any successors to County's interest in the Premises, harmless from and against any and all loss, cost, damage, expense (including reasonable attorneys' fees), claim, cause of action, judgment, penalty, fine, or liability, directly or indirectly, relating to or arising from the use, storage, release, discharge, handling, or presence of Hazardous Materials on, under, or about the Premises or Other Airport Property and caused by Lessee, Lessee's agents, employees, contractors, invitees or Trespassers. This indemnification shall include without limitation: (a) personal injury claims; (b) the payment of liens; (c) diminution in the value of the Premises or Other Airport Property: (d) damages for the loss or restriction on use of the Premises or Other Airport Property; (e) sums paid in settlement of claims; (f) reasonable attorneys' fees, consulting fees, and expert fees, (g) the cost of any investigation of site conditions, and (h) the cost of any repair, cleanup, remedial, removal, or restoration work or detoxification if required by any governmental authorities or deemed necessary in County's reasonable judgment, but shall not extend to such claims, payment, diminution, damages, sums, fees or costs to the extent caused (i) solely by an act of God or (ii) by the negligent or willful misconduct of the County, its officers, employees, contractors or agents. For any legal proceedings or actions initiated in connection with the Hazardous Materials Release, County shall have the right at its expense but not the obligation to join and participate in such proceedings or actions in which the County is a named party, and control that portion of the proceedings in which it is a named party. County may also negotiate, defend, approve, and appeal any action in which County is named as a party taken or issued by any applicable governmental authorities with regard to a Hazardous Materials Release; provided, however, claims for which Lessee may be liable pursuant to this Article 8.15 shall not be settled without Lessee's consent. Any costs or expenses incurred by County for which Lessee is responsible under this paragraph or for which Lessee has indemnified County: (i) shall be paid to County on demand, during the term of this Agreement as additional rent; and (ii) from and after the expiration or earlier termination of the Agreement shall be reimbursed by Lessee on demand. Lessee's obligations pursuant to the foregoing indemnity shall survive the expiration or termination of this Agreement and shall bind Lessee's successors and assignees and inure to the benefit of County's successors and assignees. Notwithstanding any other provision of this Agreement, this section 8.15 does not apply to Baseline Environmental Conditions or a discharge, disposal or release caused by the County, its officers, employees, contractors or agents.

- (A) This indemnity specifically includes the direct obligation of Lessee to perform, at its sole cost and expense, any remedial or other activities required or ordered by court or agency having competent jurisdiction over the subject matter, or otherwise necessary to avoid or minimize injury or liability to any person, or to prevent the spread of Hazardous Materials.
- (B) In addition, because costs of Hazardous Materials discharges are passed on to airlines through the residual rate charging mechanism at Miami International Airport and the charges to Lessee will be increased to offset such costs. Lessee agrees in order to minimize its obligations in this regard to use best efforts to assist the Aviation Department in responding to Hazardous Materials spills in or Airport property reasonably close the Premises used by Lessee by making Lessee's remediation equipment and personnel available for such emergency remediation activity. However, Lessee may provide such assistance only at the direct request of the Aviation Department and only if

Lessee's remediation equipment is intended to be utilized for the Hazardous Material spill at issue and only if Lessee's personnel have been trained to respond to the Hazardous Material spill at issue. If Lessee is directed to perform any remedial work under this Article 8.15(B) for which it is later determined that Lessee is not responsible, the Aviation Department shall reimburse Lessee for all costs associated with or arising out of Lessee's performance of such remedial work. Lessee shall cooperate with the Aviation Department in any subsequent effort by the Aviation Department to recover from the responsible parties all costs involved with the remediation effort that utilized Lessee's equipment and personnel. Lessee shall perform all such work in its own name in accordance with applicable laws. Lessee acknowledges that the County's regulatory power in this regard is independent of the County's contractual undertakings herein, and nothing herein shall affect the County's right in its regulatory capacity to impose its environmental rules, regulations, and authorities upon the Lessee in accordance with the law.

- (C) In the event Lessee fails to perform its obligations in Article 8.15(A) above, and without waiving its rights hereunder, County may, at its option, perform such remedial work as described in Article 8.15(A) above, and thereafter seek reimbursement for the costs thereof. In accordance with this Article 8, Lessee shall permit County or its designated representative access to the Premises areas to perform such remedial activities.
- (D) Whenever County has incurred costs described in this section as a result of the failure of Lessee to perform its obligations hereunder, Lessee shall, within thirty (30) days of receipt of notice thereof, reimburse County for all such expenses together with interest at the rate of 1 ½ % per month on the outstanding balance commencing on the thirty-first date following Lessee's receipt of such notice until the date of payment.
- (E) To the extent of Lessee's responsibility under this Article and without limiting its obligations under any other paragraph of this Agreement, and except to the extent of County's responsibility for environmental conditions set forth in this Article 8, Lessee shall be solely and completely responsible for responding to and complying with any administrative notice, order, request or demand, or any third party claim or demand relating to potential or actual Hazardous Materials contamination on the Premise. Lessee's responsibility under this paragraph includes but is not limited to responding to such orders on behalf of County and defending against any assertion of County's financial responsibility or individual duty to perform under such orders. Lessee shall assume, pursuant to the indemnity provision set forth in this Article 8, any liabilities or responsibilities which are assessed against County in any action described under this paragraph.

8.16 <u>Dispute Resolution</u>: County and Lessee agree that any dispute between them relating to this Article 8 will first be submitted, by written notice, to a designated representative of both County and Lessee who will meet at County's place of business or other mutually agreeable location, or by teleconference, and confer in an effort to resolve such dispute. Any decision of the representatives will be final and binding on the parties. In the event the representatives are unable to resolve any dispute within ten (10) days after submission to them, either party may refer the dispute to mediation, or institute any other available legal or equitable proceeding in order to resolve the dispute.

Waiver and Release: Lessee, on behalf of itself and its heirs, successors 8.17 and assigns, hereby waives, releases, acquits and forever discharges County, its principals, officers, directors, employees, agents, representatives and any other person acting on behalf of the County, and the successors and assigns of any of the preceding, of and from any and all claims, actions, causes of action, demands, rights, damages, costs, expenses or compensation whatsoever, direct or indirect, known or unknown, foreseen or unforeseen, which Lessee or any its heirs, successors, or assigns now has or which may arise in the future on account of or in any way related to or in connection with any past, present or future physical characteristic or condition of the Premises. including, without limitation, any Hazardous Material in, at, on, under or related to the Premises, or any violation or potential violation of any Environmental Law applicable thereto; provided, however, this Article 8.17 shall not constitute a waiver or release of any obligation of County under this Article 8. Lessee acknowledges that County would not enter into this Agreement without Lessee's agreement to the waiver and release provided herein.

8.18 <u>No Waiver of Rights Causes of Actions or Defenses</u>. Notwithstanding any language in this Agreement, including without limitation Articles 8.03, 8.04, 8.05, 8.06, 8.14, 8.15 and 8.16, Lessee does not agree to waive or release any rights, causes of action or defenses it may have against Miami-Dade County or any other party related to allegations made by the County in (i) Case No. 01-8758 CA 25 which has been filed by the County in the Florida Circuit Court of the Eleventh Judicial Circuit, and (ii) a letter dated April 9, 2001, to Lessee and others (who are referred to as "responsible parties" or "RPs"). Nothing herein shall be construed to limit or expand upon any releases previously granted to or exchanged between the parties as a result of judgments or settlements in bankruptcy or settlements entered under Case No. 01-8758 CA 25 which has been filed by the County in the Florida Circuit Court of the parties.

8.19 <u>Surrender of Premises</u>: Lessee shall surrender the Premises used by Lessee to County upon the expiration or earlier termination of this Agreement free of debris, waste, and Hazardous Materials used, stored, or disposed of by Lessee or its agents, employees, contractors, invitees or Trespassers, or otherwise discharged on the Premises or Other Airport Property for which Lessee is responsible during the term of this Agreement. The Premises shall be surrendered in a condition that complies with all applicable Environmental Requirements, and such other reasonable environmental requirements as may be imposed by County. Lessee shall not be responsible under this section 8.19 to the extent of County's obligations under this Article 8.

8.20 <u>Breach</u>: Any breach by Lessee of any provision of this Article 8 shall, after notice and a reasonable opportunity for Lessee to cure, constitute a default of the Agreement and shall entitle County to exercise any and all remedies provided in the Agreement, or as otherwise permitted by law.

8.21 <u>Survivability of Terms</u>: the terms and conditions of this Article 8, including the indemnity, waiver, and release, shall survive the termination of this Agreement.

8.22 <u>Right to Regulate</u>: As provided for in Article 20.12 of this Agreement, nothing within this Article 8 shall be construed to waive or limit, restrain, impair or interfere with the County's regulatory authority.

ARTICLE 9 Indemnification and Hold Harmless

Lessee shall indemnify and hold harmless the County and its officers, employees, agents and instrumentalities from any and all liability, losses or damages, including attorneys' fees and costs of defense, which the County or its officers, employees, agents or instrumentalities may incur as a result of claims, demands, suits, causes of actions or proceedings of any kind or nature arising out of, relating to or resulting from the performance of this Agreement by the Lessee or its employees, agents, servants, partners, principals, contractors or subcontractors, except to the extent attributable to the gross negligence or willful misconduct of the County or its agents and trespassers. Lessee shall pay all claims and losses in connection therewith and shall investigate and defend all claims, suits or actions of any kind or nature in the name of the County, where applicable, including appellate proceedings, and shall pay costs, judgments and attorney's fees which may issue thereon. Lessee expressly understands and agrees that any insurance protection required by this Agreement or otherwise provided by Lessee shall in no way limit the responsibility to indemnify, keep and save harmless and defend the County or its officers, employees, agents and instrumentalities as herein provided.

ARTICLE 10

Assignment and Subletting

The Lessee shall not assign, transfer, pledge or otherwise encumber this Agreement, or sublet all or any portion of the Premises, or allow others to use the Premises without the express prior written consent of the Department.

ARTICLE 11

<u>Insurance</u>

11.01 <u>Insurance Required</u>: In addition to such insurance as may be required by law, the Lessee shall maintain, without lapse or material change, for so long as it occupies the Premises, the following insurance:

- (A) Commercial General Liability Insurance on a comprehensive basis, including Contractual Liability, to cover the Lessee's Premises and Operations, in an amount not less than \$5,000,000 combined single limit per occurrence for bodily injury and property damage. The County must be shown as an additional insured with respect to this coverage
- (B) Automobile Liability Insurance covering all owned, non-owned and hired vehicles used by the Lessee in connection with its operations under this Agreement in an amount not less than:
 - (1) \$5,000,000 combined single limit per occurrence for bodily injury and property damage.
 - (2) \$300,000 combined single limit per occurrence for bodily injury and property damage covering vehicles when being used by the Lessee off of the AOA.
- (C) Pollution Legal Liability Insurance for Lessee's Premises and the activities of Lessee on Other Airport Property (as defined in Article 8.01(l)) in an

amount not less than \$2,000,000 per occurrence covering site assessment, site clean-up, third party claims, including, but not limited to governmental claims, legal defense costs, charges and expenses arising from any on-site and off-site loss, damage, expense or claim related to the release or any threatened release of Hazardous Material.

The insurance coverages required shall include those classifications, as listed in standard liability insurance manuals, which most nearly reflect the operations of the Lessee under this Agreement. All insurance policies required pursuant to the terms of this Agreement shall be issued in companies approved to do business under the laws of the State of Florida. Such companies must be rated no less than "A-" as to management, and no less than "VII" as to strength in accordance with the latest edition of "Best's Insurance Guide", published by A.M. Best Company, Inc., or its equivalent, subject to approval of the County Risk Management Division.

11.02 <u>Insurance Certificates Required</u>: Prior to the commencement of operations hereunder and annually thereafter, the Lessee shall furnish or cause to be furnished certificates of insurance to the Department which certificates shall clearly indicate that:

- (A) The Lessee has obtained insurance in the types, amounts and classifications as required for strict compliance with this Article;
- (B) The policy cancellation notification provisions specify at least 30 days advance written notice of cancellation to the County, except with respect to war and allied perils coverage which shall be 7 days or such lesser period as is customarily available; and
- (C) The County is named as an additional insured with respect to the Lessee's commercial general liability policies.

On said insurance certificates, unless specifically shown to be excluded thereon, commercial general liability coverage shall include contractual liability, and notification of cancellation shall include notification of material changes in the policies.

The County reserves the right to require the Lessee to provide such reasonably amended insurance coverage as it deems necessary or desirable, upon issuance of notice in writing to the Lessee, which notice shall automatically amend this Agreement effective 30 days after such notice.

11.03 <u>Compliance</u>: Compliance with the requirements of this Article 11 (Insurance) shall not relieve the Lessee of its liability under any other portion of this Agreement or any other agreement between the County and the Lessee.

11.04 <u>Right to Examine</u>: The Department reserves the right, upon reasonable notice, to examine the original or true copies of policies of insurance (including but not limited to binders, amendments, exclusions, riders and applications) to determine the true extent of coverage. The Lessee agrees to permit such inspection at the offices of the Department.

11.05 <u>Personal Property</u>: Any personal property of the Lessee or of others placed in the Premises and Airport shall be at the sole risk of the Lessee or the owners thereof, and the County shall not be liable for any loss or damage, except to the extent such loss or damage was caused by the sole active negligence of the County, as limited by Section 768.28, Florida Statutes.

ARTICLE 12 Use of Public Facilities

The County grants to the Lessee, in common with all others desiring to use the Airport, and only to the extent necessary or reasonably desirable, so long as such use does not conflict with the County's operation of the Airport, to carry out the rights granted the Lessee hereunder, the nonexclusive privilege to use the runways, taxiways, roads of egress and ingress, service roads and such other facilities and improvements as may be now in existence or hereafter constructed for the use of persons lawfully using the Airport; provided, however, that such usage shall, to the extent not included as part of the Rent, be subject to the payment of nondiscriminatory fees and other charges established by the County. Nothing herein contained shall grant to the Lessee the right to use any leasable space or area improved or unimproved which is leased to a third party, or which the County has not leased herein.

ARTICLE 13 Termination

13.01 <u>Payment Defaults</u>: Failure of the Lessee to make all payments of rentals, fees and charges required to be paid herein when due shall constitute a default, and the County may, at its option terminate this Agreement after seven calendar days notice in writing to the Lessee unless the default be cured within the notice period.

13.02 <u>Insurance Defaults</u>: The County shall have the right, upon seven calendar days written notice to the Lessee, to terminate this Agreement if the Lessee fails to provide evidence of insurance coverage in strict compliance with Article 11 hereof prior to commencement of operations, or fails to provide a renewal of said evidence upon its expiration; provided, however, that such termination shall not be effective if the Lessee provides the required evidence of insurance coverage within the notice period.

13.03 <u>Other Defaults</u>: The County shall have the right, upon 30 calendar days written notice to the Lessee, to terminate this Agreement upon the occurrence of any one or more of the following, unless the same shall have been corrected within such period, or, if correction cannot reasonably be completed within such 30 day period, in the reasonable discretion of the Department, the Lessee has commenced substantial corrective steps within such 30 day period and diligently pursues same to completion:

(A) Failure of the Lessee to comply with any covenants of this Agreement, other than the covenants to pay rentals, fees and charges when due, and the covenants to provide required evidence of insurance coverage.

(B) The conduct of any business, the performance of any service, or the merchandising of any product or service not specifically authorized herein, by the Lessee.

(C) Failure of the Lessee to comply with any Environmental Law or Environmental Requirement as those terms are defined in Article 8.01 of this Agreement.

13.04 <u>Habitual Default</u>: Notwithstanding the foregoing, in the event that the Lessee has frequently, regularly or repetitively defaulted in the performance of or breached any of the terms, covenants and conditions required herein to be kept and performed by the Lessee, in the reasonable opinion of the County and regardless of whether the Lessee has cured each individual condition of breach or default as provided

in Articles 13.01 Payment Defaults), 13.02 (Insurance Defaults) and 13.03 (Other Defaults) hereinabove, the Lessee shall be determined by the Director to be an "habitual violator." At the time that such determination is made, the Department shall issue to the Lessee a written notice advising of such determination and citing the circumstances therefor. Such notice shall also advise Lessee that there shall be no further notice or grace periods to correct any subsequent breach (es) or default(s) and that any subsequent breach (es) or default(s), of whatever nature, taken with all previous breaches and defaults, shall be considered cumulative and, collectively, shall constitute a condition of noncurable default and grounds for immediate termination of this Agreement. In the event of any such subsequent breach or default, the County may cancel this Agreement upon the giving of written notice of termination to the Lessee, such termination to be effective upon the tenth day following the date of receipt thereof and all payments due hereunder shall be payable to said date, and the Lessee shall have no further rights hereunder.

13.05 <u>Termination by Abandonment</u>: This Agreement shall be automatically terminated upon the abandonment by the Lessee of the Premises or the voluntary discontinuance of operations at the Airport for any period of time exceeding 15 consecutive calendar days, unless such abandonment or discontinuance has been caused by strike, labor disturbance, acts of God, civil disturbance or governmental order that prevents the Lessee's use of the Premises for the purposes authorized in Article 2 (Use of Premises) hereof. Such termination shall not relieve the Lessee of its rental payment obligation for the remaining term of the agreement nor does it constitute a waiver by the Lessor of its rights to recover damages for rental payments for the remaining term of the agreement and loss of future rentals.

13.06 Actions at Termination:

The Lessee shall vacate, quit, surrender up and deliver the Premises to (A) the County on or before the termination date of this Agreement, whether by lapse of time or otherwise. The Lessee shall surrender the Premises in the condition required under Article 4.03 (Maintenance and Repairs) herein, reasonable wear and tear accepted. All repairs for which the Lessee is responsible shall be completed prior to surrender. The Lessee shall deliver to the Department all keys to the Premises upon surrender. On or before the termination date of this Agreement, except in the instance of termination pursuant to Article 13.05 (Termination for Abandonment), in which event the Lessee shall be allowed up to five calendar days from date of termination, and provided that the Lessee is not in default in the payment of any rentals, fees or charges required to be paid herein, the Lessee shall remove all of its personal property from the Premises. Any personal property of the Lessee not removed in accordance with this Article may be removed by the Department for storage at the cost of the Lessee. Failure on the part of the Lessee to reclaim its personal property within 30 days from the date of termination shall constitute a gratuitous transfer of title thereof to the County for whatever disposition is deemed to be in the best interest of the County.

(B) The Lessee shall, at its expense, take all actions required by Federal, State and local laws, regulations or codes to remove from the Premises any hazardous substance or environmental contaminant, whether stored in drums, or found in vats, containers, distribution pipe lines, or the like. All such substances and contaminants shall be removed by the Lessee in a manner approved and authorized by such Federal, State or local laws, regulations or codes.

(C) If the County advises the Lessee that it has reasonable grounds to believe

that any hazardous substance or environmental contaminant has been released within the Premises or into the ground under the Premises, during the term of this Agreement or during the terms of any prior leases between the Lessee and the County for the same or substantially the same Premises, then the Lessee at its expense shall retain an approved environmental consultant to perform whatever environmental assessment may be required to determine the extent of such release. The Lessee shall comply with the recommendations and conclusions, contingent upon County approval, of such consultant regarding environmental cleanup efforts that may be required, and shall comply with any other clean up requirements imposed on the Lessee by Federal, State or local law, regulations or codes.

13.07 <u>Lien Upon Personal Property</u>: In the event of termination for default or upon termination of this Agreement by its term, the County shall have a lien upon all personal property of the Lessee to secure the payment of any unpaid rentals, fees and charges accruing under the terms of this Agreement.

13.08 <u>Right to Show Premises</u>: At any time after the Lessee has been given notice of termination or default, pursuant to this Article 13 (Termination) or other applicable provisions of this Agreement, the County shall have the right to enter on the Premises for the purpose of showing the Premises to prospective tenants or users.

13.09 <u>County Defaults</u>: This Agreement shall be subject to termination by the Lessee in the event of a default by the County in the performance of any covenant or agreement herein required to be performed by the County and the failure of the County to remedy same within a reasonable period of time following receipt of written notice from the Lessee of such default.

13.10 <u>Other Terminations</u>: This Agreement shall be subject to termination by the County or the Lessee in the event of any one or more of the following:

(A) The permanent abandonment of the Airport.

(B) The lawful assumption by the United States Government or any authorized agency thereof, of the operation, control or use of the Airport, or any substantial part or parts thereof, in such a manner as to substantially restrict the Lessee from operating therefrom for a period in excess of 90 consecutive days, provided that nothing contained herein shall be deemed to constitute a waiver by the Lessee of any right it may have against the United States to just compensation in the event of any such assumption.

(C) The issuance by any court of competent jurisdiction of any injunction in any way substantially preventing or restraining the use of the Airport, and the remaining in force of such injunction for a period in excess of 90 days.

ARTICLE 14 Special Conditions

14.01 <u>Quality of Services</u>: The Lessee shall furnish the services required and authorized, pursuant to Article 2 (Use of Premises) hereof, on a good, prompt and efficient basis and on a fair, equal and not unjustly discriminatory basis to all users thereof.

14.02 <u>Nondiscriminatory Prices</u>: The Lessee shall charge fair, reasonable, customary and not unjustly discriminatory prices for each unit of sale or service;

provided, however, that the Lessee may make reasonable, customary and nondiscriminatory discounts, rebates or similar types of price reductions to volume purchasers of the Lessee's services.

14.03 <u>County's Obligations</u>: The Lessee, in recognition of the County's obligation, pursuant to Section 22 of Part V of the Federal Aviation Administration's standard grant assurances, to enforce the provisions of Articles 14.01 (Quality of Service) and 14.02 (Nondiscriminatory Prices) above, agrees that the Department may, from time to time, promulgate standards, methods and procedures for and monitor and test the provision of services hereunder and may require the Lessee to provide copies of schedules of service charges and the bases for discounts, rebates and similar types of price reductions. Should the Department determine that the Lessee is not in compliance with the provisions of Articles 14.01 (Quality of Service) and 14.02 (Nondiscriminatory Prices) above, the first such occurrence shall be considered a curable default, pursuant to Article 13.03 (Other Defaults) hereof, and subsequent occurrence(s) shall be considered a material breach of this Agreement, entitling the County to the remedies provided in this Agreement or by law.

ARTICLE 15

Equal Employment Opportunity, Nondiscrimination and Affirmative Action

15.01 Equal Employment Opportunity: In accordance with Title 14 Code of Federal Regulation (CFR) Part 152 (Affirmative Action Employment Program), the Lessee shall not discriminate against any employee or applicant for employment because of age, sex, race, color, religion, marital status, place of birth or national origin, ancestry, in accordance with the Americans with Disabilities Act, discriminate against any otherwise qualified employees or applicants for employment with disabilities who can perform the essential functions of the job with or without reasonable accommodation. The Lessee shall take affirmative actions' to ensure that applicants are employed and that employees are treated during their employment without regard to age, sex, race, color, religion, marital status, place of birth or national origin, ancestry, or disability. Such actions include, but not limited to, the following: Employment, upgrading, transfer or demotion, recruitment advertising, layoff or termination, rates of pay or other forms of compensation, selection for training including apprenticeship.

The Lessee agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided by the County setting forth the provisions of this equal Employment Opportunity clause. The Lessee shall comply with all applicable provisions of the Civil Rights Act of 1964; Executive Order 11246 issued September 24, 1965, as amended by Executive Order 113155, revised Order No. 4 issued December 1, 1951, as amended, and the Americans with Disabilities Act. The Age Discrimination in Employment Act effective June 12, 1968, Executive Order 13166 issued August 11, 2000, Improving Access to Services for persons with Limited English Proficiency (LEP), the rules, regulations and relevant orders of the Secretary of Labor, Florida Statues § 112.041, §112.042, §112.043 and the Miami-Dade County Code, Section 11A1 through 13A1, Articles 3 and 4.

The Lessee shall assign responsibility to one of its officials to develop procedures that will assure that the policies of Equal Employment Opportunity and Affirmative Action are understood and implemented.

15.02 <u>Nondiscriminatory Access to Premises</u>: The Lessee, for itself, its personal representatives, successors in interest, and assigns, as part of the consideration hereof, does hereby covenant and agree as a covenant that: (1) no person on the grounds of race, color, sex, national origin or ancestry shall be excluded from participation in, denied the benefits of, or be otherwise subjected to discrimination in the use of the Premises; (2) that the lessee shall use the Premises in compliance with all other requirements imposed by or pursuant to the enforceable regulations of the Department of Transportation, as amended from time to time.

15.03: <u>Breach of Nondiscrimination Covenants</u>: In the event it has been determined that the Lessee has breached any enforceable nondiscrimination covenants contained in Section 15.01 Equal Employment Opportunity and Section 15.02 Nondiscriminatory Access to premises above, pursuant to the complaint procedures contained in the applicable Federal Regulations, and the Lessee fails to comply with the sanctions and/or remedies which have been prescribed, the County shall have the right to terminate this Agreement pursuant to the Termination of the Agreement section hereof.

15.04 Nondiscrimination: During the performance of this Agreement, the Lessee agrees as follows: The Lessee shall, in all solicitations or advertisements for employees placed by or on behalf of the Lessee, state that all qualified applicants will receive consideration for employment without regard to age, sex, race, color, religion, marital status, place of birth or national origin, ancestry physical handicap or disability. The Lessee shall furnish all information and reports required by Executive Order 11246 issued September 24, 1965, as amended by Executive Order 113155, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to the Lessee books, records, accounts by the County and Compliance Review Agencies for purposes of investigation to ascertain by the compliance with such rules, regulations, and orders. In the event of the Lessee's noncompliance with the nondiscrimination clauses of this Agreement or with any of the said rules, regulations, and orders, this Agreement may be canceled, terminated, or suspended in whole or in part in accordance with the Termination of Agreement section hereof and the Lessee may be declared ineligible for further contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, as amended by Executive Order 113155 and such sanctions as may be imposed and remedies invoked as provided in Executive Order 113155 and such sanctions as may be imposed and remedies invoked as provided in Executive Order 11246 as amended or by rules, regulations, and orders of the Secretary of Labor, or as otherwise provided by law.

The Lessee will include Section 15.01 Equal Employment Opportunity and Section 15.02 Nondiscriminatory Access to Premises of this Article in the Lessee subcontracts in excess of \$10,000.00, unless exempted by rules, regulations or orders of the Secretary of Labor issued pursuant to Section 204 of Executive Order 11246 issued September 24, 1965, as amended by Executive Order 113155, so that such provisions will be binding upon each sub-consultant. The Lessee shall take such action with respect to any sub-contract as the County may direct as a means of enforcing such provisions, including sanctions for noncompliance; provided, however, that in the event the Lessee becomes involved in, or is threatened with, litigation with a sub-consultant as the result of such direction by the County or by the United States, the Lessee may request the United States to enter into such litigation to protect the interests of the United States. 15.05 <u>Disability Nondiscrimination Affidavit</u>: By entering into this Agreement with the County and signing the Disability Nondiscrimination Affidavit, the Lessee attests that this is not in violation of the Americans with Disabilities Act of 1990 (and related Acts) or Miami-Dade County Resolution No. R-385-95. If the Lessee or any owner, subsidiary or other firm affiliated with or related to the Lessee is found by the responsible enforcement officer of the Courts or the County to be in violation of the Act or the Resolution, such violation shall render this Contract terminable in accordance with the Termination of Agreement section hereof. This Contract shall be void if the Lessee submits a false affidavit pursuant to this Resolution or the Lessee violated the Act or the Resolution during the term of this Contract, even if the Lessee was not in violation at the time it submitted its affidavit.

15.06 <u>Affirmative Action/Nondiscrimination of Employment Promotion and</u> <u>Procurement Practices:</u> (County Code Section 2-8.1.5): In accordance with the requirements of County Code Section 2-8.1.5, all firms with annual gross revenues in excess of \$5 million seeking to contract with Miami-Dade County shall, as a condition of award, have a written Affirmative Action Plan and Procurement Policy on file with the County's Department of Procurement Management. Said firms must also submit, as a part of their Lease to be filed with the Clerk of the Board, an appropriately completed and signed Affirmative Action Plan/Procurement Policy Affidavit.

Firms whose Boards of Directors are representative of the population make-up of the nation are exempt from this requirement and must submit, in writing, a detailed listing of their Boards of Directors, showing the race or ethnicity of each board member, to the County's Department of Procurement Management. Firms claiming exemption must submit, as part of their Lease to be filed with the Clerk of the Board, an appropriately completed and signed Exemption Affidavit in accordance with the County Code Section 2-8.1.5. These submittals shall be subject to periodic reviews to assure that the entities do not discriminate in their employment and procurement practices against minorities and women/owned businesses.

It will be the responsibility of each firm to provide verification of their gross annual revenues to determine the requirement for compliance with the County Code Section. Those firms that do not exceed \$5 million annual gross revenues must clearly state so in their Lease.

ARTICLE 16

Security and Special Provisions

16.01 <u>Security</u>: The Lessee acknowledges and accepts full responsibility for (i) the security and protection of the Premises, any improvements thereon, its equipment and property on the Airport and (ii) all breaches of federal and Lessor security requirements by Lessee's employees or those persons for whom Lessee has responsibility under Article 16.02, and (iii) control of access to the AOA through the Premises by persons and vehicles. The Lessee fully understands and acknowledges that any security measures deemed necessary by the Lessee for the protection of said Premises, equipment and property shall be the sole responsibility of the Lessee and shall involve no cost to the County. Lessee is responsible for compliance by its employees and all others for whom it is responsible with applicable security requirements relating to access, through Lessee's Premises or otherwise, to the AOA or any Security

Identification Display Area ("SIDA"). All such security measures by the Lessee shall be in accordance with FAR 107, 49 CFR Part 1542 and the Airport Security Plan.

Security Identification Display Areas Access - Identification Badges: The 16.02 Lessee shall be responsible for (i) assuring that all of Lessee's employees, and all employees and persons of entities using the Premises or a SIDA on behalf of Lessee (collectively herein, the "SIDA Users"), have appropriate SIDA Identification Badges and comply with all federal and Lessor security requirements applicable to the Premises and SIDAs, (ii) immediately reporting to MDAD all lost or stolen ID badges of a SIDA User. and (iii) immediately returning the ID badges of any SIDA User that is transferred from the Airport or terminated from the employ of the Lessee or upon termination of this Agreement. Each employee must complete the SIDA training program conducted by the Department, before an ID badge is issued. The Lessee shall pay, or cause to be paid, to the Department any fines or penalties imposed on Lessor for a violation of the security requirements by a SIDA User as well as such nondiscriminatory charges, as may be established from time to time, for lost or stolen ID badges and those not returned to the Department in accordance with this Article. The Department shall have the right to require the Lessee to conduct background investigations and to furnish certain data on such employees before the issuance of ID badges, which data may include the fingerprinting of employee applicants for the badges.

16.03 <u>AOA - Driver Training</u>: Before the Lessee shall permit any employee to operate a motor vehicle of any kind or type on the AOA, the Lessee shall require such employee to attend and successfully complete the AOA Driver Training Course conducted from time to time by the Department. The privilege of a person to operate a motor vehicle on the AOA may be withdrawn by the Department for any violation of AOA driving rules. Notwithstanding the above, the Lessee shall be responsible for ensuring that all such vehicle operators possess current, valid, appropriate Florida driver's licenses.

16.04 Alcohol and Drug Testing: The Lessee acknowledges that the County. as a public agency sponsor under the provisions of the Airport and Airway Improvement Act of 1982, as amended (the "Act"), has the obligation to establish a drug free workplace and to establish policies and programs to ensure airport safety and security. The Lessee acknowledges that the Department, on behalf of the County, has the right to require users of the Airport (Lessees, Permittees, Licensees, etc.) to establish reasonable programs to further the achievement of the objectives described herein. Accordingly, the Lessee shall establish programs for pre-employment alcohol and drug screening for all candidates for employment at the Airport who will as a part of their duties (a) be present on the AOA; (b) operate a motor vehicle of any type on the AOA; or (c) operate any equipment, motorized or not, on the AOA and for the same or similar screening based upon a reasonable suspicion that an employee, while on duty on the AOA, may be under the influence of alcohol or drugs. Notwithstanding the above, the Lessee specifically acknowledges that the County, acting through the Department, has the right and obligation to deny access to the AOA and to withdraw AOA driving privileges from any person who it has a reasonable suspicion to believe is under the influence of alcohol or drugs.

16.05 <u>Drug-Free Workplace Default</u>: The Lessee acknowledges it has provided to the County a Drug-Free Workplace Affidavit certifying that it is providing a drug-free workplace for its employees, as required by County Ordinance No. 92-15, adopted on March 17, 1992 as amended from time to time ("Ordinance"). Based on the provisions of said Ordinance, the County shall have the right, upon 30 days written notice to the Lessee, to terminate this Agreement in the event the Lessee fails to provide, as of each

anniversary of the effective date of this Agreement, the annual re-certification affidavit as required by the Ordinance; provided, however, that such termination shall not be effective if the Lessee submits the required Affidavit within the notice period.

Further, this Agreement shall be terminated upon not less than fifteen calendar days written notice to the Lessee and without liability to the County, if the Department or the County Manager determines any of the following:

- (A) That the Lessee has made a false certification in its execution of the Affidavit submitted or in its annual re-certification as required by the Ordinance;
- (B) That the Lessee has violated its original or renewal certification by failing to carry out any of the specific requirements of the Ordinance, other than the annual re-certification; or
- (C) That such a number of employees of the Lessee have been convicted of violations occurring in its workplace(s) as to indicate that the Lessee has failed to make a good faith effort to provide a drug-free workplace as required by the Ordinance.

16.06 <u>Special Programs</u>: The Lessee shall ensure that all employees so required participate in such safety, security and other training and instructional programs, as the Department or appropriate Federal agencies may from time to time require.

16.07 <u>Vehicle Permit and Company Identification</u>: Motor vehicles and equipment of the Lessee operating on the AOA must have an official motor vehicle identification permit issued pursuant to Operational Directives of the Department. In addition, company identification must be conspicuously displayed thereon.

16.08 <u>Federal Agencies Right to Consent</u>: The Lessee understands and agrees that all persons entering and working in or around arriving international aircraft and facilities used by the various Federal Inspection Services agencies may be subject to the consent and approval of such agencies. Persons not approved or consented to by the Federal Inspection Services agencies shall not be employed by the Lessee in areas under the jurisdiction or control of such federal inspection agencies.

16.09 <u>AOA - Right to Search</u>: The Lessee agrees that its vehicles, cargo, goods and other personal property are subject to being searched when attempting to enter or leave and while on the AOA. The Lessee further agrees that it shall not authorize any employee or agent to enter the AOA unless and until such employee or agent has executed a written consent-to-search form acceptable to the Department. Persons not executing such consent-to-search form shall not be employed by the Lessee at the Airport, in any job requiring access to the AOA.

It is further agreed that the Department has the right to prohibit an individual, agent or employee of the Lessee from entering the AOA based upon facts which would lead a person of reasonable prudence to believe that such individual might be inclined to engage in theft, cargo tampering, aircraft sabotage or other unlawful activities. Any person denied access to the AOA or whose prior authorization has been revoked or suspended on such grounds shall be entitled to a hearing before the Director of the Department or his authorized designee within a reasonable time. Prior to such hearing, the person denied access to the AOA shall be advised, in writing, of the reasons for such denial.

The Lessee acknowledges and understands that these provisions are for the protection of all users of the AOA and are intended to reduce the incidence of thefts, cargo tampering, aircraft sabotage and other unlawful activities at the Airport.

16.10 <u>Right of Flight</u>: There is hereby reserved to the County, its successors and assigns, for the use and benefit of the County and the public, a right of flight for the passage of aircraft in the air space above the surface of the premises herein leased, together with the right to cause in said air space such noise as may be inherent in the operation of aircraft, now known or hereafter used for navigation of or flight in the air, using said air space for landing at, taking off from or operating on Miami International Airport.

16.11 <u>Height Restrictions</u>: The Lessee expressly agrees for itself, its successors and assigns, to restrict the height of structures, objects of natural growth and other obstructions on the leased premises to such a height so as to comply with Federal Aviation Regulations, Part 77 and with the Code of Miami-Dade County, whichever is more restrictive.

ARTICLE 17 Control of Employees

17.01 <u>Control of Employees</u>: Lessee shall properly control the actions of its employees at all times that said employees are working on the Airport, ensuring that they present a neat appearance and discharge their duties in a courteous and efficient manner and that they maintain a high standard of service to the public.

17.02 <u>Lessee's Responsibility for Employee's Violations</u>: In the event the Lessee is in default of the covenants of Article 17.01 (Control of Employees) for failure to properly control its employees or by permitting its employees to improperly use the facilities by the County, the Department shall have the right to require the Lessee to conduct an investigation into any claimed violation of the covenants; if such investigation substantiates a violation, Lessee agrees to administer the appropriate discipline up to and including discharge of the offending employee.

ARTICLE 18 Civil Actions

18.01 <u>Governing Law; Venue:</u> This Agreement shall be governed and construed in accordance with the laws of the State of Florida. The venue of any action on this Agreement shall be laid in Miami-Dade County, Florida, and any action to determine the rights or obligations of the parties hereto shall be brought in the courts of the State of

Florida.

18.02 <u>Notice of Commencement of Civil Action</u>: In the event that the County or the Lessee commence a civil action where such action is based in whole or in part on an alleged breach of this Agreement, the County and the Lessee agree the service of process shall be made pursuant to the rules of Civil Procedure in the court in which the action has been filed.

18.03 <u>Registered Office/Agent; Jurisdiction</u>: Notwithstanding the provisions of Article 18.02 (Notice of Commencement of Civil Action), and in addition thereto, the Lessee, if a corporation, shall designate a registered office and a registered agent, as required by Section 48.091, Florida Statutes, such designations to be filed with the

Florida Department of State in accordance with Section 607.0501, Florida Statutes. If the Lessee is a natural person, he and his personal representative hereby submit themselves to the jurisdiction of the Courts of this State for any cause of action based in whole or in part on an alleged breach of this Agreement.

ARTICLE 19

Trust Agreement

19.01 Incorporation of Trust Agreement by Reference: Notwithstanding any of the terms, provisions and conditions of this Agreement, it is understood and agreed by the parties hereto that, to the extent of any inconsistency with or ambiguity relating to the terms and conditions of this Agreement, and the level of rents, fees or charges required hereunder and their periodic modification or adjustment as may be required by the Amended and Restated Trust Agreement dated as of December 15, 2002, by and among the County and the JP Morgan Chase Bank as Trustee and the Wachovia Bank, National Association as Co-trustee (the "Trust Agreement"), shall prevail and govern at all times during the term of this Agreement. Copies of the Trust Agreement are available for inspection in the offices of the Department during normal working hours.

19.02 Adjustment of Terms and Conditions: If, at any time during the term of this Agreement, a Federal agency or court of competent jurisdiction shall determine that any of the terms and conditions of this Agreement, including the rentals, fees and charges required to be paid hereunder to the County by the Lessee or by other Lessees under other Agreements of the County for the lease or use of facilities used for similar purposes, are unjustly discriminatory, the County shall have the right to modify such terms and conditions and to increase or otherwise adjust the rentals, fees and charges required to be paid under this Agreement in such a manner as the County shall determine is necessary and reasonable so that the rentals, fees and charges payable by the Lessee and others shall not thereafter be unjustly discriminatory to any user of like facilities and shall not result in any violation of the Trust Agreement or in any deficiency in revenues necessary to comply with the covenants of the Trust Agreement. In the event the County has modified the terms and conditions of this Agreement. including any adjustment of the rentals, fees and charges required to be paid to the County pursuant to this provision, this Agreement shall be amended to incorporate such modification of the terms and conditions including the adjustment of rentals, fees and charges upon the issuance of written notice from the Department to the Lessee.

19.03 Lessee Right to Terminate: In the event the terms and conditions of this Agreement, including the rentals, fees and charges payable hereunder, have been substantially modified pursuant to Article 19.02 (Adjustment of Terms and Conditions) above, the Lessee, at any time within one year following the effective date of such modification may terminate this Agreement by giving ninety days written notice to the County, without liability by any party to any other party.

ARTICLE 20 Other Provisions

20.01 <u>No Representation</u>: The County makes no representation, warranty, guarantee, or averment of any nature whatsoever concerning the physical condition of the Premises, and it is agreed that the County will not be responsible for any loss, damage or costs which may be incurred by the Lessee by reason of any such physical condition, except as to costs agreed to be reimbursed by County under Article 5.03.

20.02 <u>Headings</u>: Any headings preceding the text of any articles, paragraphs or sections of this Agreement shall be solely for convenience of reference and shall not

constitute a part of this Agreement, nor shall they affect its meaning, construction or effect.

20.03 <u>Interference</u>: The Lessee further expressly agrees to prevent any use of the Premises which would interfere with or adversely affect the operation or maintenance of the Airport or otherwise constitute an airport hazard.

20.04 <u>Authorized Uses Only</u>: The Lessee shall not use or permit the use of the Airport for any illegal or unauthorized purpose or for any purpose which would increase the premium rates paid by the County on, or invalidate, any insurance policies of the County or any policies of insurance written on behalf of the Lessee under this Agreement.

20.05 <u>Binding Effect</u>: The terms, conditions and covenants of this Agreement shall inure to the benefit of and be binding upon the parties hereto and their successors and assigns. This provision shall not constitute a waiver of any conditions prohibiting assignment or subletting.

20.06 <u>Federal Subordination</u>: This Agreement shall be subordinate to the provisions of any existing or future agreements between the County and the United States of America relative to the operation and maintenance of the Airport, the execution of which has been or may be required as a condition precedent to the expenditure of Federal funds for the development of the Airport. All provisions of this Agreement shall be subordinate to the right of the United States of America to lease or otherwise assume control over the Airport, or any part thereof, during time of war or national emergency for military or naval use and any provisions of this Agreement inconsistent with the provisions of such lease to, or assumption of control by, the United States of America shall be suspended.

20.07 <u>Notices</u>: All notices required or permitted to be given under the terms and provisions of this Agreement by either party to the other shall be in writing and shall be hand delivered or sent by registered or certified mail, return receipt requested, to the parties as follows:

As to the County or Aviation Department:

Director Miami-Dade County Aviation Department Post Office Box 025504 Miami, Florida 33152-5504

As to the Lessee:

President National Airport Services, Inc. 6025 NW 18th Street Bldg. #716-E, Suite #608 Miami, Florida 33126 Attn: Jorge Y. Alvarez

or to such other address as may hereafter be provided by the parties in writing. Notices by registered or certified mail shall be deemed received on the delivery date indicated by the U.S. Postal Service on the return receipt. Hand delivered notices shall be deemed received by the Lessee when presented to the local management representative of the Lessee.

20.08 <u>Rights Reserved</u>: Rights not specifically granted the Lessee by this Agreement are reserved to the County.

20.09 <u>Rights of County at Airport</u>: The County shall have the absolute right, without limitation, to make any repairs, alterations and additions to any structures and facilities at the Airport. The County shall, in the exercise of such right, be free from any and all liability to the Lessee for business damages occasioned during the making of such repairs, alterations and additions, except those occasioned by the sole active negligence of the County, its employees, or agents.

20.10 <u>Rights to be Exercised by Department</u>: Wherever in this Agreement rights are reserved to the County, such rights may be exercised by the Department.

20.11 <u>No Waiver</u>: There shall be no waiver of the right of either party to demand strict performance of any of the provisions, terms and covenants of this Agreement nor shall there be any waiver of any breach, default or non-performance hereof by either party, unless such waiver is explicitly made in writing by the other party. Any previous waiver or course of dealing shall not affect the right of either party to demand strict performance of the provisions, terms and covenants of this Agreement with respect to any subsequent event or occurrence of any subsequent breach, default or non-performance hereof by the other party.

20.12 <u>Right to Regulate</u>: Nothing in this Agreement shall be construed to waive or limit the governmental authority of the County, as a political subdivision of the State of Florida, to regulate the Lessee or its operations. Notwithstanding any provision of this Agreement, nothing herein shall bind or obligate the County, the Zoning Appeals Board, the Building and Zoning Department (as it may be renamed from time to time), the Planning Department, or any department, board or agency of the County, to agree to any specific request of the Lessee that is related in any way to the regulatory or quasi-judicial power of the County; and the County shall be released and held harmless by the Lessee from any liability, responsibility, claims, consequential damages or other damages, or losses resulting from the denial or withholding of such requests; provided, however, that this provision shall not preclude any appeal from County action wherein the sole remedy sought is reversible of the County's bad faith, capricious behavior or arbitrary action.

20.13 <u>Severability</u>: If any provision of this Agreement or the application thereof to either party to this Agreement is held invalid by a court of competent jurisdiction, such invalidity shall not affect other provisions of this Agreement which can be given effect without the invalid provision, and to this end, the provisions of this Agreement are severable.

20.14 <u>Inspections</u>: The authorized employees and representatives of the County and of any applicable Federal or State agency having jurisdiction hereof shall have the right of access to the Premises at all reasonable times for the purposes of inspection and testing to determine compliance with the provisions of this Agreement. This right of inspection and testing shall impose no duty on the County to inspect and shall impart no liability upon the County should it not make any such inspections.

20.15 <u>Payment of Taxes</u>: The Lessee shall pay all taxes and other costs lawfully assessed against its leasehold interests in the Premises, its improvements and its operations under this Agreement; provided, however, the Lessee shall not be deemed to be in default of its obligations hereunder for failure to pay such taxes pending the outcome of any legal proceedings instituted to determine the validity of such taxes. Failure to pay the taxes upon the adverse ultimate conclusion of such legal proceedings

against the Lessee shall constitute a default pursuant to Article 13.03 (Other Defaults).

20.16 <u>Quiet Enjoyment</u>: The Lessee shall control the actions of its employees, agents, invitees and those doing business with it, so as to not annoy, disturb or be offensive to others and to provide the service hereunder so as to not unreasonably create a nuisance or thing which may disturb the quiet enjoyment of any other users of the Airport.

20.17 <u>Radon Disclosure</u>: In accordance with Section 404.056, Florida Statutes, the following disclosure is hereby made:

Radon Gas: Radon is a naturally occurring radioactive gas that, when it has accumulated in a building in sufficient quantities, may present health risks to persons who are exposed to it over time. Levels of radon that exceed federal and state guidelines have been found in buildings in Florida. Additional information regarding radon and radon testing may be obtained from your county public health unit.

20.18 <u>Destruction of Premises</u>: In the event the Premises shall be destroyed or so damaged or injured by fire, windstorm, flood or other casualty (and in each such event the Lessee was not at fault in whole or in part) during the life of this Agreement that the Premises or any portion thereof are rendered untenantable, the County shall have the right, but not the obligation, to render said Premises or damaged portion thereof tenantable by repairs completed within a reasonable period of time.

- (A) <u>Total Destruction</u>: In the event the County elects not to render the Premises tenantable, if destroyed or damaged in their entirety, the Lessee shall be so notified in writing by the Department, and this Agreement shall be deemed terminated as of the date of the casualty, with the Lessee being liable only for payment of rentals on a pro rata basis as to whatever portion(s) of the Premises which were tenantable and used by the Lessee following the casualty. In such event, the Department shall endeavor to find adequate replacement premises for the Lessee in existing facilities on the Airport.
- (B) If the damaged portion of the Premises is not rendered tenantable by the County within a reasonable period of time, and the Lessee shall determine that: 1) the loss of the damaged portion of the Premises shall have a materially adverse impact on the ability of the Lessee to utilize the Premises for the purposes described in Article 2; or 2) would require the Lessee to obtain other space off the Premises in order to substantially conduct the operations of the Lessee originally conducted within the Premises, then, in either such event, upon written notice to the County, the Lessee may cancel this Agreement as of a date which shall be not later than one year from the giving of such notice, if the repairs are not completed within 90 days following such written notice of intent to cancel, or if the repairs cannot be reasonably completed within such 90-day period the County has not commenced repairs within such time. In the event of cancellation, the rent for the untenantable portion of the Premises shall be paid only to the date of such fire, windstorm, flood, or other casualty. If the Agreement is not canceled following any such casualty, the rent shall be abated as to the portion of the Premises rendered untenantable. If the casualty was caused in whole or in part by the Lessee, its officers, employees, agents, contractors or Trespassers, then the Lessee shall not have the right to terminate this Agreement and shall

be responsible under other provisions of this Agreement for payment to the County of all damage to the Premises, plus the loss of rentals attributable to the damaged or destroyed premises.

20.19 <u>Quiet Enjoyment:</u> Subject to the terms of this Agreement, specifically including, but not limited to, environmental remediation steps to be taken under Article 8, the County's right and obligation to make certain repairs, alterations, and additions under Articles 5 (Maintenance by County) and 20.09 (Rights of County at Airport), which, for purposes of this clause, includes any and all demolition, in whole or in part, of buildings and runways, and roadway systems on or off the Airport, and the reservation of easement rights to the airspace under Article 16.10 (Right of Flight), all of which provisions and others in this Agreement, the Lessee acknowledges may cause disruption and disturbance to the Lessee, and upon the observance by the Lessee of all the terms, provisions, covenants, and conditions imposed upon the Lessee hereunder, the Lessee shall peaceably and quietly hold and enjoy the Premises for the term of this Agreement; provided, however, that the County shall not be liable for any violation of this clause or for any disturbance or disruption in or to the Lessee's business, for acts or omissions of tenants, users of the Airport, third parties or when any department or agency of the County is acting in its governmental capacity or by Acts of God.

20.20 <u>Interpretation of Agreement:</u> This Agreement is the result of negotiation between the parties hereto and has been typed/printed by one party for the convenience of both parties, and the parties covenant that this Agreement shall not be construed in favor of or against any of the parties hereto.

20.21 <u>Entirety of Agreement</u>: The parties hereto agree that this Agreement sets forth the entire agreement between the parties, and there are no promises or understandings other than those stated herein. None of the provisions, terms and conditions contained in this Agreement may be added to, modified, superseded or otherwise altered, except as may be specifically authorized herein or by written instrument executed by the parties hereto.

JYA.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their appropriate officials as of the date first above written.

BOARD OF COUNTY COMMISSIONERS OF MIAMI-DADE COUNTY, FLORIDA

By: <u>K. U. Mark</u> Deputy Aviation Director ATTEST: Harvey Ruvin, Clerk By: Deputy Clerk (SEAL) NATIONAL AIRPORT SERVICES, INC. By: President 0260

/ed as t gal suff An

Assistant County Attorney

Corporate Secretary

Print Name

(SEAL)

ATTEST:



ST.	TAXIMIAY "Q" RUNWAY 12:30 UJ					
	$\mathbb{A} = \mathbb{A} \mathbb{A} = \mathbb{A}$					
Miss and	raymar "P"					
	6 Miles William Wi					
	WINB WINA					
Airfield A	KEY MAP					
CODE: SPACE CLASS SQ. FT. AIRCRAFT PARKING	MIAMI DADE AVIATION DEPARTMENT MIAMI INTERNATIONAL AIRPORT					
(As assigned by Airside Operations)	EXHIBIT P					
SCALE: N.T.S. EFS #: 6230 DATE: 3/1/2016	NATIONAL AIRPORT SERVICES					

Miami-Dade County Office of the Inspector General



APPENDIX A

Miami-Dade Aviation Department's Response (2 pages)

19-0025-I



This memorandum acknowledges receipt of the above referenced draft report of investigation, dated December 22, 2024, for National Airport Services (NAS). The Miami-Dade Aviation Department (Department) appreciates the opportunity to review the report and provide a response that will be included in the final report. The Office of the Inspector General's (OIG) recommendations are replicated below as written, bolded and italicized, followed by the Department's response.

OIG Recommendations and Department's Response

Recommendation 1: MDAD should review its procedures to ensure that MDAD's Properties and Finance sections specifically communicate to ensure that lessees do not have outstanding audit reports or a history of gross revenue underreporting that should be considered prior to extending or renewing lease agreements.

Department Response

The Department concurs. The Finance Division will further enhance communications with the Real Estate Management Division (REM) to mitigate tenant's habitual noncompliance with contract terms. Property Managers will be immediately notified of tenants with outstanding independent audit reports and encouraged to begin eviction proceedings if not remedied within 60 days. Finance staff have been reminded to guestion material adjustments generated from the audit reports and to carefully review management letters, advising Supervisors of any discrepancies noted by the independent auditors. Further, within 30 days NAS will be formally notified to discontinue its practice of reporting monthly gross revenue on a cash basis and to resubmit, within 60 days, revised monthly gross revenue reports from August 2020 to December 2023, reflecting gross revenue earned on an accrual basis prospectively. Fees due, including penalties and interest will be billed. The Real Estate Division has agreed to enhanced communication with Finance, reiterating tenants must be brought into full compliance with any outstanding agreement requirements, including outstanding fees or the submittal of independent audits before agreements are either extended or replaced. Lastly, the Audit and Management Services Department will be asked to do a follow-up audit on NAS in the next year to ensure compliance.

MDAD OIG Investigation Response – IG19-0025-I Page 2 of 2

Recommendation 2: MDAD should review its procedures regarding tenants who habitually underreport gross revenues to determine if any new procedures, enhanced penalties, and training might be implemented to minimize the occurrence and the risk to MDAD or MDAD should exercise its rights under the contracts to terminate the agreements.

Department Response

The Department concurs. The Accounts Receivable Section in Finance consistently monitors receivable balances and follows-up daily or weekly with those tenants with outstanding balances to collect payments past the due dates specified in each agreement. If unsuccessful, they immediately communicate with the Property Managers via email to request collection assistance or encourage them to begin eviction proceedings. Consistent email reminders are sent, as required by established procedures.

Recommendation 3: Generally, MDAD should periodically review outstanding fines relating to citations issued to all lessees and take action to recover the fines and penalties due, particularly before a new lease agreement is executed. Specifically, as to NAS, MDAD should review the unpaid citations issued to NAS and recover the fines and penalties due.

Department Response

The Department concurs. However, the collection of fines from issued citations is not a department responsibility. All issued citations are remitted to the Miami-Dade County Clerk of Courts, Code Compliance and Collections Section. Our understanding is they will issue additional payment notifications to the violator and/or the violator's company until the fine has been paid. If the violator does not pay the citation, they are subject to additional fines or penalties including placement of a lien on the violator's properties, by the Clerk of the Courts. When the fine is paid, we are notified, and we update our records of issued citations.

We had made efforts in the past to assist with the collection of fines by deactivating identification badges of those who have outstanding fines. However, this action created security problems for the Transportation Security Administration and our Public Safety and Security Division, requiring the practice to be stopped.

Nonetheless, we will notify the Clerk of the Courts and request they revisit these citations and employ additional enforcement efforts to obtain payment.